



# Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.

David J. Wagg \*1

<sup>1</sup> Department of Mechanical Engineering, University of Sheffield, S1 3JD, UK

## Abstract

The aim of this paper is to provide an introduction to using normal form transformations for linear and nonlinear structural dynamics examples. Starting with linear single-degree-of-freedom systems, a series of examples are presented that eventually lead to the analysis of a system of two coupled nonlinear oscillators. A key part of normal form transformations are the associated coordinate transformations. This review includes topics such as Jordan normal form and modal transformations for linear systems, while for nonlinear systems, near-identity transformations are discussed in detail. For nonlinear oscillators, the classical methods of Poincaré and Birkhoff are covered, alongside more recent approaches to normal form transformations. Other important topics such as nonlinear resonance, bifurcations, frequency detuning and the inclusion of damping are demonstrated using examples. Furthermore, the connection between normal form transformations and Lie series is described for both first and second-order differential equations. The use of normal form transformations to compute backbone curves is described along with an explanation of the relationship to nonlinear normal modes. Lastly, conclusions and possible future directions for research are given.

**Keywords:** Normal form; coordinate transformation; nonlinear oscillator; nonlinear resonance; frequency detuning

Received on March 29, 2021, Accepted on January 12, 2022, Published on January 17, 2022

## 1 Introduction

Structural dynamics involves modelling the time dependent behaviour of mechanical and structural systems, using differential equations. To give this process meaning, a relevant set of coordinates needs to be defined. If the coordinates are chosen such that the differential equation can be written in its *simplest* (or some other *most natural*) form, then this is an example of a normal form. The process of finding a normal form is undertaken by coordinate transformations, and hence the phrase *normal form transformation* is used to indicate that the process is intended to result in a normal form for the system under consideration.

So why this might be useful for a structural dynamicist? There are two main reasons. Firstly, differential equations can be difficult to solve (exactly or approximately, depending on the context) and transforming to a simpler form typically makes it easier to find potential solutions. Secondly, the transformation process reveals important information about the resonant behaviour of the system. For example, structural dynamicists have used these type of methods extensively in modal analysis [45, 64], and more recently they have been applied to nonlinear normal modes [172, 104, 129], and the understanding of nonlinear coupled oscillators [4, 124, 85, 183]. They are also used in nonlinear dynamics, particularly in the area of bifurcation analysis [55, 100], and more widely in mathematics and physics [120, 22, 116]. We start by considering some simple examples.

<sup>\*</sup>david.wagg@sheffield.ac.uk

## 1.1 Motivating examples

## 1.1.1 Example 1: Unforced, undamped linear oscillator

A common approach in structural dynamics is to use the Newtonian framework to model structures with point masses. In this situation the positions, velocities and accelerations of masses are the natural starting point for defining the relevant coordinates. Consider, for example, the point mass system shown in Fig. 1 (a). The ordinary differential equation representing the motion of this system, obtained by applying Newton's second law<sup>1</sup> is given by

$$m\ddot{x} + kx = 0, \quad \rightsquigarrow \quad \ddot{x} + \omega_n^2 x = 0, \tag{1}$$

where *x* is the vertical displacement of the mass, *m*, and *k* is the spring constant (also called the stiffness), from which  $\omega_n = \sqrt{k/m}$  is the undamped natural frequency. In Eq. (1) a dot indicates differentiation with respect to time *t* such that  $\ddot{x}$  represents acceleration of the mass. Both acceleration and displacement are assumed to act in a single, linear, coordinate direction called a *degree-of-freedom*, and as there is only one, this is a *single degree-of-freedom* system.

Structural dynamicists tend to refer to Eq. (1) by what it doesn't include — the *unforced*, *undamped*, *linear oscillator*, where linear in this context means the absence of nonlinear terms of the form  $x^2, x^3, x\dot{x}, \sin(x)$  etc. in the equation. But how do we know if Eq. (1) is a normal form? It would seem that it is already quite 'simple', but can it be simplified further?

One possibility is to reduce the order of the differentiation from second-order (meaning  $d^2/dt^2$ ) to first-order (meaning d/dt). To do this we define the following new coordinates  $x_1 = x$  and  $x_2 = \dot{x}$  such that we can write an equivalent first-order matrix differential equation for Eq. (1) which is given by

$$\begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & 0 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix},$$
(2)

which can be written in matrix notation as

$$\dot{\boldsymbol{x}} = A\boldsymbol{x},\tag{3}$$

where  $\mathbf{x} = \{x_1, x_2\}^T$  is referred to as the state vector. Now we have an equation that is 'simpler' in the sense that it is first-order in terms of differentiation, but arguably more complicated in that *A* is a non-diagonal matrix, meaning there is *coupling* between the equations in Eq. (3). However, if we now apply a coordinate transformation  $\mathbf{x} \rightarrow \mathbf{q}$  to 'diagonalise' *A*, then (assuming no other complicating terms arise in the transformation process) we would have a system of two *uncoupled* equations.

This can be done by finding the eigenvalues and eigenvectors<sup>2</sup> of *A* which for this example are  $\lambda_1 = i\omega_n$ ,  $\lambda_2 = -i\omega_n$ ,  $v_1 = \{1, \lambda_1\}^T$ , and  $v_2 = \{1, \lambda_2\}^T$  respectively. These can be used to define the following matrices

$$\Lambda = \begin{bmatrix} \mathbf{i}\omega_n & 0\\ 0 & -\mathbf{i}\omega_n \end{bmatrix}, \quad \Phi = \begin{bmatrix} 1 & 1\\ \mathbf{i}\omega_n & -\mathbf{i}\omega_n \end{bmatrix}, \quad \Phi^{-1} = \frac{1}{-2\mathbf{i}\omega_n} \begin{bmatrix} -\mathbf{i}\omega_n & -1\\ -\mathbf{i}\omega_n & 1 \end{bmatrix}.$$
(4)

Then by making the substitution  $x = \Phi q$  into Eq. (3) and multiplying by  $\Phi^{-1}$  we obtain

$$\begin{bmatrix} \dot{q}_p \\ \dot{q}_m \end{bmatrix} = \begin{bmatrix} \mathbf{i}\omega_n & 0 \\ 0 & -\mathbf{i}\omega_n \end{bmatrix} \begin{bmatrix} q_p \\ q_m \end{bmatrix}, \quad \text{or} \quad \dot{\boldsymbol{q}} = \Lambda \boldsymbol{q} \quad \text{where} \quad \boldsymbol{q} = \begin{bmatrix} q_p \\ q_m \end{bmatrix}.$$
(5)

These equations in the new coordinates,  $q_p$  and  $q_m$ , can now be solved via direct integration with initial conditions  $q_p(t_0) = q_{p0}$ , (and similarly for  $q_m$ ) such that taking the first row of the first equation in Eq. (5), we can rearrange and integrate both sides to give

$$\int_{q_{p0}}^{q_p} \frac{\mathrm{d}q_p}{q_p} = \mathbf{i}\omega_n \int_{t_0}^t \mathrm{d}t \quad \rightsquigarrow \quad \ln(q_p) - \ln(q_{p0}) = \mathbf{i}\omega_n(t - t_0) \quad \rightsquigarrow \quad \ln(\frac{q_p}{q_{p0}}) = \mathbf{i}\omega_n(t - t_0), \tag{6}$$

<sup>&</sup>lt;sup>1</sup>An early classical treatment of this example can be found in the Theory of Sound by Lord Rayleigh [145, 146], which sets out a theoretical approach to vibrations that was subsequently used as a template for many others that followed (for example; Den Hartog [32], Timoshenko [167], Bishop and Johnson [12], Warburton [185], Newland [130], Thompson [166], Géradin and Rixen [48]), and is still widely used today (e.g. Ewins [45], Meirovitch [115], Hagedorn and DasGupta [58], Inman [76], Rao and Yap [143], to list just a few).

<sup>&</sup>lt;sup>2</sup>Note that the eigenvectors are not unique, and a choice needs to be made regarding the relative scaling of these vectors. The choice used here is to scale the leading value of the vector to be one.

so finally we can obtain

$$\frac{q_p}{q_{p0}} = e^{\mathbf{i}\omega_n(t-t_0)} \quad \rightsquigarrow \quad q_p = e^{\mathbf{i}\omega_n(t-t_0)}q_{p0},\tag{7}$$

and if  $t_0 = 0$  is assumed<sup>3</sup> (and using a similar approach for  $q_m$ ) we have the solutions

$$q_p = e^{\mathbf{i}\omega_n t} q_{p0}, \quad \text{and} \quad q_m = e^{-\mathbf{i}\omega_n t} q_{m0}. \tag{8}$$

These solutions can also be expressed as

$$\boldsymbol{q} = \mathrm{e}^{\Lambda t} \boldsymbol{q}_0, \tag{9}$$

which, by direct substitution, is the solution to Eq. (5).

In general, the initial values in the  $q_0 = \{q_{p0}, q_{m0}\}^T$  vector are complex, which allows a phase lag to be incorporated into the solution. For simplicity, consider the case where the initial velocity  $\dot{x}(0) = 0$ , such that the phase lag can also be assumed to be zero, meaning that  $q_{p0} = q_{m0} = Q/2$ , where Q/2 is a real constant<sup>4</sup>. Therefore, using the  $\mathbf{x} = \Phi \mathbf{q}$  relationship (and Eq. (4)) to regain the expressions in terms of  $x_1$  and  $x_2$  we find

$$x = x_1 = Q \frac{(e^{\mathbf{i}\omega_n t} + e^{-\mathbf{i}\omega_n t})}{2} = x(0)\cos(\omega_n t), \text{ and}$$
  

$$\dot{x} = x_2 = -Q\omega_n \frac{(e^{\mathbf{i}\omega_n t} - e^{-\mathbf{i}\omega_n t})}{2\mathbf{i}} = -\omega_n x(0)\sin(\omega_n t),$$
(10)

where  $x(0) = x_0 = Q$  and  $\dot{x}(0) = \dot{x}_0 = 0$ . So by two steps, (i) writing the second-order system as an equivalent first-order system, and then (ii) using an eigenvalue coordinate transformation to diagonalise the *A* matrix, we have found a simplified version of the system that can be solved in a straightforward way using direct integration. An example of this solution is shown in Fig. 1 (b).

Solutions of the type in Eq. (8) are called the *base* solutions, and we note that other combinations of the base solutions, such as those in Eq. (10), are also solutions to the original equation. These type of base solutions will play an important role later on for nonlinear oscillators that have 'underlying' linear oscillator(s) — meaning if the nonlinear coefficients are all set to zero, the system reduces to a (set of) linear oscillator(s).

## 1.1.2 Solving the second-order equation directly

Finally for this example, we note that the same solutions can be found much more directly for the linear oscillator using an operator representation. To see this notice that Eq. (1) can be written as

$$\ddot{x} + \omega_n^2 x = 0 \equiv \left(\frac{\mathrm{d}}{\mathrm{d}t} - \mathbf{i}\omega_n\right) \left(\frac{\mathrm{d}}{\mathrm{d}t} + \mathbf{i}\omega_n\right) x = 0,\tag{11}$$

where the terms in brackets are to be thought of as operators on x. For Eq. (11) to be satisfied we have that

$$\left(\frac{\mathrm{d}}{\mathrm{d}t} - \mathbf{i}\omega_n\right)x_p = 0 \qquad \text{or} \qquad \left(\frac{\mathrm{d}}{\mathrm{d}t} + \mathbf{i}\omega_n\right)x_m = 0,$$
(12)

where the subscripts p and m have been added to distinguish between the two different solutions. We have just derived the solution to these type of scalar first order differential equations with a solution given by Eq. (8), such that there are two solutions

$$x_p = e^{\mathbf{i}\omega_n t} x_{p0} \qquad \text{and} \qquad x_m = e^{-\mathbf{i}\omega_n t} x_{m0},\tag{13}$$

assuming as before that  $t_0 = 0$ . In the general case the constants  $x_{p0}$  and  $x_{m0}$  are a complex conjugate pair

$$x_{p0} = \frac{X}{2} e^{\mathbf{i}\phi} \quad \text{and} \quad x_{m0} = \frac{X}{2} e^{-\mathbf{i}\phi}, \tag{14}$$

<sup>&</sup>lt;sup>3</sup>Note that many texts implicitly assume that  $t_0 = 0$ , but for some nonlinear and piecewise linear systems  $t_0$  must be retained through the analysis.

<sup>&</sup>lt;sup>4</sup>Note that we choose Q/2 rather than Q for convenience in forming the sine and cosine solutions.

*Journal of Structural Dynamics*, 1, (pp. 138-216) 2022 Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.



**Fig. 1:** (a) A schematic representation of a single degree-of-freedom point mass oscillator consisting of mass *m* kg and spring constant *k* N/m. (b) The cosine solution formed from the projection onto the real axis of the two complex vectors of the 'base' solutions. Note that in this case, the phase  $\phi$  can only be 0 or  $n\pi$  for n = 1, 2, 3... so we show the case for  $\phi = 0$ . (c) Solutions in state space (also called the *phase plane* for this type of system). Green arrows indicate time evolution of the trajectories. Expressions for the phase plane solutions for this system can be obtained by taking the ratio of the two velocity expressions in Eq. (2) to give  $\frac{x_2}{x_1} = \frac{dx_2/dt}{dx_1/dt} = \frac{dx_2}{dx_1} = -\frac{\omega_n^2 x_1}{x_2}$ . This can be rearranged to give  $-\omega_n^2 x_1 dx_1 = x_2 dx_2$ , which can then be integrated directly and, by incorporating the 1/2 factors into the constant of integration, the following expression is obtained  $-\omega_n^2 x_1^2 = x_2^2 - C \sim x_2^2 + \omega_n^2 x_1^2 = C$ . This is the equation of an ellipse, which has a size which depends on the constant *C*, and *C* in turn depends of the initial conditions,  $\mathbf{x}_0$ . As the constant *C* is arbitrary, an infinite family of periodic solutions exist which are solutions to the differential equation. In the case where  $\omega_n = 1$  the equation governing the motion in the phase plane becomes  $x_2^2 + x_1^2 = C$ , and in this case instead of ellipses the orbits would be circles.

where *X* is a real constant representing displacement amplitude and  $\phi$  represents the phase difference between the starting point on the sine wave solution and t = 0. The base solutions in Eq. (13) can be added (also using Eq. (14)) to give the general solution to Eq. (1) as

$$x = x_p + x_m = X \frac{(e^{i(\omega_n t + \phi)} + e^{-i(\omega_n t + \phi)})}{2} = X \cos(\omega_n t + \phi) = C_1 \cos(\omega_n t) + C_2 \sin(\omega_n t),$$
(15)

where  $C_1$  and  $C_2$  are constants to be determined. Now differentiating Eq. (15) and using the initial conditions, we can obtain the values of  $C_1$  and  $C_2$  and the general solution for Eq. (1) as

$$x = x(0)\cos(\omega_n t) + \frac{\dot{x}(0)}{\omega_n}\sin(\omega_n t),$$

$$\dot{x} = -\omega_n x(0)\sin(\omega_n t) + \dot{x}(0)\cos(\omega_n t).$$
(16)

In the case when  $\dot{x}(0) = 0$  then this reduces to the same as Eq. (10). An example of the family of periodic orbits that exist in the phase plane for this type of solution (with  $\dot{x}(0)=0$ ) is shown in Fig. 1 (c).

#### 1.1.3 Example 2: Small angle approximation for the simple pendulum

. ( )

A simple example of a nonlinear oscillator is the equation of motion governing the swing of a simple pendulum, shown in Fig. 2 (a) which has a governing equation of motion given by

$$\ddot{\theta} + \frac{g}{\ell}\sin\theta = 0,\tag{17}$$

where  $\theta$  is the angle of the pendulum from the downward rest position,  $\ell$  is the length of the pendulum and *g* is gravitational acceleration. Under certain conditions it is possible to find exact solutions using elliptic functions [132], but in many situations it is helpful to consider an approximate version of Eq. (17)<sup>5</sup>.

<sup>&</sup>lt;sup>5</sup>In 1892 Henri Poincaré published his work relating to the *three-body problem* from celestial mechanics [137, 7]. In this case, the governing differential equations were nonlinear, without any exact solution and Poincaré instead developed new ways to understand the dynamics without exact solutions. The pioneering work started by Poincaré eventually lead to the field of nonlinear dynamics (also called chaos theory, or dynamical systems theory), with a major early contribution from Birkhoff [11]. A short summary of the history of nonlinear dynamics can be found in Holmes [72]. More modern comprehensive overviews of the subject can be found, for example, in; Guckenheimer and Holmes [55], Glendinning [52], Thompson and Stewart [165], and Strogatz [162].

*Journal of Structural Dynamics*, 1, (pp. 138-216) 2022 Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.



**Fig. 2:** (a) A simple pendulum of length  $\ell$  m and mass *m* kg. (b) The conservative backbone curve obtained from Eq. (26) with parameter values  $\omega_n = 1$ ,  $\alpha = 0.25$  and  $U \in [0, 1]$ . Then when U = 0,  $\omega_r = \omega_n = 1$ , and as *U* increases,  $\omega_r < \omega_n$  which corresponds to *softening* behaviour. (c) Showing the forced-damped envelope curve (blue solid and dashed lines) and conservative backbone curve (red solid line) for the Duffing oscillator  $\ddot{x} + 2\zeta \omega_n \dot{x} + \omega_n^2 x + \alpha x^3 = F \cos(\Omega t)$ . Parameter values are  $\zeta = 0.0625$ ,  $\omega_n = 4$  rads/s,  $\alpha = 0.3$  m<sup>-2</sup>s<sup>-2</sup> and F = 0.016 N/kg. Note that the blue dashed line corresponds to unstable solutions. For a detailed discussion of the relationship between conservative backbone curves and forced-damped curves, and particularly the crossing point, see Cenedese and Haller [28, 27].

One way to simplify this equation is to consider the case when the angle of swing is small  $|\theta| \ll 1$ . Then we can approximate  $\sin \theta$  as a Taylor series so that  $\sin \theta \approx \theta - \frac{\theta^3}{31} + \frac{\theta^5}{51} + \dots$ , and then Eq. (17) becomes<sup>6</sup>

$$\ddot{\theta} + \omega_n^2 \theta - \alpha \theta^3 \approx 0, \tag{18}$$

where in this case  $\omega_n = \sqrt{\frac{g}{\ell}}$  and  $\alpha = \frac{g}{6\ell}$ . This equation is very similar to Eq. (1), except for the addition of the cubic nonlinear term  $-\alpha\theta^3$ . Eq. (18) is a version of the *Duffing equation*, named after Georg Duffing, which under certain conditions has exact solutions via elliptic integrals (see for example Chapter 4 in Kovacic and Brennan [96]). Despite the existence of exact solutions, the Duffing oscillator is widely used as an example for normal form analysis and many other approximate methods (as will be the case in this paper) because it is one of the 'simplest' systems to exhibit many of the key characteristics of nonlinear dynamics.

As in the previous example, it is possible to write Eq. (18) in an equivalent *first-order* form which can be achieved by defining  $x_1 = \theta$  and  $x_2 = \dot{\theta}$  so that<sup>7</sup>

$$\dot{x}_1 = x_2, \qquad \text{or} \qquad \dot{x} = f(x), \qquad \text{with} \qquad x = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}, \qquad (19)$$

where  $f(\cdot)$  on the right-hand side is used to denote a nonlinear function of the states,  $x_1$  and  $x_2$ .

Before seeking an (approximate) solution of a nonlinear equation like Eq. (19), an obvious question is: *Is there a coordinate transformation that simplifies this equation?* Ideally a coordinate transformation that eliminates the nonlinear terms would lead to a linear equation, which can be solved exactly, just like the linear oscillator example described above. Although it is possible to eliminate some types of nonlinear terms (see Example 10, Section 4.1.1), for Eq. (19) eliminating the cubic nonlinear terms (as we will describe in Example 11, Section 4.2.2) is not possible. However, the question still remains, as to whether there is a simpler form of the equation, which we call the *normal form*.

As we will describe in Section 4.2.2, the normal form for Eq. (19) can be obtained by first applying a diagonalising transformation  $x \to q$  (e.g. as we did above by using Eq. (4) to obtain Eq. (5)) and then a *near-identity* transformation  $q \to u$  up to third order (meaning including terms in the series solution up to cubic)<sup>8</sup>

<sup>&</sup>lt;sup>6</sup>This is based on the assumption that a truncated number of terms in the Taylor series will provide a sufficiently accurate approximation to the full solution when  $|\theta| \ll 1$ .

<sup>&</sup>lt;sup>7</sup>If required, at this stage substituting the arc-length relationship  $x = \ell \theta$  can be used to transform from angular to arc-length coordinates, and because  $|\theta| \ll 1$ , the arc-length will be close to the tangential linear displacement. In this case all parameters except  $\alpha$  remain the same, and the new value is given by  $\alpha = g/6\ell^3$ .

<sup>&</sup>lt;sup>8</sup>Note the classical method of removing nonlinear terms order-by-order in the differential equation and/or pushing them to a higher order can also be used and is described in Section 4.1.1.

which gives the normal form as [80]

$$\dot{u}_p = \mathbf{i}\omega_n u_p - \frac{3\mathbf{i}\alpha}{2\omega_r} u_p^2 u_m,$$
(20)

 $\dot{u}_m = -\mathbf{i}\omega_n u_m + \frac{3\mathbf{i}\alpha}{2\omega_r}u_p u_m^2$ , where  $\mathbf{u} = [u_p, u_m]^T$  and  $\omega_r$  is the nonlinear response frequency that is 'close' to<sup>9</sup> the undamped natural frequency

 $\omega_n$ . The associated approximate solution for displacement, x, (again up to third order) is given by

$$x = \left(U + \frac{3\alpha U^3}{16\omega_r^2}\right)\cos(\omega_r t) - \frac{\alpha U^3}{32\omega_r^2}\cos(3\omega_r t) + \dots$$
(21)

where dots are used to indicate that there are further terms in the series solution, and U is the amplitude of the base solutions<sup>10</sup> in terms of  $u_p$  and  $u_m$ , namely

$$u_p = \frac{U}{2} e^{\mathbf{i}\omega_r t}, \quad \text{and} \quad u_m = \frac{U}{2} e^{-\mathbf{i}\omega_r t}.$$
 (22)

However, normal form transformations are not unique, and an alternative approach that avoids the use of complex notation, as in Eq. (20), (and is therefore a type of *real* normal form, [104]) gives the normal form equation for the Duffing oscillator as [128]

$$\ddot{u} + \omega_n^2 u - 3\alpha (u_p^2 u_m + u_p u_m^2) = 0,$$
(23)

or alternatively, using the fact that  $u = u_p + u_m$ , this type of normal form can be expressed in terms of the base-solution coordinates

$$\begin{aligned} \ddot{u}_{p} + \omega_{n}^{2} u_{p} - 3\alpha u_{p}^{2} u_{m} &= 0, \\ \ddot{u}_{m} + \omega_{n}^{2} u_{m} - 3\alpha u_{p} u_{m}^{2} &= 0. \end{aligned}$$
(24)

In this case, the approximate solution for *x* is given by

$$x = U\cos(\omega_r t) - \frac{\alpha U^3}{32\omega_r^2}\cos(3\omega_r t) + \dots$$
(25)

Several observations can be made about these normal form equations and their approximate solutions:

- We are claiming that Eq. (20), Eq. (23) and Eq. (24) are all normal form expressions for this Duffing example. Details of the equivalence of these different normal form styles are given in Section 4.2.5. In fact, these are different *styles* of normal form that arise from the different approaches taken, and choices made during the transformation process<sup>11</sup>.
- 2. The concept of a normal form coordinate transformation making things 'simpler' in the nonlinear case is not always very helpful. It could be argued that Eq. (20), Eq. (23) and Eq. (24) are not simpler than the starting equations. However, as will be seen, they are more *useful*.
- 3. The approximate solutions Eq. (21) and Eq. (25) (the full derivations of which are given in Section 4.2.2 and Section 4.2.4 respectively) are not the same. This again is a consequence of the different approaches taken. In particular, there are *free functions*<sup>12</sup> in the process that can be chosen to give different outcomes. In this example, free function choices can be made so that Eq. (21) reduces to exactly the same as Eq. (25), if so required (an example of this is given in Section 4.4.2).

<sup>&</sup>lt;sup>9</sup>The expression for  $\omega_r$  will depend on the type of approximation (often called *detuning*) assumed in the normal form transformation. In many cases  $\omega_r \approx \omega_n$  is used in the second terms of the coefficient expressions of Eq. (20) and Eq. (21). Note also that some authors give just one of the normal form equations in Eq. (20) with the implication that the other is the conjugate expression [124, 85]. Detuning will be discussed further in Section 4.6.

<sup>&</sup>lt;sup>10</sup>Note, as this is an undamped single-degree-of-freedom oscillator phase  $\phi = 0$  has been assumed in the base solutions. To keep the formulations as simple as possible,  $\phi = 0$  will be used by default where appropriate in this paper. However cases for  $\phi \neq 0$  will be discussed in Section 3 and Section 5. Note also, that for nonlinear examples, these base solutions will be used extensively in this paper, but in general higher frequency solution terms could also be included, as is the case in the harmonic balance method.

<sup>&</sup>lt;sup>11</sup>Here we follow Murdock [120] who defines the final appearance of the normal form the *style*, and the approach taken to achieve it the *format*.

<sup>&</sup>lt;sup>12</sup>This is the terminology of [85]. In general these are choices of coefficient values in order to reduce certain terms in the final expressions. For example [85] describe some of the choices that reduce Eq. (21) to Eq. (25), and we shall discuss in more detail in Section 4.7.

- **4.** The normal form style based on ordinary differential equations in first-order form (i.e. Eq. (20)) is projected into the complex domain, whereas the styles which operate on ordinary differential equations in second-order form (i.e. Eq. (23) and Eq. (24)) remain real.
- **5.** The formats of Eq. (20) and Eq. (24) use 2N equations to represent a system with N degrees-of-freedom. However, the format of Eq. (23) uses N equations. This is useful when there are multiple degrees-of-freedom, and the 'compact' form of Eq. (23) means that there is no increase in the number of equations<sup>13</sup>

## 1.1.4 Summary and overview of the methodology

Overall, it is reasonable to ask; *What useful information can be obtained from the normal form?*. Well, for linear systems, coordinate transformations have been developed extensively as part of modal analysis, as will be discussed in Section 3. For nonlinear systems there are four things that can be potentially useful in structural dynamics:

- I. Firstly, for a limited number of simple nonlinear systems we can obtain an approximate solution for x, such as Eq. (21) and Eq. (25) <sup>14</sup>.
- **II.** Secondly, we obtain information on the *nonlinear resonances*, a topic that will be covered in more detail in Section 4.1.2
- **III.** Third we can get a frequency amplitude relationship(s) known as a *backbone curve*(s)<sup>15</sup>, which we introduce in Section 4.8, and these are one way to analyse the *nonlinear normal modes of a system*.
- **IV.** Fourth normal form transformations can be used to simplify models of important dynamic phenomena, such as bifurcations. We will describe how this relates to the Hopf bifurcation in Section 4.9.
- V. Finally, normal form transformations can be used as an integral part of other methods, notably model-order reduction and system identification. Although these topics are not discussed in detail in this paper, they are mentioned in Section 6.3 where we will discuss future research directions.

It should be noted that, apart from IV, all the other useful outcomes listed above can be (and usually are) obtained using other methods. Some comments giving advantages and disadvantages of normal form techniques compared with other methods are given in Section 6.2. We note here that normal forms are often regarded as being algebraically intense, and some authors have developed symbolic computation methods (particularly using Maple) in order to try and mitigate this problem [10, 190, 193, 122].

In order to obtain the useful outcomes listed in I. to V. above, we will use a two part methodology containing the steps:

- A. Use a coordinate transformation to 'simplify' the structure of the governing equations of motion based on a chosen normal form style and format, and
- **B.** Substitute an assumed solution into the normal form equations of motion to obtain some addition insights, such as the backbone curves mentioned in III above.

It is usually the case that we do A followed by B, but in some techniques the steps can be interlinked, as we will describe. It is also the case that the choice of the assumed solution can be significant, and so we will devote quite a lot of space in the paper to discussing this topic.

If we consider how this applies to the pendulum example given above (Section 1.1.3), firstly a coordinate transformation is used to obtain the normal form equations given by Eq. (24) (Step A). Then we can choose the base solutions given in Eq. (22) as the assumed solution, and substitute into either of the normal form expressions in Eq. (24) (Step B) to obtain

$$\omega_r^2 = \omega_n^2 - \frac{3\alpha U^2}{4} + \dots$$
(26)

This gives the backbone curve relationship (which is a truncated asymptotic series) relating the amplitude of the base solutions, U, and the response frequency  $\omega_r$ . An example is shown in Fig. 2 (b) for the case of softening nonlinearity, i.e. with  $\alpha$  negative, as in Eq. (26). One useful property of backbone curves is to represent the

<sup>&</sup>lt;sup>13</sup>There have also been attempts to develop methods that can be used for simultaneous normal form transformation and model-order reduction, and this topic is the subject of a recent review paper [173].

<sup>&</sup>lt;sup>14</sup>Where this is possible for nonlinear systems will be discussed in Section 4.

<sup>&</sup>lt;sup>15</sup>The use of the term "backbone curve" goes back at least as far as the PhD thesis of Rauscher [144]. The term "skeleton curve" is also sometimes used to mean the same thing.

amplitude-frequency dependence of the system, and as such are also a way visualising the *nonlinear normal modes*, a topic we will discuss further in Section 5.

Another useful property of backbone curves arises from the relationship between a backbone curve and the forced-damped response of the oscillator. An example of this is shown for a Duffing oscillator with a hardening nonlinear term (i.e. with  $\alpha$  positive) in Fig. 2 (c). Note that in Fig. 2 (both (b) and (c)) the backbone curves shown are so-called *conservative* in the sense that they are obtained from equations of motion without damping or forcing terms. In Fig. 2 (c) the forced-damped response is obtained by taking the maximum displacement value, *x*, per forcing period, and the blue solid curves show stable solutions, whereas the blue dashed curve relate to unstable solutions. The backbone curve in Fig. 2 (c) captures the hardening effect in the forced-damped response, and the two curves intersect very closely to the peak of the resonance (see [27] for a discussion of the relationship between these curves.).

The topic of backbone curves has been of great interest in the structural dynamics community in recent years — see for example [59, 39, 70, 159] (which is just a small selection of recent papers) and references therein. In fact, the relationship between backbone curves, and nonlinear resonances are arguably the most important application of normal forms (and other relevant methods) for the structural dynamicist. It is worth noting that for systems with damping, *damped backbone curves* have also been defined [97, 15, 5], and the area of identifying backbone curves is another topic of great recent interest [163, 112, 148, 149, 150, 90, 50, 53, 119, 173, 101, 119].

## 1.1.5 Structure of this paper

It should be noted that this paper is not intended to be a comprehensive review of the topic of normal forms — that can be found in [124, 120, 22, 116, 85], or structural dynamics, for that see the wide range of texts on the subject, such as [145, 146, 32, 167, 12, 185, 130, 166, 48, 189, 45, 64, 115, 123, 58, 76, 143, 183, 102] (to name just a selection) or nonlinear normal modes (the literature on this topic is already very large) see [152, 142, 14, 175, 155, 127, 103, 172, 3, 81, 8, 136, 170, 160, 2, 18, 88, 134, 135, 99, 65, 59, 39, 70, 60, 159, 177] and references therein<sup>16</sup>. Instead, the intention is to give an introductory treatment on using normal form transformations that relate to structural dynamics. As a result there will be selective coverage of topics that may be of interest to structural dynamicists working on nonlinear problems where normal form techniques might be useful. That said, the first part of the review deals with linear vibration problems. The reasons for this are (i) to cover selected normal form topics related to linear problems, (ii) to introduce preliminary material needed for the sections on nonlinear problems, and (iii) show a connection between the linear and nonlinear methods. However, readers that are primarily interested in the nonlinear examples may want to skip the sections on linear topics and go straight to the nonlinear parts.

The intended logic is to progress by starting with single degree-of-freedom linear systems then multi-degreeof-freedom linear systems, followed by single degree-of-freedom nonlinear systems, and finally multi-degree-offreedom nonlinear systems<sup>17</sup>. Throughout we also make a distinction between equations of motion written in second-order form (the classical approach for structural dynamics) and equations of motion written in first-order form (the classical approach for mathematics and dynamical systems theory).

Because linear systems have exact solutions, they can be tackled without using transformation approaches. Despite this, we hope to show that even if a linear system can be solved using a method exactly, without a transformation, there are still situations where it can be advantageous to use the transformation approach. As a result, the rest of this review paper is organised as follows. Firstly in Section 2 we examine single degree-of-freedom linear systems, including the unforced, undamped and unforced, damped linear oscillators. This Section starts with a review of some exact solutions methods, before considering the transformation approach. Then in Section 3, methods relating to multi-degree-of-freedom linear systems are described, where the main focus is on modal coordinate transformations. The topics relating to nonlinear systems start in Section 4, where near identity transformations are introduced in Section 4.2. In this Section a comparison is given between complex and real normal form styles. In addition, we cover topics such as derivation of the homological equations using Lie series, Section 4.3, the Hamiltonian normal form, Section 4.4, damping in normal form transformations, Section 4.5, frequency detuning Section 4.6, backbone curves, Section 4.8 and normal form transformations

<sup>&</sup>lt;sup>16</sup>Note that we are also not including control related topics in this review.

<sup>&</sup>lt;sup>17</sup>As this is a tutorial paper, we devote most space to single degree-of-freedom examples, and only show one multi-degree-of-freedom nonlinear example. For the reader that is most interested in multi-degree-of-freedom nonlinear applications multiple references to other sources are given.

of bifurcation phenomena Section 4.9. Then in Section 5, normal form transformations for coupled nonlinear oscillators are considered. Finally conclusions and future directions for research are given in Section 6.

## 1.1.6 Notation and terminology

In compiling this type of overview it is very difficult to maintain a consistent set of mathematical notation across all parts of the paper. The methodology used here is to use x,  $x_i$ , and  $\mathbf{x}$  for physical displacement coordinates (bold denoting a vector and non-bold a scalar). Then q,  $q_i$ , and  $\mathbf{q}$  for transformed coordinates of some type when the transformation is linear. Following that u,  $u_i$ , and  $\mathbf{u}$  for transformed coordinates of some type when the transformation is nonlinear.

However, we also need to use the state vector, x, which contains both displacement and velocity coordinates. In order to try and differentiate the state vector from the displacement vector we denote it as bold *and* italic x. This is a very subtle difference, but we hope it is sufficient to allow readers to distinguish between the two conventions. The same convention will be applied to the transformed coordinates q and u, which are both 2N long vectors.

Modal matrices will be applied for both the *N*-degree-of-freedom case and 2*N*-degree-of-freedom (state space) case. The convention adopted here is to use  $\Psi$  to represent the modal matrix in the *N*-degree-of-freedom case and  $\Phi$  for the 2*N*-degree-of-freedom case. The  $\ell^{\text{th}}$  individual modal vectors will then be given by  $\psi_{\ell}$  and  $\phi_{\ell}$  respectively. Likewise to distinguish between the (diagonal) eigenvector matrices,  $\hat{\Lambda}$  is used for the *N*-degree-of-freedom case. Note also that the  $\Box$  symbol is used to indicate the end point of the examples.

Finally, we will avoid the use of the term *canonical* in this review. This term has numerous meanings and interpretations across science and mathematics. For example, the phrases *canonical coordinates* and *canonical form* are used widely, but can have subtle differences in meaning depending on the exact context.

## 2 Single degree-of-freedom linear systems

## 2.1 The exponential matrix solution

In Section 1.1.1 we showed that, for the linear oscillator example, the solution to the equation in the form of  $\dot{q} = \Lambda q$ , (i.e. Eq. (5)) was given by  $q = e^{\Lambda t} q_0$  (i.e. Eq. (9)). In fact this relationship holds for any system of this type for which  $\Lambda$  is a diagonal matrix.

Furthermore, there is a more direct way of obtaining a solution which avoids the need to diagonalise the system matrix. For linear systems of this type with non-diagonal matrices, *A* (as in Eq. (2)) for the linear oscillator example), it is possible to show that the associated equation of motion,  $\dot{x} = Ax$  (i.e. Eq. (3)) has the solution

$$\boldsymbol{x} = \mathrm{e}^{At} \boldsymbol{x}_0 \tag{27}$$

without needing a transformation. This is because the term e<sup>At</sup> is an *exponential matrix* defined by

$$e^{At} = \sum_{k=0}^{\infty} \frac{1}{k!} A^k t^k = I + At + \frac{1}{2} A^2 t^2 + \dots$$
(28)

Now let's consider how this relates to the undamped linear oscillator example.

## Example 3: Undamped linear oscillator example revisited

We start by considering the undamped linear oscillator defined by Eq. (1) from Section 1.1.1. To make things as simple as possible, let us first consider the case when  $\omega_n = 1$  so that

$$A = \begin{bmatrix} 0 & 1\\ -1 & 0 \end{bmatrix},\tag{29}$$

then

$$\mathbf{e}^{At} = \left[ \begin{array}{cc} 1 & 0 \\ 0 & 1 \end{array} \right] + \left[ \begin{array}{cc} 0 & 1 \\ -1 & 0 \end{array} \right] t + \left[ \begin{array}{cc} -1 & 0 \\ 0 & -1 \end{array} \right] \frac{t^2}{2!} + \left[ \begin{array}{cc} 0 & -1 \\ 1 & 0 \end{array} \right] \frac{t^3}{3!} + \left[ \begin{array}{cc} 1 & 0 \\ 0 & 1 \end{array} \right] \frac{t^4}{4!} + \dots$$

Journal of Structural Dynamics, 1, (pp. 138-216) 2022 Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.

which, becomes

$$\mathbf{e}^{At} = \begin{bmatrix} 1 - \frac{t^2}{2!} + \frac{t^4}{4!} - \dots & t - \frac{t^3}{3!} + \frac{t^5}{5!} - \dots \\ -t + \frac{t^3}{3!} - \frac{t^5}{5!} + \dots & 1 - \frac{t^2}{2!} + \frac{t^4}{4!} - \dots \end{bmatrix} = \begin{bmatrix} \cos(t) & \sin(t) \\ -\sin(t) & \cos(t) \end{bmatrix}.$$
(30)

This then gives the solution of Eq. (2) as

$$\begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} \cos(t) & \sin(t) \\ -\sin(t) & \cos(t) \end{bmatrix} \begin{bmatrix} x_1(0) \\ x_2(0) \end{bmatrix} \rightsquigarrow \begin{array}{c} x = x(0)\cos(t) + \dot{x}(0)\sin(t), \\ \dot{x} = -x(0)\sin(t) + \dot{x}(0)\cos(t), \end{array}$$
(31)

which is familiar as the solution to the undamped, unforced oscillator.

If we set  $\dot{x}(0) = 0$  in Eq. (31), we obtain the same result as derived in Section 1.1.1, Eq. (10) (remembering that  $\omega_n = 1$ ). Of course, this is a highly simplified example in which the series solution for  $e^{At}$  is straightforward to compute. In the case of a more complicated *A* matrix, a more systematic approach can be applied using the Cayley-Hamilton theorem [89]. The Cayley-Hamilton theorem can be used to show that a square matrix satisfies its own characteristic equation. This in turn can be used to express all powers of  $A^k$  for  $k \ge 2$  as a linear combination of *A* and *I*. To demonstrate this, we consider once more the undamped linear oscillator.

#### Example 4: Undamped linear oscillator solution using the Cayley-Hamilton theorem

Let us consider Eq. (1) but now take the case when  $\omega_n \neq 1$  so that the A matrix becomes

$$A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & 0 \end{bmatrix}.$$
(32)

To find the characteristic equation we take  $|A - \lambda I| = 0$ , where  $\lambda$  are the eigenvalues of *A*, from which we find that

$$\lambda^2 + \omega_n^2 = 0,$$

and so  $\lambda_1 = i\omega_n$  and  $\lambda_2 = -i\omega_n$ . From the Cayley-Hamilton theorem we have that

$$A^2 + \omega_n^2 I = 0 \quad \rightsquigarrow \quad A^2 = -\omega_n^2 I.$$

Now we can use this to define the following relationships

$$\begin{split} A^{3} &= A^{2}A = -\omega_{n}^{2}IA \\ A^{4} &= A^{3}A = -\omega_{n}^{2}IA^{2} = -\omega_{n}^{2}I(-\omega_{n}^{2}I) = \omega_{n}^{4}I \\ A^{5} &= A^{4}A = \omega_{n}^{4}IA \\ A^{6} &= A^{5}A = \omega_{n}^{4}IA^{2} = \omega_{n}^{4}I(-\omega_{n}^{2}I) = -\omega_{n}^{6}I \\ & \cdots \\ A^{2k} &= (-1)^{k}\omega_{n}^{2k}I \\ A^{2k+1} &= (-1)^{k}\omega_{n}^{2k}A \end{split}$$

Then taking the series solution for e<sup>At</sup> and separating odd and even powers gives

$$e^{At} = \sum_{k=0}^{\infty} \frac{A^{k} t^{k}}{k!} = \sum_{k=0}^{\infty} \frac{A^{2k} t^{2k}}{2k!} + \sum_{k=0}^{\infty} \frac{A^{2k+1} t^{2k+1}}{(2k+1)!},$$
  
$$= I \sum_{k=0}^{\infty} \frac{(-1)^{k} \omega_{n}^{2k} t^{2k}}{2k!} + A \sum_{k=0}^{\infty} \frac{(-1)^{k} \omega_{n}^{2k} t^{2k+1}}{(2k+1)!},$$
  
$$= \cos(\omega_{n} t) I + \frac{1}{\omega_{n}} \sin(\omega_{n} t) A.$$
  
(33)

From this the solution to the governing equation can be expressed as

$$\begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} \cos(\omega_n t) & \frac{1}{\omega_n} \sin(\omega_n t) \\ -\omega_n \sin(\omega_n t) & \cos(\omega_n t) \end{bmatrix} \begin{bmatrix} x_1(0) \\ x_2(0) \end{bmatrix},$$
(34)

147 | doi:10.25518/2684-6500.84

 $\square$ 

where  $x_1(0) = x(0)$  and  $x_2(0) = \dot{x}(0)$ .

However, it is worth noting that there is a more efficient way to find this result. This is achieved by using the fact that the Cayley-Hamilton theorem also implies<sup>18</sup> that if  $\lambda$  is an eigenvalue of A then

$$e^{\lambda t} = \alpha(t)\lambda t + \beta(t) \quad \Leftrightarrow \quad e^{At} = \alpha(t)At + \beta(t)I,$$
(35)

where  $\alpha(t)$  and  $\beta(t)$  are time dependent coefficients to be determined. In this example *A* has two eigenvalues which gives two simultaneous equations

$$e^{\mathbf{i}\omega_n t} = \alpha(t)\mathbf{i}\omega_n t + \beta(t),$$
  

$$e^{-\mathbf{i}\omega_n t} = -\alpha(t)\mathbf{i}\omega_n t + \beta(t),$$
(36)

from which we can find directly that

$$\alpha(t) = \frac{e^{i\omega_n t} - e^{-i\omega_n t}}{2i\omega_n t} = \frac{1}{\omega_n t} \sin(\omega_n t),$$

$$\beta(t) = \frac{e^{i\omega_n t} + e^{-i\omega_n t}}{2} = \cos(\omega_n t),$$
(37)

and so

$$e^{At} = \alpha(t)At + \beta(t)I = \frac{1}{\omega_n}\sin(\omega_n t)A + \cos(\omega_n t)I,$$
(38)

which is exactly the same as Eq. (33).

We now consider the case when viscous damping is included.

#### Example 5: Damped, unforced linear oscillator

Now we will consider a damped, unforced linear oscillator given by

$$\ddot{x} + 2\zeta \omega_n \dot{x} + \omega_n^2 x = 0, \tag{39}$$

where  $\zeta = \frac{c}{2m\omega_n}$  is the damping ratio, *c* is the viscous damping coefficient (kg/s) and  $\omega_n$  is the natural frequency as previously defined<sup>19</sup>. As before we define the state vector  $\mathbf{x} = \{x_1, x_2\}^T$ , where  $x_1 = x$  is the displacement and  $x_2 = \dot{x}$  is the velocity. Using these definitions notice that  $\dot{x}_1 = x_2 = \dot{x}$ , and  $\dot{x}_2 = \ddot{x}$  which enables the system to be written in first-order form

$$\dot{x}_1 = x_2 \\ \dot{x}_2 = -2\zeta \omega_n x_2 - \omega_n^2 x_1, \quad \text{or} \quad \begin{bmatrix} \dot{x}_1 \\ \dot{x}_2 \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & -2\zeta \omega_n \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix},$$

which can be written in matrix form as Eq. (3), with solution given by Eq. (27). However, in this case the A matrix is more complicated, as it is given by

$$A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & -2\zeta\omega_n \end{bmatrix}.$$
 (40)

However, we can still apply the Cayley-Hamilton theory to this example without too much difficulty using Eq. (35). In this case the eigenvalues are complex conjugate  $\lambda_{1,2} = -\zeta \omega_n \pm i\omega_n \sqrt{1-\zeta^2} = -\zeta \omega_n \pm i\omega_d$ , where we are always assuming the underdamped case<sup>20</sup>, where  $\zeta < 1$ . Using Eq. (35) we obtain

$$e^{\lambda_1 t} = \alpha(t)\lambda_1 t + \beta(t),$$
  

$$e^{\lambda_2 t} = \alpha(t)\lambda_2 t + \beta(t),$$
(41)

<sup>19</sup>This can also be derived by starting with the equation of motion in the form  $m\ddot{x} + c\dot{x} + kx = 0$  and dividing through by m.

 $\Box$ 

<sup>&</sup>lt;sup>18</sup>This is because if the characteristic polynomial is  $p(\lambda) = \det(A - \lambda I) = 0$ , then the C-H theorem states that p(A) = 0. So an analytic function represented as a series expansion  $f(\lambda) = \sum_{k=0}^{\infty} \beta_k \lambda^k$  can be re-expressed as  $f(\lambda) = q(\lambda)p(\lambda) + r(\lambda)$ , where  $r(\lambda)$  is a residual polynomial of degree N - 1, and  $q(\lambda)p(\lambda)$  represents the rest of the terms in the series. Because  $p(\lambda) = 0$  this reduces to  $f(\lambda) = r(\lambda)$ . In this example  $f(\lambda t) = r(\lambda t) = \alpha(t)\lambda t + \beta(t)$ .

<sup>&</sup>lt;sup>20</sup>This is because we are interested in oscillations.

from which we can find directly that

$$\alpha(t) = \frac{e^{\lambda_1 t} - e^{\lambda_2 t}}{2\mathbf{i}\omega_d t} = \frac{1}{\omega_d t} e^{-\zeta \omega_n t} \sin(\omega_d t),$$

$$\beta(t) = \frac{e^{\lambda_1 t} + e^{\lambda_2 t} + \alpha_2 \zeta \omega_n t}{2} = e^{-\zeta \omega_n t} (\cos(\omega_d t) + \frac{\zeta \omega_n}{\omega_d} \sin(\omega_d t)),$$
(42)

and so

$$e^{At} = \alpha(t)At + \beta(t)I = e^{-\zeta \omega_n t} \frac{1}{\omega_d} \sin(\omega_d t)A + e^{-\zeta \omega_n t} (\cos(\omega_d t) + \frac{\zeta \omega_n}{\omega_d} \sin(\omega_d t))I,$$

$$= e^{-\zeta \omega_n t} \begin{bmatrix} \cos(\omega_d t) + \frac{\zeta \omega_n}{\omega_d} \sin(\omega_d t) & \frac{1}{\omega_d} \sin(\omega_d t) \\ \frac{-\omega_n^2}{\omega_d} \sin(\omega_d t) & \cos(\omega_d t) + \frac{-\zeta \omega_n}{\omega_d} \sin(\omega_d t) \end{bmatrix}$$
(43)

which when substituted into Eq. (27) gives the well known transient solution to the unforced, damped linear oscillator.  $\hfill \Box$ 

Note that for complex eigenvalues  $\lambda_{1,2} = a \pm ib$ , Eq. (43) can be written more compactly as

$$e^{At} = e^{at} \left( \frac{\sin(bt)}{b} (A - aI) + \cos(bt)I \right),$$

which is known as Putzer's spectral formula [140].

An alternative approach to dealing with the A matrix in the damped oscillator, Eq. (40) is to split<sup>21</sup> it into two giving

$$A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & -2\zeta\omega_n \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & 0 \end{bmatrix} + \begin{bmatrix} 0 & 0 \\ 0 & -2\zeta\omega_n \end{bmatrix},$$
(44)

which we will write as A = W + D. We might then hope to use the fact that

$$e^{At} = e^{(W+D)t} = e^{Wt}e^{Dt},$$
(45)

as we have already found an expression for  $e^{Wt}$  and  $e^{Dt}$  has a simple diagonal form. Unfortunately, this is only possible if WD - DW = 0. This is usually referred to as the commutator operation defined (for matrices) as

$$[W,D] = WD - DW, (46)$$

and if [W,D] = 0 then W and D are said to commute. It is left for the reader to convince themselves that for this example  $[W,D] \neq 0$  and therefore W and D do not commute.

However there is a solution in the case when W and D do not commute, which is provided by the Zassenhaus formula that is derived from the Baker–Campbell–Hausdorff formula [113, 25]

$$e^{At} = e^{(W+D)t} = e^{Wt} e^{Dt} e^{\frac{-t^2}{2} [W,D]} e^{\frac{t^3}{6} (2[W,[W,D]] + [D,[W,D]])} \dots$$
(47)

which is an infinite product of successive exponential terms with repeated commutator brackets. It should be clear that when [W,D] = 0 and all higher order brackets are also zero, then Eq. (47) reduces to Eq. (45).

Note that up until now we have not transformed A in order to find  $e^{At}$ . Of course if A is diagonal, then it is very easy to find  $e^{At}$  because if

$$A = \begin{bmatrix} a_{11} & 0 \\ 0 & a_{22} \end{bmatrix}, \text{ then } A^k = \begin{bmatrix} a_{11}^k & 0 \\ 0 & a_{22}^k \end{bmatrix}, \text{ and } e^A = \begin{bmatrix} e^{a_{11}} & 0 \\ 0 & e^{a_{22}} \end{bmatrix}.$$
 (48)

So an often used approach in structural dynamics analysis is to try and diagonalise *A*, just as we did in both the motivating examples in Section 1.1. Notice that the approach to diagonalise *A* in Section 1.1 led to complex diagonal matrices. This is natural for oscillatory solutions, but also somewhat more difficult to deal with than real matrices. As a result an alternative for oscillatory solutions that retains a real matrix structure is to use *Jordan normal form*, which will be considered next.

<sup>&</sup>lt;sup>21</sup>We use the term "splitting" some what loosely here, a more formal discussion of splitting methods can be found in [120].

## 2.2 Jordan normal form

For linear systems of the type defined by  $\dot{\mathbf{x}} = A\mathbf{x}$  ( i.e. Eq. (3)), a common method used to simplify the system is Jordan normal form<sup>22</sup>. Depending on the problem at hand, the simplified matrix will be either diagonal, upper triangular, or real. In the analysis of structural dynamics systems we are typically only interested in vibrations (i.e. underdamped systems) where the eigenvalues of *A* will be complex  $\lambda_{1,2} = a \pm ib$ , and therefore we only consider this case. We have already seen examples of the diagonal form with complex eigenvalues, e.g. in Example 1 Section 1.1.1, and so we will not consider that case again here. Instead we consider the other Jordan normal form that is relevant to underdamped linear systems with complex eigenvalues<sup>23</sup>.

Firstly, we assume that there is a transformation matrix, P, where x = Py such that

$$\dot{\mathbf{y}} = P^{-1}\dot{\mathbf{x}} = P^{-1}A\mathbf{x} = P^{-1}AP\mathbf{y} \tag{49}$$

and it can be shown, [52], that if  $\lambda_{1,2} = a \pm ib$  we can find *P* such that

$$P^{-1}AP = \begin{bmatrix} a & -b \\ b & a \end{bmatrix} = B,$$
(50)

such that

$$\dot{\mathbf{y}} = B\mathbf{y} \tag{51}$$

is the final transformed equation. Notice that *B* is not diagonal, but it is a real matrix.

This form allows us to avoid complex matrices, which would be the result if we transformed A to a diagonal eigenvalue matrix (as we did in Section 1.1.1). Notice that we can split B to give

$$B = \begin{bmatrix} 0 & -b \\ b & 0 \end{bmatrix} + \begin{bmatrix} a & 0 \\ 0 & a \end{bmatrix},$$
(52)

or as we used before B = W + D. Checking that [W,D] = 0 in this case, we see that *D* can be exponentiated using Eq. (48) and  $e^W$  can be found in the same way as for Eq. (32), which leads to

$$\mathbf{e}^{Bt} = \mathbf{e}^{Wt} \mathbf{e}^{Dt} = \begin{bmatrix} \cos(bt) & -\sin(bt) \\ \sin(bt) & \cos(bt) \end{bmatrix} \begin{bmatrix} \mathbf{e}^{at} & 0 \\ 0 & \mathbf{e}^{at} \end{bmatrix} = \mathbf{e}^{at} \begin{bmatrix} \cos(bt) & -\sin(bt) \\ \sin(bt) & \cos(bt) \end{bmatrix}.$$
(53)

We note that the solution to the transformed equation is  $y = e^{Bt}y_0$ , where  $y_0 = P^{-1}x_0$ . Because x = Py then  $x = Pe^{Bt}y_0 = Pe^{Bt}P^{-1}x_0$  with the result that by comparison with Eq. (27) we see that

$$e^{At} = P e^{Bt} P^{-1}. (54)$$

But what should we choose as *P*? Well we know that  $P^{-1}AP = B$  so that means that AP = PB and we can also exploit the fact that for a complex eigenvalue a + ib there is a corresponding complex eigenvector  $\mathbf{v}$  such that  $A\mathbf{v} = (a + ib)\mathbf{v}$  and because *A* is real  $A\operatorname{Re}(\mathbf{v}) = \operatorname{Re}((a + ib)\mathbf{v})$  and  $A\operatorname{Im}(\mathbf{v}) = \operatorname{Im}((a + ib)\mathbf{v})$ . So if we take  $P = [\operatorname{Im}(\mathbf{v}), \operatorname{Re}(\mathbf{v})]$  then we have

$$A[\operatorname{Im}(\mathbf{v}), \operatorname{Re}(\mathbf{v})] = [\operatorname{Im}((a+\mathbf{i}b)\mathbf{v}), \operatorname{Re}((a+\mathbf{i}b)\mathbf{v})],$$
  
$$= [a\operatorname{Im}(\mathbf{v}) + b\operatorname{Re}(\mathbf{v}), a\operatorname{Re}(\mathbf{v}) - b\operatorname{Im}(\mathbf{v})],$$
  
$$= [\operatorname{Im}(\mathbf{v}), \operatorname{Re}(\mathbf{v})] \begin{bmatrix} a & -b \\ b & a \end{bmatrix} = PB.$$
(55)

Now we consider how this analysis can be applied to the damped linear oscillator example.

<sup>&</sup>lt;sup>22</sup>Here we follow the approach taken by [52], but only discuss oscillatory systems that have complex eigenvalues.

<sup>&</sup>lt;sup>23</sup>Throughout this paper we primarily use the diagonal case with complex eigenvalues. An example of the real Jordan normal form is shown in Section 4.9.1. For those interested in other Jordan normal forms applied in the nonlinear case, discussions can be found in [16, 120, 124, 85].

## Example 6: Damped, unforced linear oscillator revisited

In the example of the damped linear oscillator we know that the eigenvalues are  $\lambda_{1,2} = -\zeta \omega_n \pm i\omega_d$ , where  $\omega_d = \omega_n \sqrt{1-\zeta^2}$  is the damped natural frequency. Therefore we have that  $a = -\zeta \omega_n$  and  $b = \omega_d$ . We also know that the eigenvectors are given by  $v_1 = \{1, \lambda_1\}^T$ , and  $v_2 = \{1, \lambda_2\}^T$ . The analysis above corresponds to a choice for *P* based on  $v_1$ , and so

$$P = [\operatorname{Im}(\mathbf{v}_1), \operatorname{Re}(\mathbf{v}_1)] = \begin{bmatrix} 0 & 1\\ b & a \end{bmatrix} = \begin{bmatrix} 0 & 1\\ \omega_d & -\zeta \omega_n \end{bmatrix} \quad \text{and} \quad P^{-1} = \begin{bmatrix} \frac{\zeta \omega_n}{\omega_d} & \frac{1}{\omega_d}\\ 1 & 0 \end{bmatrix}.$$
(56)

Next we can write down the *B* matrix

$$B = P^{-1}AP = \begin{bmatrix} -\zeta \omega_n & -\omega_d \\ \omega_d & -\zeta \omega_n \end{bmatrix}.$$
(57)

Now from Eq. (53) we have

$$e^{Bt} = e^{-\zeta \omega_n t} \begin{bmatrix} \cos(\omega_d t) & -\sin(\omega_d t) \\ \sin(\omega_d t) & \cos(\omega_d t) \end{bmatrix}.$$
(58)

Finally substituting Eq. (56) and Eq. (58) into Eq. (54) gives exactly the same solution as we derived using the Cayley-Hamilton methods, namely Eq. (43).  $\Box$ 

## 2.3 The method of "reduction of order"

At the end of the first motivating example of Section 1.1 we showed a method of solving the undamped, unforced linear oscillator equation  $(\ddot{x} + \omega_n^2 x = 0 \text{ i.e. Eq. (1)})$  directly using the idea of an operator formulation. Now we show how that can be combined with another normal form technique called *the method of reduction of order*<sup>24</sup>. To do this we revisit the damped linear oscillator again

## Example 7: Damped, unforced linear oscillator revisited again

In this method we apply a transformation where  $x = q(t)e^{-\zeta \omega_n t}$ . This leads to the relationships

$$\begin{aligned} x &= q(t)e^{-\zeta\omega_n t}, \\ \dot{x} &= \dot{q}e^{-\zeta\omega_n t} - \zeta\omega_n qe^{-\zeta\omega_n t}, \\ \ddot{x} &= \ddot{q}e^{-\zeta\omega_n t} - 2\zeta\omega_n \dot{q}e^{-\zeta\omega_n t} + \zeta^2\omega_n^2 qe^{-\zeta\omega_n t}. \end{aligned}$$

$$(59)$$

Substituting Eq. (59) into the equation of motion for the damped linear oscillator (Eq. (39)) gives the following outcome.

Start with Eq. (39): 
$$\ddot{x} + 2\zeta \omega_n \dot{x} + \omega_n^2 x = 0$$
, substituting Eq. (59)  $\rightarrow$   
 $\ddot{q}e^{-\zeta\omega_n t} - 2\zeta \omega_n \dot{q}e^{-\zeta\omega_n t} + \zeta^2 \omega_n^2 qe^{-\zeta\omega_n t} + 2\zeta \omega_n (\dot{q}e^{-\zeta\omega_n t} - \zeta \omega_n qe^{-\zeta\omega_n t}) + \omega_n^2 q(t)e^{-\zeta\omega_n t} = 0, \rightarrow$   
 $\ddot{q}e^{-\zeta\omega_n t} + q(\zeta^2 \omega_n^2 - 2\zeta^2 \omega_n^2 + \omega_n^2)e^{-\zeta\omega_n t} = 0, \rightarrow$   
 $\ddot{q} + \omega_d^2 q = 0,$  (60)

where  $\omega_d = \omega_n \sqrt{1 - \zeta^2}$  is the damped natural frequency. So now we see that the last equation in Eq. (60) is the transformed version of Eq. (39) and appears to be an "undamped" linear oscillator, but with a damped natural frequency,  $\omega_d$ .

Now applying the operator method from Section 1.1.2 to  $\ddot{q} + \omega_d^2 q = 0$ , we can obtain a solution for q(t) as

$$q(t) = q(0)\cos(\omega_d t) + \frac{\dot{q}(0)}{\omega_d}\sin(\omega_d t),$$
(61)

<sup>&</sup>lt;sup>24</sup>There is potential for confusion with the idea of reducing the order from  $d^2/dt^2$  terms to d/dt terms. In fact, the method in the form presented here can be interpreted as a variation on the WKB (Wentzel, Kramers & Brillouin — see for example [89, 9]) method, and is sometimes just called the "normal form" for example [161].

and so the complete solution,  $x = q(t)e^{-\zeta \omega_n t}$  is given by

$$x(t) = e^{-\zeta \omega_n t} (q(0) \cos(\omega_d t) + \frac{\dot{q}(0)}{\omega_d} \sin(\omega_d t)), \rightsquigarrow$$

$$x(t) = e^{-\zeta \omega_n t} (x(0) \cos(\omega_d t) + \frac{\dot{x}(0) + \zeta \omega_n x(0)}{\omega_d} \sin(\omega_d t)),$$
(62)

once the initial conditions are taken into consideration (i.e. by using the first equation in Eq. (62) and its derivative with t = 0).

This method gives a solution that is equivalent to the previous solutions for the damped, linear oscillator, but may not be particularly familiar to many structural dynamicists. It is notably shorter and more direct than the previous methods for this example.

## 2.4 Summary

The exponential matrix solution provides a direct way to develop solutions for linear single-degree-of-freedom oscillators, as described in Section 2.1. Alternatives, such as the Jordan normal form, can be used to obtain oscillatory solutions whilst maintaining real matrix expressions. A key part of this later method (and also required for some of the other methods discussed) is the computation of eigenvalues and eigenvectors. As is well known in structural dynamics, these can be related to resonant frequencies and mode shapes, and they form a key part of the discussion as we move on to talk about multi-degree-of-freedom linear systems.

## 3 Multi-degree-of-freedom linear systems

One of the most important topics in structural dynamics is the analysis of linear multi-degree-of-freedom systems. For a vector of discrete displacement coordinates,  $\mathbf{x} = [x_1, x_2, x_3, \dots, x_N]^T$  the equation of motion governing the (linear) dynamic behaviour is commonly written as

$$M\ddot{\mathbf{x}} + C\dot{\mathbf{x}} + K\mathbf{x} = \mathbf{F}_e,$$

where  $\mathbf{F}_e$  is the external forcing vector. Note that **x** *is not* the state vector<sup>25</sup>. The matrices *M*, *C* and *K* represent the mass, damping and stiffness properties of the structure respectively<sup>26</sup>.

The matrices M, C and K are not usually diagonal<sup>27</sup>, so the equations are coupled, and as a result it is useful to see if the system of equations can be simplified by applying a coordinate transformation. This transformation is called by several names, often a *modal transformation*, to signify the role that the *mode shapes* play<sup>28</sup>. The aim of the modal transformation is (if possible) to simultaneously diagonalise M, C and K resulting in N uncoupled second-order differential equations. All though it is not normally discussed in terms of normal form transformations, the modal transformation is a way of using a coordinate transformation to simplify the governing equations, and therefore fits exactly our definition of a normal form, assuming that we arrive at a system that can be considered to be in it's "simplest" or normal form.

As we noted in Section 1.1.3, coordinate transformations can be carried out with the governing differential equations in either second-order form or in first-order form. We will consider both cases.

## 3.1 The modal transformation for multi-degree-of-freedom linear systems

First consider the case when the governing differential equations are in second-order form (i.e. containing  $d^2/dt^2$  terms).

(63)

<sup>&</sup>lt;sup>25</sup>The state vector is bold *and* italic.

<sup>&</sup>lt;sup>26</sup>There are a large number of texts that deal with these type of linear vibration problems, see for example [115, 76]

<sup>&</sup>lt;sup>27</sup> In the analysis of vibration problems there are many specific structures of these matrices that arise, e.g., for gyroscopic systems, for which the total stiffness matrix is not symmetric but is a sum of symmetric and skew-symmetric matrices. See for example, [45]. In this paper we will consider only simple examples where the mass matrix is diagonal.

<sup>&</sup>lt;sup>28</sup>For a detailed description of mode shapes and related topics the interested reader is referred to specialist books on this, such as [45, 64].

## 3.1.1 The undamped, unforced case

As we are only interested in lightly damped systems it is useful to consider the case of the undamped, unforced system, first. When C = 0 and  $\mathbf{F}_e = 0$ , the governing equation becomes

$$M\ddot{\mathbf{x}} + K\mathbf{x} = 0 \quad \rightsquigarrow \quad \ddot{\mathbf{x}} + M^{-1}K\mathbf{x} = 0, \tag{64}$$

which can be treated as a linear matrix eigenvalue problem with an assumed solution of  $\mathbf{x} = \psi_j e^{i\omega_{nj}t}$  where  $\psi_j$  is the eigenvector (related to the mode-shape) and  $\omega_{nj}$  is the natural frequency (related to the eigenvalue) of the *j*<sup>th</sup> mode for the undamped, unforced system<sup>29</sup>.

Substituting the assumed solution into Eq. (64) leads to

$$\omega_{nj}^2 \psi_j = M^{-1} K \psi_j, \tag{65}$$

where  $\psi_j$  is the *j*th eigenvector and  $\omega_{nj}^2$  the corresponding *j*th eigenvalue of  $M^{-1}K$ . These eigenvectors and eigenvalues can be used to derive a coordinate transformation that replaces *M* and *K* with diagonal matrices. To do this, premultiplying by *M* and the  $\ell^{\text{th}}$  mode-shape, Eq. (65) can be rewritten as

$$\omega_{n\ell}^2 \psi_{\ell}^T M \psi_j = \psi_{\ell}^T K \psi_j \quad \text{or the equivalent expression} \quad \omega_{n\ell}^2 \psi_j^T M \psi_{\ell} = \psi_j^T K \psi_{\ell}. \tag{66}$$

Taking the transpose of this last equation and assuming that *M* and *K* are symmetric (such that  $M^T = M$  and  $K^T = K$ ) allows this expression to be rewritten in the following steps

$$\omega_{n\ell}^2 \psi_j^T M^T \psi_\ell = \psi_j^T K^T \psi_\ell, \rightsquigarrow \ \omega_{n\ell}^2 \psi_\ell^T (\psi_j^T M^T)^T = \psi_\ell^T (\psi_j K^T)^T, \rightsquigarrow \ \omega_{n\ell}^2 \psi_\ell^T M \psi_j = \psi_\ell^T K \psi_j, \tag{67}$$

by using the fact that  $M\psi_j = (\psi_i^T M^T)^T$ . Now subtracting the last of Eq. (67) from the first of Eq. (66) gives

$$(\boldsymbol{\omega}_{nj}^2 - \boldsymbol{\omega}_{n\ell}^2)\boldsymbol{\psi}_{\ell}^T \boldsymbol{M} \boldsymbol{\psi}_j = \boldsymbol{\psi}_{\ell}^T \boldsymbol{K} \boldsymbol{\psi}_j - \boldsymbol{\psi}_{\ell}^T \boldsymbol{K} \boldsymbol{\psi}_j = 0.$$
(68)

From this it can be seen that the system has the property of *orthogonality*, meaning that when  $\omega_{nj}^2 \neq \omega_{n\ell}^2$ , then  $\psi_{\ell}^T M \psi_i = 0$  for  $\ell \neq j$  must be true for Eq. (68) to hold.

This property is crucial to modal analysis for linear systems. It means that if we form a modal matrix  $\Psi = [\psi_1, \psi_2, ..., \psi_N]$ , then substituting  $\mathbf{x} = \Psi \mathbf{q}$  into Eq. (64) and pre-multiplying by  $\Psi^T$  leads to a transformed set of *decoupled* governing equations given by

$$\Psi^{T}\Psi\ddot{\mathbf{q}} + \Psi^{T}M^{-1}K\Psi\mathbf{q} = 0 \quad \rightsquigarrow \quad I\ddot{\mathbf{q}} + \hat{\Lambda}\mathbf{q} = 0 \tag{69}$$

where the diagonal matrix,  $\hat{\Lambda}$  contains the system eigenvalues, given by

$$\hat{\Lambda} = \begin{bmatrix} \omega_{n1}^2 & 0 & \dots & 0 & 0\\ 0 & \omega_{n2}^2 & \dots & 0 & 0\\ \vdots & \vdots & \vdots & \ddots & 0\\ 0 & 0 & \dots & 0 & \omega_{nN}^2 \end{bmatrix},$$
(70)

where  $\omega_{nj}$  for j = 1, 2, 3..., N are the natural frequencies of the system, and  $\Psi^T \Psi = I$  because  $\Psi$  is an orthogonal matrix. Note that the eigenvectors,  $\psi_j$  are not unique, and can be scaled — see for example the discussion in [13, 45]. For the purposes of our analysis, the scaling chosen must be such that  $\Psi^T M^{-1} K \Psi$  gives the result in Eq. (70).

## 3.1.2 Example 8: Modal transformation of an unforced, undamped three degree-of-freedom linear system

Consider the three degree-of-freedom oscillator shown in Fig. 3 (a) (with dampers  $c_{\ell} = 0$ ) for the case where the springs and masses are equal  $k_{\ell} = k$  and  $m_{\ell} = m$  for  $\ell = 1, 2, 3, 4$ .

<sup>&</sup>lt;sup>29</sup>Although in general both the eigenvalues and eigenvectors can be complex, in classical vibration theory, the nature of the underlying assumptions leads to *real* eigenvectors, which is why they are directly relatable to the mode-shapes of the system under consideration. If the eigenvectors are not real, then the system is said to have "complex modes".

The equation of motion of the system is given by

$$M\ddot{\mathbf{x}} + K\mathbf{x} = 0, \quad \text{where} \quad M = \begin{bmatrix} m & 0 & 0 \\ 0 & m & 0 \\ 0 & 0 & m \end{bmatrix} = mI, \quad K = \begin{bmatrix} 2k & -k & 0 \\ -k & 2k & -k \\ 0 & -k & 2k \end{bmatrix} = k \begin{bmatrix} 2 & -1 & 0 \\ -1 & 2 & -1 \\ 0 & -1 & 2 \end{bmatrix},$$

where *I* is a  $3 \times 3$  identity matrix. The eigenvalues and eigenvectors of  $M^{-1}K$  can be computed either numerically or by solving the characteristic equation. First we note that

$$M^{-1}K = \frac{k}{m} \begin{bmatrix} 2 & -1 & 0\\ -1 & 2 & -1\\ 0 & -1 & 2 \end{bmatrix} \rightsquigarrow \hat{\Lambda} = \frac{k}{m} \begin{bmatrix} 2 - \sqrt{2} & 0 & 0\\ 0 & 2 & 0\\ 0 & 0 & 2 + \sqrt{2} \end{bmatrix}, \text{ and } \Psi = \begin{bmatrix} \frac{1}{2} & -\frac{1}{\sqrt{2}} & -\frac{1}{2}\\ \frac{1}{\sqrt{2}} & 0 & \frac{1}{\sqrt{2}}\\ \frac{1}{2} & \frac{1}{\sqrt{2}} & -\frac{1}{2} \end{bmatrix},$$
(71)

for given values of *m* and *k*, where  $\hat{\Lambda}$  is the matrix of eigenvalues and  $\Psi = [v_1, v_2, v_1]$  is the matrix of eigenvectors,  $v_j$  for j = 1, 2, 3.<sup>30</sup>.

Making the transformation  $\mathbf{x} = \Psi \mathbf{q}$  decouples the system into

$$\ddot{\mathbf{q}} + \hat{\Lambda} \mathbf{q} = 0 \quad \rightsquigarrow \quad \begin{aligned} \ddot{q}_1 + \omega_{n_1}^2 q_1 &= 0, \\ \ddot{q}_2 + \omega_{n_2}^2 q_2 &= 0, \\ \ddot{q}_3 + \omega_{n_3}^2 q_2 &= 0, \end{aligned} \tag{72}$$

where  $\omega_{n1} = \sqrt{(k/m)(2-\sqrt{2})}$ ,  $\omega_{n2} = \sqrt{(2k/m)}$ , and  $\omega_{n3} = \sqrt{(k/m)(2+\sqrt{2})}$  are the natural frequencies of the system from  $\hat{\Lambda}$ . Each of these expressions can be solved exactly using (for example) Eq. (10). So, assuming the initial conditions are  $\mathbf{x} = [1 \text{ mm}, 0, 1 \text{ mm}]^T$  and  $\dot{\mathbf{x}} = [0, 0, 0]^T$  using  $\mathbf{q} = \Psi^{-1}\mathbf{x}$  gives  $q_1(0) = 1$ ,  $q_2(0) = 0$  and  $q_3(0) = -1$  and  $\dot{q}_1(0) = \dot{q}_2(0) = \dot{q}_3(0) = 0$ . This leads to

$$\begin{array}{ll} q_{1} = q_{1}(0)\cos(\omega_{n1}t) & x_{1} = \frac{1}{2}\cos(\omega_{n1}t) + \frac{1}{2}\cos(\omega_{n3}t), \\ q_{2} = q_{2}(0)\cos(\omega_{n2}t) & \text{then using} \quad \mathbf{x} = \Psi \mathbf{q} \quad \rightsquigarrow \quad x_{1} = \frac{1}{2}\cos(\omega_{n1}t) + \frac{1}{2}\cos(\omega_{n3}t), \\ x_{2} = \frac{1}{\sqrt{2}}\cos(\omega_{n1}t) - \frac{1}{\sqrt{2}}\cos(\omega_{n3}t), \\ x_{3} = \frac{1}{2}\cos(\omega_{n1}t) + \frac{1}{2}\cos(\omega_{n3}t). \end{array}$$

A similar approach can be applied to the case when damping is included (e.g. from Eq. (63) use  $M\ddot{x} + C\dot{x} + Kx = 0$ ). Here a simultaneous diagonalisation of the three matrices M, C and K is required, which puts addition restrictions onto the structure of the matrices, which is usually satisfied in a structural dynamics context by using *Rayleigh damping* so that  $C = \alpha M + \beta K$  is assumed<sup>31</sup>. In this case, the concept of *lambda matrices* also applies [105]. We now consider an alternative state space approach where damping is included.

## 3.2 State Space Form for Discrete Linear Systems

Now consider the case where the discrete system is written in first-order form. In this case, Eq. (63) (with  $\mathbf{F}_e = 0$ ) may be written in terms of the state vector  $\mathbf{x} = [\mathbf{x}^T, \dot{\mathbf{x}}^T]^T$  as<sup>32</sup>

$$\dot{\boldsymbol{x}} = A\boldsymbol{x} + \begin{bmatrix} 0_N \\ M^{-1}\mathbf{F}_e \end{bmatrix}, \quad A = \begin{bmatrix} 0_{NN} & I \\ -M^{-1}K & -M^{-1}C \end{bmatrix},$$
(73)

where  $0_{NN}$  is an  $N \times N$  matrix of zeros,  $0_N$  is a  $N \times 1$  zero vector and *I* is an  $N \times N$  identity matrix where *N* is the number of degrees-of-freedom. A coordinate transform for *x* is now required such that *A* can be replaced by a diagonal matrix such that the states become decoupled<sup>33</sup>. As was done in Section 3.1, we consider using the eigenvectors of *A* to create such a transformation.

<sup>&</sup>lt;sup>30</sup>Note that the eigenvectors can be scaled. Here, for simplicity, they are left unscaled. For a discussion on this type of scaling, such as mass normalised modes, see [51, 45].

<sup>&</sup>lt;sup>31</sup> In fact, there is a more general concept called *extended Rayleigh damping* in which this is just the first term of a series expansion [30]. There is also the related classical approach of Caughey and O'Kelly [26]. In addition, the whole subject of damping in structural dynamics is a large one, including many other types of damping models, see for example [106, 121, 82, 1].

<sup>&</sup>lt;sup>32</sup>There are alternative formulations of first-order state space that are used for nonlinear normal form transformation as described by Jain and Haller [79].

<sup>&</sup>lt;sup>33</sup>There are situations where Jordan normal form is preferable to diagonal matrices, but these are not considered here.



**Fig. 3:** (a) A three degree-of-freedom point mass oscillator consisting of masses  $m_{\ell}$ , dampers  $c_{\ell}$  and springs  $k_{\ell}$ . (Note that the rollers are assumed to be frictionless). (b) The exact solutions for the damped state space example (Section 3.2.1) given by equation Eq. (78). Parameter values  $m_{\ell} = 1.0$  kg, dampers  $c_{\ell} = 1.0$  kg/s and springs  $k_{\ell} = 10$  N/m. The initial conditions are given by  $\mathbf{x} = [1 \text{ mm}, 0, 1 \text{ mm}, 0, 0, 0]^T$ . Note that the response of masses 1 and 3 are identical.

Using an assumed solution of  $\mathbf{x} = \phi_{\ell} e^{\lambda_{\ell} t}$  where  $\phi_{\ell}$  is the eigenvector and  $\lambda_{\ell}$  is the associated eigenvalue, the damped, unforced (2*N* state space) system may be written as<sup>34</sup>

$$\lambda_{\ell}\phi_{\ell} = A\phi_{\ell}.\tag{74}$$

Using this expression, a modal matrix,  $\Phi$ , can be defined in which  $\phi_{\ell}$  forms the  $\ell^{\text{th}}$  column. The corresponding diagonal eigenvalue matrix is  $\Lambda$  where the  $\ell^{\text{th}}$  diagonal value is  $\lambda_{\ell}$ . Assuming that the eigenvectors are orthogonal, this allows the matrix relationship of Eq. (74) to be written as

$$\Phi \Lambda = A \Phi \quad \rightsquigarrow \quad \Lambda = \Phi^T A \Phi. \tag{75}$$

Now applying the coordinate transformation  $x = \Phi q$  to the *unforced* version of Eq. (73) gives

$$\Phi \dot{\boldsymbol{q}} = A \Phi \boldsymbol{q},$$

which when multiplied by  $\Phi^T$  combined with Eq. (75)) and the fact that  $\Phi^T \Phi = I$  (assuming orthogonal eigenvectors) gives

$$\dot{\boldsymbol{q}} = \Lambda \boldsymbol{q} \quad \rightsquigarrow \quad \boldsymbol{q} = \mathrm{e}^{\Lambda t} \boldsymbol{q}_0. \tag{76}$$

Because  $\Lambda$  is a diagonal matrix, the system is now decoupled with respect to the states (which include the displacements and velocities). Notice that damping is still included in the matrix, *A*, which will lead to complex eigenvalues and eigenvectors for underdamped vibrations<sup>35</sup>. An example is considered next.

#### 3.2.1 Example 9: State space eigenvector transformation of an unforced, damped three degree-offreedom linear system

We now consider the same example as in Section 3.1.2, the three degree-of-freedom oscillator shown in Fig. 3 (a), but with viscous damping now included. It is assumed that the springs, dampers and masses are equal  $k_{\ell} = k$ ,  $c_{\ell} = c$  and  $m_{\ell} = m$  for  $\ell = 1, 2, 3, 4$ , and that the damping is proportional to the stiffness via C = 0.1K. This leads to an unforced state space equation of motion of

$\dot{\boldsymbol{x}} = A\boldsymbol{x},$		0	0	0	1	0	0	]
	$A = \frac{k}{m}$	0	0	0	0	1	0	
		0	0	0	0	0	1	
		2	-1	0	0.2	-0.1	0	,
		-1	2	-1	-0.1	0.2	-0.1	
		0	-1	2	0	-0.1	0.2	

<sup>&</sup>lt;sup>34</sup>We note that it is also possible to extend methods developed in Section 2 to solve this problem.

<sup>&</sup>lt;sup>35</sup>Note this should not be confused with *complex modes* which arise for systems with non-proportional damping. See, [45] for a more detailed discussion.

where  $\mathbf{x} = [\mathbf{x}^T, \dot{\mathbf{x}}^T]^T$ . The state space formulation is decoupled using the eigenvalues and eigenvectors of *A* which are typically computed using numerical methods, and leads to expressions of the form

$$\Phi\Lambda = A\Phi, \ \Phi = \begin{bmatrix} v_1 & \overline{v}_1 & v_2 & \overline{v}_2 & v_3 & \overline{v}_3 \\ \lambda_1 v_1 & \overline{\lambda}_1 \overline{v}_1 & \lambda_2 v_2 & \overline{\lambda}_2 \overline{v}_2 & \lambda_3 v_3 & \overline{\lambda}_3 \overline{v}_3 \end{bmatrix}, \ \Lambda = \begin{bmatrix} \lambda_1 & 0 & 0 & 0 & 0 & 0 \\ 0 & \overline{\lambda}_1 & 0 & 0 & 0 & 0 \\ 0 & 0 & \lambda_2 & 0 & 0 & 0 \\ 0 & 0 & 0 & \overline{\lambda}_2 & 0 & 0 \\ 0 & 0 & 0 & 0 & \overline{\lambda}_3 & 0 \\ 0 & 0 & 0 & 0 & 0 & \overline{\lambda}_3 \end{bmatrix},$$
(77)

where the eigenvalues  $\lambda_{\ell}$  appear in complex conjugate pairs and the eigenvectors are related to those from Section 3.1.2, namely  $v_j$  for j = 1, 2, 3. from Eq. (71). Each of the  $\lambda_{\ell}$  eigenvalues and the conjugate can be expressed as

$$\frac{\lambda_{\ell} = -\zeta_{\ell} \omega_{n\ell} + \mathbf{i} \, \omega_{d\ell}}{\overline{\lambda}_{\ell} = -\zeta_{\ell} \omega_{n\ell} - \mathbf{i} \, \omega_{d\ell}}$$

where  $\zeta_{\ell}$  is the  $\ell$ th modal damping ratio, and  $\omega_{d\ell}$  is the associated damped modal frequency<sup>36</sup>. For example, the eigenvalues and eigenvectors obtained when k/m = 10 are listed in Table 1<sup>37</sup>.

Applying the transformation  $\mathbf{x} = \Phi \mathbf{q}$  (where  $\mathbf{q}$  is a  $4 \times 1$  vector) results in the decoupled equation  $\mathbf{\dot{q}} = \Lambda \mathbf{q}$  using Eq. (76). The initial conditions are  $\mathbf{x} = [1 \text{ mm}, 0, 1 \text{ mm}, 0, 0, 0]^T$  which can be transformed into the modal coordinates using  $\mathbf{q} = \Phi^{-1} \mathbf{x}$  resulting in

$$\boldsymbol{q}(0) = \begin{bmatrix} 0.5001 - \mathbf{i}0.0610\\ 0.5001 + \mathbf{i}0.0610\\ 0.0000 - \mathbf{i}0.0000\\ 0.0000 + \mathbf{i}0.0000\\ -0.5005 + \mathbf{i}0.1529\\ -0.5005 - \mathbf{i}0.1529 \end{bmatrix} = \begin{bmatrix} \frac{\mathcal{Q}_1}{2} e^{\mathbf{i}\varphi_1}\\ \frac{\mathcal{Q}_1}{2} e^{-\mathbf{i}\varphi_1}\\ 0\\ 0\\ \frac{\mathcal{Q}_3}{2} e^{\mathbf{i}\varphi_3}\\ \frac{\mathcal{Q}_3}{2} e^{-\mathbf{i}\varphi_3} \end{bmatrix},$$

where  $Q_1 = 1.0076$ ,  $Q_3 = 1.0468$ ,  $\varphi_1 = -0.1213$  and  $\varphi_3 = 2.8451$ . Now using Eq. (76) with the initial conditions gives

$$\begin{aligned} q_1 &= q_1(0) e^{\lambda_1 t} = \frac{Q_1}{2} e^{\mathbf{i} \phi_1} e^{\lambda_1 t} = \frac{Q_1}{2} e^{-\zeta_1 \omega_{n1} t} e^{\mathbf{i} (\omega_{d1} + \phi_1)}, \\ q_2 &= q_2(0) e^{\overline{\lambda}_1 t} = \frac{Q_1}{2} e^{-\mathbf{i} \phi_1} e^{\overline{\lambda} t} = \frac{Q_1}{2} e^{-\zeta_1 \omega_{n1} t} e^{-\mathbf{i} (\omega_{d1} + \phi_1)}, \\ q_3 &= q_3(0) e^{\lambda_2 t} = 0, \\ q_4 &= q_4(0) e^{\overline{\lambda}_2 t} = 0, \\ q_5 &= q_5(0) e^{\lambda_3 t} = \frac{Q_3}{2} e^{\mathbf{i} \phi_3} e^{\lambda_3 t} = Q_3 e^{-\zeta_3 \omega_{n3} t} e^{\mathbf{i} (\omega_{d3} + \phi_3)}, \\ q_6 &= q_6(0) e^{\overline{\lambda}_3 t} = \frac{Q_3}{2} e^{\mathbf{i} \phi_1} e^{\overline{\lambda}_3 t} = Q_3 e^{-\zeta_3 \omega_{n3} t} e^{-\mathbf{i} (\omega_{d3} + \phi_3)}. \end{aligned}$$

Then using the modal matrix relationship  $x = \Phi q$  gives the displacements as

$$x_{1} = v_{11}Q_{1}e^{-\zeta_{1}\omega_{n1}t}\cos(\omega_{d1}t + \varphi_{1}) + v_{31}Q_{3}e^{-\zeta_{3}\omega_{n3}t}\cos(\omega_{d3}t + \varphi_{3}),$$
  

$$x_{2} = v_{12}Q_{1}e^{-\zeta_{1}\omega_{n1}t}\cos(\omega_{d1}t + \varphi_{1}) + v_{32}Q_{3}e^{-\zeta_{3}\omega_{n3}t}\cos(\omega_{d3}t + \varphi_{3}),$$
  

$$x_{3} = v_{13}Q_{1}e^{-\zeta_{1}\omega_{n1}t}\cos(\omega_{d1}t + \varphi_{1}) + v_{33}Q_{3}e^{-\zeta_{3}\omega_{n3}t}\cos(\omega_{d3}t + \varphi_{3}).$$
(78)

<sup>&</sup>lt;sup>36</sup>Note that these expressions can be used to show that  $\zeta_{\ell} = 1/((\text{Re} \lambda_{\ell}/\text{Im} \lambda_{\ell})^2 + 1)$ .

<sup>&</sup>lt;sup>37</sup>Note that most numerical routines do not automatically return the eigenvalues or eigenvectors in the format of Eq. (77). The ordering convention is that the eigenvalues are sorted based on the  $\omega_{d\ell}$  values with the smallest first. The eigenvectors are ordered accordingly, but also typically need scaling to obtain the format of Eq. (77). Each eigenvector needs to be scaled by the appropriate complex and/or real constant(s) in order to achieve this.

l	$\lambda_\ell$	$v_{1\ell}$	$v_{2\ell}$	$v_{3\ell}$
1	-0.293 + i2.403	0.4999 + i 0.0000	-0.7070 + i0.0000	-0.4995 + i0.0000
2	-1.000 + i4.359	0.7070 + i0.0000	0.0000 + i0.0000	0.7064 - i0.0000
3	-1.707 + i5.588	0.4999 + i0.0000	0.7070 - i0.0000	-0.4995 + i0.0000
4		-0.1464 + i 1.2011	0.7070-i3.0819	0.8527 – <b>i</b> 2.7912
5		-0.2071 + i1.6986	-0.0000 + i0.0000	-1.2058 + i 3.9473
6		-0.1464 + i 1.2011	-0.7070 + i 3.0819	0.8527 – <b>i</b> 2.7912

**Table 1:** Eigenvalues and eigenvectors for damped state space example with k/m = 10, and C = 0.1K.

An example set of simulation results are shown in Fig. 3 (b). Here the exponential decay of the damped, unforced solutions can be seen for the displacements of the three masses.

## 3.3 Summary

Examples 8 and 9 show several important (and well known) features that are worth commenting on:

- 1. Both transformations have reduced the equations of motion to a series of uncoupled linear oscillators for which the exact solutions were derived in Section 1.1.1 and Section 2.
- **2.** The exact modal solutions found in **q** coordinates can be transformed into **x** coordinates using  $\mathbf{x} = \Psi \mathbf{q}$  or  $\mathbf{x} = \Phi \mathbf{q}$ , such that the **x** displacements can be found as the sum of the modal contributions, which is sometimes referred to as *superposition* of solutions.
- **3.** The modal matrix is independent of time, and so it is straightforward to find the inverse relationship  $\mathbf{q} = \Psi^{-1}\mathbf{x}$  or  $\mathbf{q} = \Phi^{-1}\mathbf{x}$  when required (or if the eigenvectors are orthogonal  $\mathbf{q} = \Psi^T \mathbf{x}$  or  $\mathbf{q} = \Phi^T \mathbf{x}$ ). For example, if initial conditions are given in terms of the  $\mathbf{x}$  need to be transformed into the modal domain.

Although these are well known results, they will be helpful in formulating an approach to nonlinear coordinate transformations in Section 4.

## 4 Normal form transformations of nonlinear systems

## 4.1 Introduction

The ultimate aim of this paper is to use normal form transformations to help understand the dynamics of multi-degree-of-freedom nonlinear oscillators<sup>38</sup> defined by

$$M\ddot{\mathbf{x}} + C\dot{\mathbf{x}} + K\mathbf{x} + \mathscr{N}(\mathbf{x}, \dot{\mathbf{x}}) = 0, \tag{79}$$

where *M*, *C* and *K* are mass, damping and stiffness matrices and the nonlinear terms  $\mathscr{N}$  are assumed (in general) to be a function of the displacement and velocity vectors. Using the linear modes  $\mathbf{x} = \Psi \mathbf{q}$ , as defined in Section 3.1, a modal transformation can be applied to this expression which leads to

$$\ddot{\mathbf{q}} + D\dot{\mathbf{q}} + \hat{\Lambda}\mathbf{q} + \mathbf{N}_q(\mathbf{q}, \dot{\mathbf{q}}) = 0, \tag{80}$$

where  $\mathbf{N}_q(\mathbf{q}, \dot{\mathbf{q}}) = \Psi^T \mathscr{N}(\Psi \mathbf{q}, \Psi \dot{\mathbf{q}})$  and  $D = \alpha I + \beta \hat{\Lambda}$  is the diagonal proportional damping matrix (see Section 3). Eq. (79) can also be written as an equivalent state space system of first-order differential equations

$$\dot{\boldsymbol{x}} = A\boldsymbol{x} + \tilde{\mathcal{N}}(\boldsymbol{x}) + \boldsymbol{F},$$

where

$$A = \begin{bmatrix} 0 & I \\ -M^{-1}K & -M^{-1}C \end{bmatrix}, \quad \tilde{\mathcal{N}} = \left\{ \begin{array}{c} 0 \\ -M^{-1}\mathcal{N}(\mathbf{x}) \end{array} \right\},$$

<sup>&</sup>lt;sup>38</sup>Note also that we do not consider normal form transformations of forced nonlinear systems in this paper. For an introduction to that topic see [85, 128].

where  $\tilde{\mathcal{N}}(\mathbf{x})$  is obtained from  $\mathcal{N}(\mathbf{x}, \dot{\mathbf{x}})$  via the relationship  $\mathbf{x} = [\mathbf{x}, \dot{\mathbf{x}}]^T$ . Now transforming the state space equation using the linear modal transformation  $\mathbf{x} = \Phi \mathbf{q}$  (as defined in Section 3.2) leads to

$$\dot{\boldsymbol{q}} = \Lambda \boldsymbol{q} + \boldsymbol{N}(\boldsymbol{q}), \tag{81}$$

where  $N(q) = \Phi^{-1} \tilde{\mathcal{N}}(\Phi q)$ .

In this Section we will introduce a general approach to multi-degree-of-freedom nonlinear oscillators, but we will limit the examples to just single degree-of-freedom for now<sup>39</sup>. Examples of coupled (i.e. more than one-degree-of-freedom) nonlinear oscillators are algebraically intense and lengthy, and for this reason we delay discussion of them until Section 5. To get started we discuss some of the key topics relating to normal form transformations, which are motivated with the following example.

# 4.1.1 Example 10: the real normal form for an undamped oscillator that has both quadratic and cubic nonlinear terms

Consider an undamped oscillator that has both quadratic and cubic nonlinear terms written as<sup>40</sup>

$$\ddot{q} + \omega_n^2 q + \beta q^2 + \alpha q^3 = 0$$
, or  $\ddot{q} + \omega_n^2 q + n^{(2)}(q) + n^{(3)}(q) = 0$ , (82)

where  $\omega_n$  is the natural frequency,  $n^{(2)}(q) = \beta q^2$  denotes the quadratic nonlinear terms and  $n^{(3)}(q) = \alpha q^3$  denotes the cubic nonlinear terms. In addition,  $\beta$  and  $\alpha$  are nonlinear coefficients (and in general we write  $n^{(k)}(q)$  for the *k*th oder term). We want to find a nonlinear coordinate transformation from coordinate *q* to a new coordinate *u*, and if possible obtain a simplified system of the form

$$\ddot{u} + \omega_n^2 u + n_u^{(2)}(u) + n_u^{(3)}(u) = 0,$$
(83)

where  $n_u^{(2)}(u)$  denotes the simplified quadratic nonlinear terms and  $n_u^{(3)}(u)$  denotes the simplified cubic nonlinear terms. In addition, the transformed system has linear part equivalent to

$$\begin{bmatrix} \dot{u}_p \\ \dot{u}_m \end{bmatrix} = \begin{bmatrix} \mathbf{i}\omega_n & 0 \\ 0 & -\mathbf{i}\omega_n \end{bmatrix} \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \mathcal{O}(2), \tag{84}$$

where  $u = u_p + u_m$ . Next we make a coordinate transformation of the form

$$q = u + h^{(2)}(u, \dot{u}) + h^{(3)}(u, \dot{u}) + \dots \quad \rightsquigarrow \quad q = u_p + u_m + h^{(2)}(u_p, u_m) + h^{(3)}(u_p, u_m) + \dots$$
(85)

where the dots are used to denote that the asymptotic series expansion has higher order terms, and we have included up to quadratic and cubic terms to match the nonlinear terms in Eq.  $(82)^{41}$ . Substituting the first of Eq. (85) into the second of Eq. (82) and replacing  $\ddot{u}$  using Eq. (83) gives

$$\omega_n^2 u + n_u^{(2)} + n_u^{(3)} = \frac{d^2}{dt^2} (h^{(2)} + h^{(3)}) + \omega_n^2 u + \omega_n^2 (h^{(2)} + h^{(3)}) + n^{(2)} (u + h^{(2)} + h^{(3)}) + n^{(3)} (u + h^{(2)} + h^{(3)}),$$
(86)

so that taking the *u* terms order by order

Order 
$$k = 1$$
:  $\omega_n^2 u = \omega_n^2 u$ ,  
Order  $k = 2$ :  $n_u^{(2)}(u) = \frac{d^2 h^{(2)}(u)}{dt^2} + \omega_n^2 h^{(2)}(u) + n^{(2)}(u)$ ,  
Order  $k = 3$ :  $n_u^{(3)}(u) = \frac{d^2 h^{(3)}(u)}{dt^2} + \omega_n^2 h^{(3)}(u) + n^{(2)}_{(3)}(u + ...) + n^{(3)}(u + ...)^{(3)}$ ,  
Order  $k = 4$ : ...  
:

<sup>&</sup>lt;sup>39</sup>This allows us to neglect phase lag in this Section, and thus keep the examples as simple as possible. It should be noted however, that in many applications, including a phase lag may be required. We will introduce phase lag in Section 5.

<sup>&</sup>lt;sup>40</sup>For single-degree-of-freedom examples in second order form there is no modal transformation so we take x = q at the start.

<sup>&</sup>lt;sup>41</sup>Note also that the  $h^{(k)}$  terms are functions of both u and  $\dot{u}$  in general, where  $\dot{u} = i\omega_n(u_p - u_m)$ .

where the  $n_{(3)}^{(2)}(u+...)$  term represents the cubic (order k = 3 terms) generated when  $q = u + h^{(2)} + ...$  is substituted into the  $n^{(2)}$  nonlinear terms, and  $n^{(3)}(u+...)^{(3)}$  represents the cubic (order k = 3) terms generated when  $q = u + h^{(2)} + ...$  is substituted into the  $n^{(3)}$  term. So now we can rearrange Eq. (87) for the order k = 2 and order k = 3 expressions to obtain *homological equations* as

Order 
$$k = 2:$$
  $-\frac{d^2 h^{(2)}}{dt^2} - \omega_n^2 h^{(2)} = n^{(2)} - n_u^{(2)},$   
Order  $k = 3:$   $-\frac{d^2 h^{(3)}}{dt^2} - \omega_n^2 h^{(3)} = n_{(3)}^{(2)} + n^{(3)} - n_u^{(3)},$ 
(88)

which can be considered in turn.

So, let's consider the order k = 2 homological equation first. We anticipate from previous studies that the quadratic terms in this example can be eliminated, so we first try setting  $n_u^{(2)} = 0$  in the k = 2 homological equation in Eq. (88). Then we take  $u = u_p + u_m$  and expand the  $h^{(2)}$  term to mirror the expansion of the  $n^{(2)}$  nonlinear term which gives

$$h^{(2)}(u_p, u_m) = \beta (u_p^2 + 2u_p u_m + u_m^2), \quad \text{and} \quad h^{(2)}(u_p, u_m) = b_1 u_p^2 + b_2 u_p u_m + b_3 u_m^2, \tag{89}$$

where the  $b_j$  are coefficients that are yet to be identified. Substituting this into the order k = 2 homological equation with  $n_u^{(2)} = 0$  gives

$$-\frac{\mathrm{d}^2}{\mathrm{d}t^2}(b_1u_p^2 + b_2u_pu_m + b_3u_m^2) - \omega_n^2(b_1u_p^2 + b_2u_pu_m + b_3u_m^2) = \beta(u_p^2 + 2u_pu_m + u_m^2) + \mathcal{O}(3).$$
(90)

In order to compute the  $\frac{d^2}{dt^2}$  derivatives we use the chain rule and expressions in Eq. (84)<sup>42</sup>. For example for the  $u_p^2$  term

$$\frac{\mathrm{d}}{\mathrm{d}t}(u_p^2) = \frac{\partial \left(u_p^2\right)}{\partial u_p}\dot{u}_p = 2u_p\mathbf{i}\omega_n u_p = 2\mathbf{i}\omega_n u_p^2, \quad \rightsquigarrow \frac{\mathrm{d}^2}{\mathrm{d}t^2}(u_p^2) = \frac{\mathrm{d}}{\mathrm{d}t}\left(\frac{\mathrm{d}}{\mathrm{d}t}(u_p^2)\right) = 2\mathbf{i}\omega_n\frac{\partial \left(u_p^2\right)}{\partial u_p}\dot{u}_p = -4\omega_n^2 u_p^2,$$
and similarly
$$\frac{\mathrm{d}^2}{\mathrm{d}t^2}(u_p u_m) = \frac{\partial \left(u_p u_m\right)}{\partial u_p}\dot{u}_p + \frac{\partial \left(u_p u_m\right)}{\partial u_m}\dot{u}_m = 0 \quad \text{and} \quad \frac{\mathrm{d}^2}{\mathrm{d}t^2}(u_m^2) = -4\omega_n^2 u_m^2,$$
(91)

which when substituted into Eq. (90) leads to

$$4b_1\omega_n^2u_p^2 + 4b_3\omega_n^2u_p^2 - \omega_n^2b_1u_p^2 - \omega_n^2b_2u_pu_m - \omega_n^2b_3u_p^2 = \beta u_p^2 + \beta 2u_pu_m + \beta u_m^2,$$
(92)

and by comparing coefficients we obtain the relationships

$$b_{1} 3\omega_{n}^{2} = \beta$$

$$-\omega_{n}^{2}b_{2} = 2\beta$$

$$b_{3} 3\omega_{n}^{2} = \beta$$

$$b_{3} = \frac{\beta}{3\omega_{n}^{2}}$$

$$b_{2} = -2\frac{\beta}{\omega_{n}^{2}}$$

$$b_{3} = \frac{\beta}{3\omega_{n}^{2}}$$
(93)

which confirms that the assumption of  $n_u^{(2)} = 0$  is correct. Substituting the coefficient values from Eq. (93) into Eq. (85) gives the coordinate transformation for linear and quadratic terms as

$$q = u_p + u_m + \frac{\beta}{3\omega_n^2} u_p^2 - 2\frac{\beta}{\omega_n^2} u_p u_m + \frac{\beta}{3\omega_n^2} u_m^2 + \mathcal{O}(3).$$
(94)

Next we need to compute the k = 3 homological equation to deal with the cubic terms. As before we take  $u = u_p + u_m$  and expand the  $h^{(3)}$  term to mirror the expansion of the  $n^{(3)}$  nonlinear term so that

$$n^{(3)}(u_p, u_m) = \alpha(u_p^3 + 3u_p^2 u_m + 3u_p u_m^2 + u_m^2), \quad \text{and} \quad h^{(3)}(u_p, u_m) = b_4 u_p^3 + b_5 u_p^2 u_m + b_6 u_p u_m^2 + b_7 u_m^3, \tag{95}$$

<sup>&</sup>lt;sup>42</sup>This will be formalised in Section 4.2 as a computation of the *Lie derivative*. Note also, that the use of Eq. (84) truncated at  $\mathcal{O}(2)$  means that this is a first order approximation of the time derivatives.

where the  $b_j$  are coefficients that are yet to be identified. For the cubic terms we cannot (at this stage) assume that the  $n_u^{(3)}$  is zero, and we also have to account for the  $n_{(3)}^{(2)}$  in Eq. (88). Both of these terms are also expanded to mirror the  $n^{(3)}$  nonlinear term which gives

$$n_{(3)}^{(2)}(u_p, u_m) = \frac{2\beta^2}{3\omega_n^2}(u_p^3 - 5u_p^2 u_m - 5u_p u_m^2 + u_m^2), \quad \text{and} \quad n_u^{(3)}(u_p, u_m) = n_{u1}^* u_p^3 + n_{u2}^* u_p^2 u_m + n_{u3}^* u_p u_m^2 + n_{u4}^* u_m^3, \quad (96)$$

where the  $n_{uj}^*$  are coefficients that are yet to be identified. Now we substitute all of these relationships into the k = 3 homological equation from Eq. (88) to give

$$-\frac{d^{2}}{dt^{2}}(b_{4}u_{p}^{3}+b_{5}u_{p}^{2}u_{m}+b_{6}u_{p}u_{m}^{2}+b_{7}u_{m}^{3})-\omega_{n}^{2}(b_{4}u_{p}^{3}+b_{5}u_{p}^{2}u_{m}+b_{6}u_{p}u_{m}^{2}+b_{7}u_{m}^{3})$$

$$=\frac{2\beta^{2}}{3\omega_{n}^{2}}(u_{p}^{3}-5u_{p}^{2}u_{m}-5u_{p}u_{m}^{2}+u_{m}^{2})+\alpha(u_{p}^{3}+3u_{p}^{2}u_{m}+3u_{p}u_{m}^{2}+u_{m}^{2})-(n_{u1}^{*}u_{p}^{3}+n_{u2}^{*}u_{p}^{2}u_{m}+n_{u3}^{*}u_{p}u_{m}^{2}+n_{u4}^{*}u_{m}^{3}).$$
(97)

Using the chain rule and expressions in Eq. (84) gives

$$-\frac{d^2}{dt^2}(b_4u_p^3 + b_5u_p^2u_m + b_6u_pu_m^2 + b_7u_m^3) = -9\omega_n^2b_4u_p^3 - \omega_n^2b_5u_p^2u_m - \omega_n^2b_6u_pu_m^2 - 9\omega_n^2b_7u_m^3,$$
(98)

which when substituted into Eq. (97) gives

12

$$(8\omega_n^2)b_4u_p^3 + (8\omega_n^2)b_7u_m^3 = \frac{2\beta^2}{3\omega_n^2}(u_p^3 - 5u_p^2u_m - 5u_pu_m^2 + u_m^2) + \alpha(u_p^3 + 3u_p^2u_m + 3u_pu_m^2 + u_m^2) - (n_{u1}^*u_p^3 + n_{u2}^*u_p^2u_m + n_{u3}^*u_pu_m^2 + n_{u4}^*u_m^3).$$

$$(99)$$

We want to satisfy this equation with as many of the  $n_{uj}^*$  set to zero as possible. Note that on the left-hand side of Eq. (99), the  $u_p^2 u_m$  and  $u_p u_m^2$  no longer exists, and so on the right-hand side  $n_{u2}^*$  and  $n_{u3}^*$  cannot be zero. Conversely  $n_{u1}^*$  and  $n_{u4}^*$  can be set to zero, because the  $b_4$  and  $b_7$  coefficients can be chosen as required to satisfy Eq. (99). This gives relationships

$$8\omega_{n}^{2}b_{4} = \frac{2\beta^{2}}{3\omega_{n}^{2}} + \alpha$$

$$\frac{-10\beta^{2}}{3\omega_{n}^{2}} + 3\alpha - n_{u2}^{*} = 0$$

$$\frac{-10\beta^{2}}{3\omega_{n}^{2}} + 3\alpha - n_{u3}^{*} = 0$$

$$w$$

$$n_{u2}^{*} = n_{u3}^{*} = -\frac{10\beta^{2}}{3\omega_{n}^{2}} + 3\alpha$$

$$(100)$$

$$m_{u2}^{*} = n_{u3}^{*} = -\frac{10\beta^{2}}{3\omega_{n}^{2}} + 3\alpha$$

Substituting these coefficient values into Eq. (85) gives the coordinate transformation for linear, quadratic and cubic terms as

$$q = u_p + u_m + \frac{\beta}{3\omega_n^2}u_p^2 - 2\frac{\beta}{\omega_n^2}u_p u_m + \frac{\beta}{3\omega_n^2}u_m^2 + \left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right)u_p^3 + \left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right)u_m^3 + \mathcal{O}(4),$$
(101)

and substituting the  $n_{u2}^*$  and  $n_{u3}^*$  values into Eq. (83) gives

$$\ddot{u} + \omega_n^2 u + \left(3\alpha - \frac{10\beta^2}{3\omega_n^2}\right) (u_p^2 u_m + u_p u_m^2) = 0,$$
(102)

or alternatively, using the fact that  $u = u_p + u_m$ , we can write a pair of normal form equations as

$$\ddot{u}_{p} + \omega_{n}^{2}u_{p} + \left(3\alpha - \frac{10\beta^{2}}{3\omega_{n}^{2}}\right)u_{p}^{2}u_{m} = 0,$$
  
$$\ddot{u}_{m} + \omega_{n}^{2}u_{m} + \left(3\alpha - \frac{10\beta^{2}}{3\omega_{n}^{2}}\right)u_{p}u_{m}^{2} = 0.$$
 (103)

This example shows how a near-identity transformation can be used to eliminate the quadratic term in this type of nonlinear oscillator equation. However, for the cubic terms this was not possible and we shall develop a more complete formulation for dealing with this process in Section 4.2. We note also that:

1. We have only considered terms up to cubic order, and the expansions in Eq. (87) can be continued to higher orders. To deal with these higher order terms, order k = 4 terms must be added to the assumed solutions in Eq. (85), which in this case would mean

$$q = u + h^{(2)}(u, \dot{u}) + h^{(3)}(u, \dot{u}) + h^{(4)}(u, \dot{u}) + \dots \quad \rightsquigarrow \quad q = u_p + u_m + h^{(2)}(u_p, u_m) + h^{(3)}(u_p, u_m) + h^{(4)}(u_p, u_m) + \dots$$
(104)

which in turn lead to more higher order terms being generated in the expansion of the nonlinear term. This process can be undertaken as many times as required, and is one way to analyse the convergence of the series solution — see for an example [104]. However, specific proofs have been developed to show that the coefficients can be found at any order in the near identity transform providing certain conditions are met, and this will be discussed in Section 4.1.2.

- **2.** The fact that we could solve the order k = 2 homological equation with  $n^{(2)}(u) = 0$  means that there are no nonlinear resonances from the quadratic terms. However, the order k = 3 homological equation had some  $n_u^{(3)}$  terms that could not be eliminated, and these correspond to resonant terms. This will be discussed in more detail in Section 4.1.2 and methods for dealing with these terms are formalised in Section 4.2.
- **3.** The quadratic terms in this example generated terms at cubic order (i.e. the  $n_{(2)}^{(3)}$  terms) which then affected the result of the cubic order (k = 3 order) homological equation. This is actually a generic property of these types of oscillators, and as a result, the normal form computation will need to be carried out to at least the first *k* odd order that appears.

As a numerical example we show the solutions for *x* up to  $\mathcal{O}(3)$  that can be obtained directly from Eq. (94) because x = q in this case. Taking the assumed solutions for  $u_p$  and  $u_m$  to be the expressions given by Eq. (22) with  $\omega_r = \omega_n$  leads to

$$x = U\left(\frac{\mathrm{e}^{\mathrm{i}\omega_n t} + \mathrm{e}^{-\mathrm{i}\omega_n t}}{2}\right) + \frac{\beta}{3\omega_n^2}\frac{U^2}{2}\left(\frac{\mathrm{e}^{2\mathrm{i}\omega_n t} + \mathrm{e}^{-2\mathrm{i}\omega_n t}}{2}\right) - 2\frac{\beta}{\omega_n^2}\frac{U^2}{4} \quad \rightsquigarrow = U\cos(\omega_n t) + \frac{\beta U^2}{6\omega_n^2}\cos(2\omega_n t) - \frac{\beta U^2}{2\omega_n^2}.$$
(105)

Considering the case at the initial conditions when t = 0, then

$$x(0) = U_0 - \frac{\beta U_0^2}{3\omega_n^2} \quad \rightsquigarrow \quad U_0 = \frac{3\omega_n^2}{2\beta} \left( 1 - \sqrt{1 - \frac{4\beta x(0)}{3\omega_n^2}} \right).$$
(106)

Notice that the quadratic term, which arises from asymmetry, generates an amplitude-dependent offset from zero in the response in Eq. (105). It also always tends to create a softening nonlinear effect in the frequency-amplitude response — a phenomena that was introduced in Example 2, see Fig. 2, in the context of backbone curves, and will also be described in Example 18. Note also, that if the cubic coefficients in the normal form are zero, then additional higher-orders should be computed in order to determine the type of nonlinear behaviour in the system. A numerical example is shown in Fig. 4 (b) where the normal form solutions from Eq. (105) with Eq. (106) is compared to a direct numerical computation of the equations of motion.

## 4.1.2 Nonlinear resonance

Now let us consider in more detail the resonant terms that occurred in the order k = 3 homological equation from Example 10 (Section 4.1.1). First we assume that any terms of the type  $n_{(2)}^{(3)}$  (or in general  $n_{(k-1)}^{(k)}$ ) are combined with the  $n^{(3)}$  (or in general  $n^{(k)}$ ) on the right hand side of the order k = 3 homological equation in Eq. (88). As a result the *k*th order homological equation (for a single-degree-of-freedom oscillator) can be written as

Order 
$$k: -\frac{\mathrm{d}^2 h^{(k)}}{\mathrm{d}t^2} - \omega_n^2 h^{(k)} = n^{(k)} - n_u^{(k)},$$
 (107)

where  $n^{(k)}$  is now understood to include the *k* order terms from order *k* and order k-1. Then we will generalise the  $h^{(k)}$ ,  $n^{(k)}$  and  $n_{\mu}^{(k)}$  terms in the following way

$$h^{(k)} = \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}}, \quad n^{(k)} = \sum_{\mathbf{m}_{k}} n_{\mathbf{m}_{k}}^{*} u_{p}^{m_{p}} u_{m}^{m_{m}}, \quad \text{and} \quad n_{u}^{(k)} = \sum_{\mathbf{m}_{k}} n_{u,\mathbf{m}_{k}}^{*} u_{p}^{m_{p}} u_{m}^{m_{m}}, \tag{108}$$

161 | doi:10.25518/2684-6500.84



**Fig. 4:** (a) Eigenvalues for a damped (purple circles) and undamped (green circles) single degree-of-freedom oscillator. The straight line AA divides the damped eigenvalues (purple) from the origin and therefore these eigenvalues can be said to be in the Poincaré domain. No such line can be drawn for the undamped (green) eigenvalues, and these are therefore in the Siegel domain. (b) The solution, *x*, (where *x* = *q*) to the oscillator from Example 10 with  $\alpha = 0$ . The solution is computed using Eq. (105) with Eq. (106) and the result shown with red crosses. This normal form solution is compared with a reference solution generated from a fourth-order Runge-Kutta integration (the 'odeint' numerical integration routine from the Python numerical library — blue solid line). Parameter values are: x(0) = 0.01,  $\omega_n = \sqrt{10}$  rad/s and  $\beta = 1$  N/(kgm<sup>2</sup>),  $\alpha = 0$ .

where  $\mathbf{m}_k$  is based on a vector notation for multi-indices such that  $\mathbf{m}_k = (m_p, m_m)$ , where  $m_p$  and  $m_m$  are whole numbers in the range  $0 \le (m_p, m_m) \le k$  with the additional condition that  $m_p + m_m = k$  and the sum in (108) is over the set of all such indices [129]. Examples of  $\mathbf{m}_k$  and  $h^{(k)}$  for k = 2,3,4 are given in Table 2 along with the associated vector  $\mathbf{u}^*$  which we will use later in the generalisations of this process.

k	$\mathbf{m}_k$	$h^{(k)}$	<b>u</b> *	$L_k$
2	(2,0), (1,1), (0,2)	$b_{(2,0)}u_p^2 + b_{(1,1)}u_pu_m + b_{(0,2)}u_m^2$	$[u_p^2 \ u_p u_m \ u_m^2]^T$	3
3	(3,0), (2,1), (1,2) (0,3)	$b_{(3,0)}u_p^3 + b_{(2,1)}u_p^2u_m + b_{(1,2)}u_pu_m^2 + b_{(0,3)}u_m^3$	$[u_p^3 \ u_p^2 u_m \ u_p u_m^2 \ u_m^3]^T$	4
4	(4,0), (3,1), (2,2), (1,3)	$b_{(4,0)}u_p^4 + b_{(3,1)}u_p^3u_m^2 + b_{(2,2)}u_p^2u_m^2 + b_{(1,3)}u_p^2u_m^3 +$	$[u_p^4 \ u_p^3 u_m^2 \ u_p^2 u_m^2 \ u_p^2 u_m^2 \ u_p^2 u_m^3 \ u_m^4]^T$	5
	(0,4)	$b_{(0,4)}u_m^4$		

**Table 2:** Examples of  $\mathbf{m}_k$ ,  $h^{(k)}$ , and  $\mathbf{u}^*$  for k = 2, 3, 4. Note that we will only use this notation for the  $b_j$  coefficients in this subsection. The  $\mathbf{m}_k$  notation is useful to motivate the discussion, and show how the coefficients relate to the  $\mathbf{u}^*$  vector terms, but is too complex for use in general.  $L_k$  is the length of the  $\mathbf{u}^*$  vector.

If we substitute the relationships in Eq. (108) into Eq. (107), then the order k homological equation becomes

$$-\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \frac{\mathrm{d}^{2} u_{p}^{m_{p}} u_{m}^{m_{m}}}{\mathrm{d}t^{2}} - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = \sum_{\mathbf{m}_{k}} n_{\mathbf{m}_{k}}^{*} u_{p}^{m_{p}} u_{m}^{m_{m}} - \sum_{\mathbf{m}_{k}} n_{u,\mathbf{m}_{k}}^{*} u_{p}^{m_{p}} u_{m}^{m_{m}}.$$
(109)

Then, we need to evaluate the  $\frac{d^2 u_p^{m_p} u_m^{m_m}}{dt^2}$  derivative term using the chain rule and expressions in Eq. (84), as we did for Example 10, only this time we write these latter relationships as  $\dot{u}_p = \lambda_p u_p$  and  $\dot{u}_m = \lambda_m u_m$ , where  $\lambda_p$  and  $\lambda_m$  are the eigenvalues, which for this example are  $\lambda_p = \mathbf{i}\omega_n$  and  $\lambda_m = -\mathbf{i}\omega_n$ . Using these relationships, Eq. (109) becomes

$$-\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{\mathrm{d}u_{p}^{m_{p}} u_{m}^{m_{m}}}{\mathrm{d}t} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = -\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{\partial u_{p}^{m_{p}} u_{m}^{m_{m}}}{\partial u_{p}} \dot{u}_{p} + \frac{\partial u_{p}^{m_{p}} u_{m}^{m_{m}}}{\partial u_{m}} \dot{u}_{m} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = -\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \frac{\mathrm{d}}{\mathrm{d}t} \left( \frac{\partial u_{p}^{m_{p}} u_{m}^{m_{m}}}{\partial u_{p}} \lambda_{p} u_{p} + \frac{\partial u_{p}^{m_{p}} u_{m}^{m}}{\partial u_{m}} \lambda_{m} u_{m} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = -\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \frac{\mathrm{d}}{\mathrm{d}t} \left( [m_{p}\lambda_{p} + m_{m}\lambda_{m}] u_{p}^{m_{p}} u_{m}^{m} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = -\sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \left( [m_{p}\lambda_{p} + m_{m}\lambda_{m}]^{2} u_{p}^{m_{p}} u_{m}^{m} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \left( [m_{p}\lambda_{p} + m_{m}\lambda_{m}]^{2} u_{p}^{m_{p}} u_{m}^{m} \right) - \omega_{n}^{2} \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} = \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} \left( [m_{p}\lambda_{p} + m_{m}\lambda_{m}]^{2} - \omega_{n}^{2} \right) u_{p}^{m_{p}} u_{m}^{m_{m}} = \sum_{\mathbf{m}_{k}} b_{\mathbf{m}_{k}} u_{p}^{m_{p}} u_{m}^{m_{m}} - \sum_{\mathbf{m}_{k}} n_{k}^{m_{p}} u_{m}^{m_{m}}.$$

$$(110)$$

From this we can see that the expression  $(-[m_p\lambda_p + m_m\lambda_m]^2 - \omega_n^2)$  plays an important role in relation to the homological equation. Considering each of the  $\mathbf{m}_k$  terms we can set  $n_{u,\mathbf{m}_k}^* = 0$  and solve for

$$b_{\mathbf{m}_{k}} = \frac{n_{\mathbf{m}_{k}}^{*}}{(-[m_{p}\lambda_{p} + m_{m}\lambda_{m}]^{2} - \omega_{n}^{2})} \quad \text{for} \quad (-[m_{p}\lambda_{p} + m_{m}\lambda_{m}]^{2} - \omega_{n}^{2}) \neq 0,$$
(111)

and because in this example,  $\lambda_p^2 = \lambda_m^2 = -\omega_n^2$ , the relationship becomes

$$b_{\mathbf{m}_{k}} = \frac{n_{\mathbf{m}_{k}}^{*}}{([m_{p} - m_{m}]^{2} - 1)\omega_{n}^{2}} \quad \text{for} \quad ([m_{p} - m_{m}]^{2} - 1) \neq 0.$$
(112)

In the case when  $(-[m_p\lambda_p + m_m\lambda_m]^2 - \omega_n^2) = 0$  then the corresponding  $n_{u,\mathbf{m}_k}^*$  coefficient in Eq. (110) cannot be set to zero, and the term is said to be resonant. So for the resonant case we have

$$-[m_p\lambda_p + m_m\lambda_m]^2 - \omega_n^2 = 0 \quad \rightsquigarrow \quad m_p\lambda_p + m_m\lambda_m = +\mathbf{i}\omega_n \quad \text{and} \quad m_p\lambda_p + m_m\lambda_m = -\mathbf{i}\omega_n$$

$$\rightsquigarrow \quad m_p\mathbf{i}\omega_n - m_m\mathbf{i}\omega_n = \mathbf{i}\omega_n \quad \text{and} \quad m_p\mathbf{i}\omega_n - m_m\mathbf{i}\omega_n = -\mathbf{i}\omega_n,$$
(113)

which for the cubic terms in Example 10 is true when  $\mathbf{m}_k = (2,1)$  and  $\mathbf{m}_k = (1,2)$ . Notice from Table 2, Eq. (112) and Eq. (113) that  $m_p - m_m = \pm 1$  results in a resonant terms for the types of nonlinear examples considered here. As a result, by inspecting Table 2 it can be noticed that the *k* even nonlinearities will not have resonant terms of this type, whereas the *k* odd terms will have them.

Notice also that the key relationship in Eq. (113) depends only on the eigenvalues of the system. More formally, undamped oscillator equations that have pairs of complex conjugate eigenvalues that lie on the imaginary axis, and are said to be in the *Siegel domain* [4, 120, 85] (defined by the fact that no straight line can be drawn which divides the eigenvalues from the origin — see Fig. 4 (a)). On the contrary, eigenvalues for damped oscillators, that typically both have negative real parts, are said to be in the *Poincaré domain*, which is a case that will be considered in more detail in Section 4.5. An example showing both cases is shown schematically in Fig. 4 (a).

For eigenvalues in the Poincaré domain, there are only a finite number of resonant terms, and theories developed by Poincaré and Dulac give specific conditions for the existence of near-identity transformation in this case — see [4, 16, 85] for details. For eigenvalues in the Siegel domain, the situation is more complex, because there are an infinite number of resonant terms. In Example 10, for example the eigenvalues are  $\lambda_{p,m} = \pm i\omega_n$ , and then the following relationship holds [85]

$$\lambda_p = (n+1)\lambda_p + n\lambda_m \text{ for } n = 1, 2, 3, 4....$$
 (114)

which means there are an infinite number of potential resonant combinations, and therefore potential resonant terms, and these are defined as *trivial* resonance terms in [169, 172] and *unconditionally* resonant terms in [183]. Siegel and others (discussions can be found in [4, 16, 85]), have studied this case in detail and present

an number of ways of developing appropriate near-identity transformations<sup>43</sup>. In this work, we consider only systems relevant to structural dynamics which are essentially "well behaved" and so the behaviour of the asymptotic series expansions is assumed to be a reasonable approximation to the complete solution throughout the paper<sup>44</sup>. That said, the trivial resonances serve an important role in the normal form process for undamped oscillators, as they highlight which terms cannot be removed by the transformation (and therefore they are unconditional in the sense that they do not need a specific condition to be true in order to occur).

This is in contrast to the concept of a nonlinear resonant phenomena called *internal resonance* which occurs for systems with more than a single degree-of-freedom. For example consider a system with *N* degrees-of-freedom, and *N* pairs of complex conjugate eigenvalues  $\lambda_{pj}$ ,  $\lambda_{mj}$  for j = 1, 2, 3..., N. Then an internal resonance<sup>45</sup> exists for an eigenvalue,  $\lambda_{p\ell}$ , if

$$\lambda_{p\ell} = r_{p1}\lambda_{p1} + r_{p2}\lambda_{p2} + r_{p3}\lambda_{p3} + \dots \quad \rightsquigarrow \quad \lambda_{\ell} = \sum_{j=1}^{N} r_{pj}\lambda_{pj} \quad \text{for} \quad j = 1, 2, 3..., N,$$
(115)

where  $r_{pj}$  are integers. The  $r_{pj}$  values define the condition on the righthand side of Eq. (115) for this type of resonance to occur, and therefore this can also be called a *conditional* resonance.

Nonlinear resonance is a key topic for normal form transformations, and we will show multiple examples in the rest of this paper. Next however, we consider a subtle but important point regarding the notation in normal form transformations.

## 4.1.3 *k*-order and $\varepsilon$ -order notation in normal form transformations

So far in introducing the method of normal forms for nonlinear oscillator systems we have shown how the analysis can be separated into *k*-order homological equations, and then solved one order after another until the required level of accuracy is obtained for the specific problem being considered [104]. However, there is an alternative form of notation used in the normal form literature, which we will call the  $\varepsilon$ -order notation for normal form transformations. This notation is used in a variety of ways including:

- 1. In exactly the same way as the *k*-order notation, to keep track of the polynomial nonlinear terms. In this case  $\varepsilon$  is often called a *book-keeping* parameter, and will often be attributed the value  $\varepsilon = 1$ .
- **2.** For cases when the nonlinear terms are not separated into an ascending sum of polynomial nonlinear terms. Here  $\varepsilon$ -order notation can be used instead. As with point 1. in this case  $\varepsilon$  is often called a *book-keeping* parameter.
- **3.** As part of a series expansion used in the normal form method, as will be used in the method of Lie series in Section 4.3. This has been used in particular in connection with the Hamiltonian normal form, first introduced by Birkhoff [11]. The use of the  $\varepsilon$ -order notation in this context goes back at least as far as the work of Hori [74]. In this case  $\varepsilon$  is a Lie series expansion parameter and is generally considered to be small  $\varepsilon \ll 1$ .
- 4. Within a Taylor series expansion when dealing with nonlinear functions that are not already in the form of polynomial nonlinearities [183]. In this case  $\varepsilon$  is a Taylor series expansion parameter and is generally considered to be small  $\varepsilon \ll 1$ .
- 5. Related to points 3. and 4. normal form methods can also be formulated in terms of perturbation theory (see for example Nayfeh [124]), where  $\varepsilon$  is the perturbation parameter and is generally considered to be small  $\varepsilon \ll 1$ .
- 6. Related to point 5., in some normal form analysis,  $\varepsilon$  has been used to link the size of the nonlinear term to the size the viscous damping and external forcing terms, which are all assumed to be small [183]. This is a case of practical importance in structural dynamics, where large resonant responses can occur, for small nonlinearity, damping and forcing [183].
- 7. To include parameter variation close to the equilibrium point around which the normal form expansion is being carried out. This will be discussed further in Section 4.6.

<sup>&</sup>lt;sup>43</sup>These and other authors also discuss the use of Jordan normal form for the different eigenvalue cases including those with degenerate behaviour [4, 16, 120]. This is beyond the scope of this current paper.

<sup>&</sup>lt;sup>44</sup> Formally we consider only systems that are diagonalisable with full rank diagonal matrices of distinct eigenvalues. Often referred to as the semisimple case [4, 120]. For cases where this assumption regarding asymptotic series expansions does not necessarily hold, see Murdock [120]. For an example of the non-semisimple case see [43].

<sup>&</sup>lt;sup>45</sup>There are also other names for this phenomena, such as "modal coupling", and "autoparametric resonance", see for example [164]. In terms of recent applications the area of micro- and nano-mechanical structures offers some very nice examples [157, 31].

Murdock [120] gives a clear explanation of the difference between the *k*-order and  $\varepsilon$ -order notation in terms of five specific formats. The *k*-order notation arises primarily in the so-called *direct* formats (e.g. Example 10, Section 4.1.1), and the  $\varepsilon$ -order notation primarily in the *generated* formats (as will be described in Section 4.3). As noted by Murdock, in some methods, such as those by Hori [74] and Deprit [34],  $\varepsilon$  is treated as an independent variable, and so cannot be eliminated. In order to try and retain 'the best of both worlds' we will develop a generalised theory of near-identity transformations based on  $\varepsilon$ -order notation, that can in certain circumstances use  $\varepsilon = 1$  and revert to the *k*-order approach. Next we consider the near-identity transformation process in more detail.

## 4.2 Near-identity transformations of differential equations

In this Section we consider near-identity transformations for both first-order and second-order sets of differential equation. It is helpful to maintain the distinction in variables between these two classes of differential equation. So we introduce the following notation: *FO* will denote the case for first-order sets of differential equations and *SO* will denote the case for second-order sets of differential equations. In both cases, *t* is the scalar independent variable and remains untransformed (or its transform is the identity). In addition for the *SO* case  $\mathbf{q} = \mathbf{q}(t)$  is an  $N \times 1$  vector containing the dependent (displacement) variables, while  $\dot{\mathbf{q}}$  and  $\ddot{\mathbf{q}}$  are  $N \times 1$  vectors containing the velocities and accelerations, respectively. For the *FO* case,  $\mathbf{q} = [\mathbf{q}, \dot{\mathbf{q}}]^T$  is the  $2N \times 1$  state vector, and  $\dot{\mathbf{q}}$  is the time derivative of the state vector.

The functional form of the near-identity transformation will be defined as the following two mappings

$$T_{FO}: (t, \boldsymbol{q}) \to (t, \boldsymbol{u}): \qquad t = t \qquad \text{and} \qquad \boldsymbol{q} = \boldsymbol{T}(\boldsymbol{u}, \boldsymbol{\varepsilon}) = \boldsymbol{u} + \boldsymbol{\varepsilon} \boldsymbol{h}_1(\boldsymbol{u}) + \boldsymbol{\varepsilon}^2 \boldsymbol{h}_2(\boldsymbol{u}) + \dots$$

$$T_{SO}: (t, \boldsymbol{q}) \to (t, \boldsymbol{u}): \qquad t = t \qquad \text{and} \qquad \boldsymbol{q} = \boldsymbol{T}(\boldsymbol{u}, \dot{\boldsymbol{u}}, \boldsymbol{\varepsilon}) = \boldsymbol{u} + \boldsymbol{\varepsilon} \boldsymbol{h}_1(\boldsymbol{u}, \dot{\boldsymbol{u}}) + \boldsymbol{\varepsilon}^2 \boldsymbol{h}_2(\boldsymbol{u}, \dot{\boldsymbol{u}}) + \dots$$
(116)

Notice also that  $\mathbf{h}_j$  is assumed in general to be a function of both  $\mathbf{u}$  and  $\dot{\mathbf{u}}^{46}$ . The nonlinear functions  $h_j(\mathbf{u})$  and  $\mathbf{h}_j(\mathbf{u}, \dot{\mathbf{u}})$  are defined as

$$h_1 = b_1 u^*, \quad h_2 = b_2 u^+$$
 and  $h_1 = b_1 u^*, \quad h_2 = b_2 u^+$  (117)

where  $\mathbf{b}_1$  is an  $2N \times L_k$  coefficient matrix,  $\mathbf{b}_1$  is an  $N \times L_k$  coefficient matrix, and  $\mathbf{u}^*$  is an  $L_k \times 1$  vector of polynomial functions of the elements of  $\mathbf{u}$  or  $\mathbf{u}$ , which for a single-degree-of-freedom oscillator are  $u_p$  and  $u_m$  as can be seen in Table 2. Similarly,  $\mathbf{b}_2$  is an  $2N \times L_k$  coefficient matrix,  $\mathbf{b}_2$  is an  $N \times L_k$  coefficient matrix, and  $\mathbf{u}^+$  is an  $L_k \times 1$  vector of polynomial functions of the elements of  $\mathbf{u}$  or  $\mathbf{u}$ . For example, if we consider the oscillator system from Example 10 and rewrite the near-identity transformation in the form of Eq. (116), then Eq. (101) becomes

$$q = u_p + u_m + \left[\frac{\beta}{3\omega_n^2} - 2\frac{\beta}{\omega_n^2} \frac{\beta}{3\omega_n^2}\right] \begin{bmatrix} u_p^2 \\ u_p u_m \\ u_n^2 \end{bmatrix} + \left[\left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right) \ 0 \ 0 \ \left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right)\right] \begin{bmatrix} u_p^3 \\ u_p^2 u_m \\ u_p u_m^2 \\ u_n^3 \end{bmatrix} + \dots,$$
so taking
$$\mathbf{b}_1 = \left[\frac{\beta}{3\omega_n^2} - 2\frac{\beta}{\omega_n^2} \frac{\beta}{3\omega_n^2}\right], \quad \text{and} \quad \mathbf{b}_2 = \left[\left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right) \ 0 \ 0 \ \left(\frac{\beta^2}{12\omega_n^4} + \frac{\alpha}{8\omega_n^2}\right)\right] \right]$$

$$\Rightarrow q = u + \mathbf{b}_1 \mathbf{u}^* + \mathbf{b}_2 \mathbf{u}^+ + \dots \text{ and with } \varepsilon \text{ notation } q = u + \varepsilon \mathbf{b}_1 \mathbf{u}^* + \varepsilon^2 \mathbf{b}_2 \mathbf{u}^+ + \dots$$
(118)

where  $u = u_p + u_m$ , N = 1,  $L_{k=2} = 3$ ,  $L_{k=3} = 4$  and  $\varepsilon = 1$ .

## 4.2.1 First-order equations

First let us consider first-order differential equations of the form

$$\dot{\boldsymbol{q}} = \mathscr{F}(\boldsymbol{q}, \boldsymbol{\varepsilon}) = \Lambda \boldsymbol{q} + \boldsymbol{N}(\boldsymbol{q}) = \Lambda \boldsymbol{q} + \boldsymbol{\varepsilon} \boldsymbol{n}_1(\boldsymbol{q}) + \boldsymbol{\varepsilon}^2 \boldsymbol{n}_2(\boldsymbol{q}) + \dots$$
(119)

<sup>&</sup>lt;sup>46</sup>Of course the state vector already includes the velocities because  $q = [\mathbf{q}, \dot{\mathbf{q}}]^T$ , so this comment applies only to second-order sets of differential equations.

where  $\mathscr{F}(\cdot)$  represents the general expression for the nonlinear function of q,  $\Lambda$  is a diagonal matrix containing the system eigenvalues, N(q) contains the nonlinear terms (i.e. as defined in Eq. (81)), which are approximated by the  $n_i(q)$  terms. Define the structure of the transformed differential equation to be

$$\dot{\boldsymbol{u}} = \mathscr{G}(\boldsymbol{u}, \boldsymbol{\varepsilon}) = \Lambda \boldsymbol{u} + \boldsymbol{\varepsilon} \boldsymbol{n}_{u1}(\boldsymbol{u}) + \boldsymbol{\varepsilon}^2 \boldsymbol{n}_{u2}(\boldsymbol{u}) + \dots,$$
(120)

where  $\mathscr{G}(\cdot)$  is the transformed nonlinear expression, and  $n_{uj}$  are the transformed vectors of nonlinear terms. Here the objective of the transformation process is to simplify, as much as possible, the transformed vectors of nonlinear terms  $n_{uj}$ . The standard approach to this problem is to substitute (the upper equation in) Eq. (116) into Eq. (119) to give

$$\dot{\boldsymbol{u}} + \frac{\mathrm{d}}{\mathrm{d}t} \left( \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots \right) = \Lambda (\boldsymbol{u} + \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots) + \varepsilon \boldsymbol{n}_1 (\boldsymbol{u} + \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots) + \varepsilon^2 \boldsymbol{n}_2 (\boldsymbol{u} + \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots) \ldots$$
(121)

and by substituting for  $\dot{u}$  using Eq. (120) gives

$$\Lambda \boldsymbol{u} + \varepsilon \boldsymbol{n}_{u1}(\boldsymbol{u}) + \varepsilon^2 \boldsymbol{n}_{u2}(\boldsymbol{u}) + \frac{\mathrm{d}}{\mathrm{d}t} \left( \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots \right) =$$

$$\Lambda \boldsymbol{u} + \varepsilon \Lambda \boldsymbol{h}_1 + \varepsilon^2 \Lambda \boldsymbol{h}_2 + \varepsilon \boldsymbol{n}_1 (\boldsymbol{u} + \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots) + \varepsilon^2 \boldsymbol{n}_2 (\boldsymbol{u} + \varepsilon \boldsymbol{h}_1 + \varepsilon^2 \boldsymbol{h}_2 + \ldots) \dots$$
(122)

For cases when  $n_1$  and  $n_2$  are relatively simple, the substituted terms in Eq. (122) can be expanded directly. More generally the nonlinear terms  $n_1$  and  $n_2$  can be expanded as Taylor series using

$$\boldsymbol{n}_i(\boldsymbol{u}+\boldsymbol{\varepsilon}\boldsymbol{h}_1(\boldsymbol{u})+\ldots)=\boldsymbol{n}_i(\boldsymbol{u})+\boldsymbol{\varepsilon}\boldsymbol{D}\{\boldsymbol{n}_i(\boldsymbol{u})\}\boldsymbol{h}_1(\boldsymbol{u})+\ldots,$$
(123)

where  $D\{n_i\}$  is the Jacobian of  $n_i$ . Using this Taylor series expansion and then equating  $\varepsilon$  terms in Eq. (122) gives

$$\boldsymbol{\varepsilon}^0: \quad \boldsymbol{\Lambda} \boldsymbol{\boldsymbol{u}} = \boldsymbol{\Lambda} \boldsymbol{\boldsymbol{u}}, \tag{124}$$

$$\boldsymbol{\varepsilon}^{1}: \quad \frac{\mathrm{d}\boldsymbol{h}_{1}}{\mathrm{d}t} - \boldsymbol{\Lambda}\boldsymbol{h}_{1} = \boldsymbol{n}_{1} - \boldsymbol{n}_{u1}, \tag{125}$$

$$\varepsilon^{2}: \quad \frac{\mathrm{d}\boldsymbol{h}_{2}}{\mathrm{d}t} - \Lambda \boldsymbol{h}_{2} = D\{\boldsymbol{n}_{1}\}\boldsymbol{h}_{1} + \boldsymbol{n}_{2} - \boldsymbol{n}_{u2}, \tag{126}$$

To deal with the time derivatives in Eq. (125) and Eq. (126) we use the Lie derivative. For example, consider computing the Lie derivate of the term  $h_1 = b_1 u^*$  in Eq. (125) to order  $\varepsilon^0$ . First  $b_1$  is a matrix of constants, so we are interested only in the Lie derivative of  $u^*$ , which is given by

$$\frac{\mathrm{d}\mathbf{u}^*}{\mathrm{d}t}\Big|_{\varepsilon=0} = \frac{\partial \mathbf{u}^*}{\partial \boldsymbol{u}} \dot{\boldsymbol{u}}\Big|_{\varepsilon=0} = \frac{\partial \mathbf{u}^*}{\partial \boldsymbol{u}} \Lambda \boldsymbol{u} \equiv \Lambda^* \mathbf{u}^*, \tag{127}$$

where  $\Lambda^*$  is a constant  $L_k \times L_k$  diagonal matrix and  $L_k$  is the length of the vector  $\mathbf{u}^*$ . Substituting this, and relations for  $\mathbf{h}_1$  from Eq. (117) into the homological equations Eq. (125) and Eq. (126) gives

$$\varepsilon^{1}: \quad \frac{\partial h_{1}}{\partial u} \Delta u - \Delta h_{1} = n_{1} - n_{u1}, \qquad \qquad \qquad \qquad b_{1} \Delta^{*} \mathbf{u}^{*} - \Delta b_{1} \mathbf{u}^{*} = n_{1} - n_{u1}, \qquad (128)$$
  

$$\varepsilon^{2}: \quad \frac{\partial h_{2}}{\partial u} \Delta u - \Delta h_{2} = D\{n_{1}\}h_{1} + n_{2} - n_{u2}, \qquad \qquad \qquad \qquad \qquad \qquad b_{2} \Delta^{+} \mathbf{u}^{+} - \Delta b_{2} \mathbf{u}^{+} = D\{n_{1}\}b_{1} \mathbf{u}^{*} + n_{2} - n_{u2},$$

$$\varepsilon^{2}: \quad \frac{\partial u}{\partial u}\Lambda u - \Lambda h_{2} = D\{n_{1}\}h_{1} + n_{2} - n_{u2}, \qquad \qquad \Rightarrow \qquad b_{2}\Lambda^{+}\mathbf{u}^{+} - \Lambda b_{2}\mathbf{u}^{+} = D\{n_{1}\}b_{1}\mathbf{u}^{*} + n_{2} - n_{u2},$$
(129)

where  $\Lambda^+$  is a diagonal matrix obtained from taking the Lie derivative of  $\mathbf{u}^+$  using Eq. (127), and we have truncated the expressions at order- $\varepsilon^2$ .

These relationships are called the order- $\varepsilon^1$  and order- $\varepsilon^2$  homological equations (without frequency detuning which will be added in Section 4.6). Notice that the left-hand side of the homological equations in Eq. (128) and Eq. (129) can be reformulated as a Lie bracket for the two vector fields involved in the transformation process such that

$$\frac{\partial \boldsymbol{h}_{\ell}}{\partial \boldsymbol{u}} \mathscr{G}_{0} - \frac{\partial \mathscr{G}_{0}}{\partial \boldsymbol{u}} \boldsymbol{h}_{\ell} = [\boldsymbol{h}_{\ell}, \mathscr{G}_{0}], \quad \text{where} \quad \mathscr{G}_{0} = \Lambda \boldsymbol{u} = \dot{\boldsymbol{u}}|_{\varepsilon = 0} \quad \text{for} \quad \ell = 1, 2, 3...$$
(130)

166 | doi:10.25518/2684-6500.84

where  $[\cdot, \cdot]$  indicates the bracketing operation, and it should be understood that as all functions in Eq. (130) are functions of  $\boldsymbol{u}$ , the bracket is an operator  $[\cdot, \cdot](\boldsymbol{u})$ . As a result for the order- $\varepsilon^1$  homological operator we can write<sup>47</sup>

$$[\boldsymbol{h}_1, \mathscr{G}_0] = \boldsymbol{n} - \boldsymbol{n}_u = \mathscr{H}_1. \tag{131}$$

Notice also the similarity of the structure of these homological equations and those derived in Example 10, particularly in Eq. (88), although is should be noted that in this Section we are dealing with first-order equations. A direct comparison can be made with the results in Section 4.3.2. Next we consider an example of how to compute the order- $\varepsilon^1$  homological equation in practice.

It is important to note, that in the next example we use only the order- $\varepsilon^1$  homological equation because the example contains only a single odd polynomial nonlinear term. As we saw in Example 10, if there are both even and odd polynomial terms, the even terms will lead to terms at the next order, and so in that case the order- $\varepsilon^2$  homological equation also needs to be included in the analysis. An example of this type will be considered in Section 4.8.1.

#### 4.2.2 Example 11: the undamped, unforced Duffing oscillator

The undamped, unforced Duffing oscillator can be written as

$$\ddot{x} + \omega_n^2 x + \varepsilon \alpha x^3 = 0, \tag{132}$$

where  $\omega_n$  is the linear natural frequency,  $\varepsilon$  is the book-keeping parameter and  $\alpha$  is a nonlinear coefficient.

For the classical method of treating the Duffing oscillator, a first-order representation of Eq. (132) is defined in terms of the state vector  $\mathbf{x} = \{x_1 \ x_2\}^T = \{x \ \dot{x}\}^T$ . Then the matrix and vector form of Eq. (132) is given by

$$\dot{\boldsymbol{x}} = A\boldsymbol{x} + \tilde{\mathcal{N}}(\boldsymbol{x}): \quad A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & 0 \end{bmatrix}, \quad \tilde{\mathcal{N}}(\boldsymbol{x}) = \begin{cases} 0 \\ -\varepsilon\alpha x_1^3 \end{cases}$$

Transforming using the diagonalising modal transform from Eq. (75), where  $x = \Phi q$  can be used to give an equation of the form of Eq. (119) as

$$\dot{\boldsymbol{q}} = \Lambda \boldsymbol{q} + \varepsilon \boldsymbol{n}_1(\boldsymbol{q}): \quad \Lambda = \begin{bmatrix} \mathbf{i}\omega_n & 0\\ 0 & -\mathbf{i}\omega_n \end{bmatrix}, \quad \Phi = \begin{bmatrix} 1 & 1\\ \mathbf{i}\omega_n & -\mathbf{i}\omega_n \end{bmatrix}, \quad \Phi^{-1} = \frac{1}{-2\mathbf{i}\omega_n} \begin{bmatrix} -\mathbf{i}\omega_n & -1\\ -\mathbf{i}\omega_n & 1 \end{bmatrix}, \quad (133)$$

where the nonlinear function  $\mathbf{n}_1(\mathbf{q}) = \Phi^{-1} \tilde{\mathcal{N}}(\Phi \mathbf{q})$ , and  $\mathbf{q} = [q_p \ q_m]^T$ . Note there is just one nonlinear term, and so we set  $\mathbf{n}_2 = 0, \dots$  etc. and then as there is just a single term we set  $\mathbf{n}_1(\mathbf{q}) = \mathbf{n}(\mathbf{q})$  which can be expressed as

$$\boldsymbol{n}(\boldsymbol{q}) = \Phi^{-1} \tilde{\mathcal{N}}(\Phi \boldsymbol{q}) = \frac{\varepsilon \alpha}{2\mathbf{i}\omega_n} \begin{bmatrix} -(q_p + q_m)^3 \\ (q_p + q_m)^3 \end{bmatrix} = \frac{\alpha}{2\mathbf{i}\omega_n} \begin{bmatrix} -1 & -3 & -3 & -1 \\ 1 & 3 & 3 & 1 \end{bmatrix} \begin{bmatrix} q_p^3 \\ q_p^2 q_m \\ q_p q_m^2 \\ q_m^3 \end{bmatrix} = \boldsymbol{n}^* \mathbf{q}^*, \quad (134)$$

where

$$\boldsymbol{n}^{*} = \frac{\alpha}{2\mathbf{i}\omega_{n}} \begin{bmatrix} -1 & -3 & -3 & -1 \\ 1 & 3 & 3 & 1 \end{bmatrix}, \quad \text{and} \quad \boldsymbol{q}^{*} = \begin{bmatrix} q_{p}^{3} \\ q_{p}^{2}q_{m} \\ q_{p}q_{m}^{2} \\ q_{m}^{3} \end{bmatrix}.$$
(135)

<sup>&</sup>lt;sup>47</sup>It can be shown that because of the structure of the basis functions in  $\mathbf{u}^*$ , the Lie derivative of  $\mathbf{u}^*$  is a constant matrix times  $\mathbf{u}^*$ . In other words, the derivative of  $\mathbf{u}^*$  with respect to time has the same characteristics as the exponential function, which is of course the underlying solution to all these differential equations. It is also important to realise that this is the classical Lie derivative of a scalar function along the vector field  $\mathscr{G}_0$ . In fact, we use the format defined by [151], where there are a set of scalar functions,  $u_1^*, u_2^*, u_3^*, ..., u_L^*$ , grouped together in a single vector  $\mathbf{u}^*$ .

Substituting the near-identity transformation, Eq. (116), into this expression is required as part of the process in deriving the homological equations, as we described above. However, if the analysis is carried out retaining only terms up to order  $\varepsilon^1$  this will result in

$$\varepsilon n(q) \rightarrow \varepsilon n(u + \varepsilon h_1 + ...) = \varepsilon n(u) + \mathcal{O}(\varepsilon^2),$$

and therefore to order  $\varepsilon^1$ 

$$\boldsymbol{n}(\boldsymbol{q}) = \boldsymbol{n}^* \boldsymbol{q}^* \to \boldsymbol{n}^* \boldsymbol{u}^* \quad \text{where for this example} \quad \boldsymbol{u}^* = \begin{bmatrix} u_p^3 \\ u_p^2 u_m \\ u_p u_m^2 \\ u_m^3 \end{bmatrix},$$
(136)

and  $n^*$  represents the coefficient matrix and  $u^*$  is the vector of nonlinear basis terms.

Now the order- $\varepsilon^1$  homological equation, (Eq. (128)) can most easily be evaluated if the structure of  $n_u$  (from Eq. (120) with  $n_u = n_{u1}$ ) and h (from Eq. (117) with  $h = h_1$ ) are set to be the same as n, which in turn is defined by the problem at hand. Therefore we set  $h(u) = bu^*$  and  $n_u(u) = n_u^*u^*$ , where b and  $n_u^*$  are coefficient matrices of the same dimensions as  $n^*$ . In fact, as  $u^*$  follows the structure of the nonlinear terms in n(q), the coefficient matrices b and  $n_u^*$  are the only remaining unknowns that are to be determined from the homological equation. Having defined h(u) we see that  $\frac{\partial h}{\partial u} = b \frac{\partial u^*}{\partial u}$  and in this example  $u = [u_p \ u_m]^T$ , then<sup>48</sup>

$$\frac{\partial \mathbf{u}^{*}}{\partial \boldsymbol{u}} = \begin{bmatrix} 3u_{p}^{2} & 0\\ 2u_{p}u_{m} & u_{p}^{2}\\ u_{m}^{2} & 2u_{p}u_{m}\\ 0 & 3u_{m}^{2} \end{bmatrix}, \quad \text{and} \quad \frac{\partial \mathbf{u}^{*}}{\partial \mathbf{u}} \Lambda \mathbf{u} \equiv \begin{bmatrix} 3\mathbf{i}\omega_{n} & 0 & 0 & 0\\ 0 & \mathbf{i}\omega_{n} & 0 & 0\\ 0 & 0 & -\mathbf{i}\omega_{n} & 0\\ 0 & 0 & 0 & -3\mathbf{i}\omega_{n} \end{bmatrix} \begin{bmatrix} u_{p}^{3}\\ u_{p}^{2}u_{m}\\ u_{p}u_{m}^{2}\\ u_{m}^{3} \end{bmatrix} = \Lambda^{*}\mathbf{u}^{*}. \quad (138)$$

Now using the expressions developed above, we obtain the Lie bracket expression (Eq. (130)) for the Duffing example as

$$\begin{bmatrix} \mathbf{h}, \mathscr{G}_{0} \end{bmatrix} = \begin{bmatrix} b_{1} & b_{2} & b_{3} & b_{4} \\ b_{5} & b_{6} & b_{7} & b_{8} \end{bmatrix} \begin{bmatrix} \mathbf{3i}\omega_{n} & 0 & 0 & 0 \\ 0 & \mathbf{i}\omega_{n} & 0 & 0 \\ 0 & 0 & -\mathbf{i}\omega_{n} & 0 \\ 0 & 0 & 0 & -\mathbf{3i}\omega_{n} \end{bmatrix} \begin{bmatrix} u_{p}^{3} \\ u_{p}^{2} \\ u_{m}^{3} \end{bmatrix}$$
(139)
$$-\begin{bmatrix} \mathbf{i}\omega_{n} & 0 \\ 0 & -\mathbf{i}\omega_{n} \end{bmatrix} \begin{bmatrix} b_{1} & b_{2} & b_{3} & b_{4} \\ b_{5} & b_{6} & b_{7} & b_{8} \end{bmatrix} \begin{bmatrix} u_{p}^{3} \\ u_{p}^{2} \\ u_{m}^{2} \\ u_{p}^{2} \\ u_{m}^{3} \end{bmatrix}.$$

We know from the order- $\varepsilon^1$  homological equation (Eq. (131)) that  $\mathbf{n} - \mathbf{n}_u = [\mathbf{h}, \mathscr{G}_0]$  and so using Eq. (134), Eq. (136) and Eq. (139) gives,

$$\begin{pmatrix} \alpha \\ 2\mathbf{i}\omega_{n} \begin{bmatrix} -1 & -3 & -3 & -1 \\ 1 & 3 & 3 & 1 \end{bmatrix} - \begin{bmatrix} n_{u1}^{*} & n_{u2}^{*} & n_{u3}^{*} & n_{u4}^{*} \\ n_{u5}^{*} & n_{u6}^{*} & n_{u7}^{*} & n_{u8}^{*} \end{bmatrix} \mathbf{u}^{*} = \begin{bmatrix} b_{1}(2\mathbf{i}\omega_{n}) & b_{2}(0) & b_{3}(-2\mathbf{i}\omega_{n}) & b_{4}(-4\mathbf{i}\omega_{n}) \\ b_{5}(4\mathbf{i}\omega_{n}) & b_{6}(2\mathbf{i}\omega_{n}) & b_{7}(0) & b_{8}(-2\mathbf{i}\omega_{n}) \end{bmatrix} \mathbf{u}^{*}.$$

$$(140)$$

<sup>48</sup>Note that the second equation in Eq. (138) gives an  $L \times 1$  vector that has been re-expressed as an  $L \times L$  matrix,  $\Lambda^*$  multiplied by  $\mathbf{u}^*$  (which is an  $L \times 1$  vector). We note that  $\Lambda^*$  can be obtained directly from

$$\Lambda^* = I_L \odot \left[ \mathbf{1}_L^T \otimes \left( \left[ \frac{\partial \mathbf{u}^*}{\partial \mathbf{u}} \Lambda \mathbf{u} \right] \oslash \mathbf{u}^* \right) \right]$$
(137)

where  $I_L$  is an  $L \times L$  identity matrix,  $\mathbf{1}_L$  is a  $L \times 1$  vector of ones,  $\odot$  denotes Hadamard product,  $\otimes$  denotes outer product, and  $\oslash$  denotes Hadamard division. Note that Neild and Wagg [128] derive an alternative method for computing  $\Lambda^*$  which based on the indices of the terms in  $\mathbf{u}^*$ . This method may be preferred in practice, particularly for multi-degree-of-freedom nonlinear oscillators.

To solve for the unknown coefficients,  $b_i$  and  $n_{ui}^*$  for i = 1, 2..8 we use the following logic. If there is a zero on the right hand-side of Eq. (140), then  $n_{ui}^* = n_i^*$  from solving the lefthand side of Eq. (140). If there is not a zero on the right hand-side of Eq. (140), then  $n_{ui}^* = 0$  is assumed, and the coefficient  $b_i$  can be solved for. Note that in this example this choice means that  $b_2$  and  $b_7$  can have any value, and following [85] we refer to these coefficients as the *free functions*, which will be discussed in Section 4.7. Using this approach the coefficient matrices can be found as

$$\boldsymbol{b} = \begin{bmatrix} \frac{\alpha}{4\omega_n^2} & 0 & \frac{-3\alpha}{4\omega_n^2} & \frac{-\alpha}{8\omega_n^2} \\ \frac{-\alpha}{8\omega_n^2} & \frac{-3\alpha}{4\omega_n^2} & 0 & \frac{\alpha}{4\omega_n^2} \end{bmatrix} \text{ and } \boldsymbol{n}_u^* = \begin{bmatrix} 0 & \frac{-3\alpha}{2\mathbf{i}\omega_n} & 0 & 0 \\ 0 & 0 & \frac{3\alpha}{2\mathbf{i}\omega_n} & 0 \end{bmatrix}.$$

Now the normal form (Eq. (120)), up to order  $\varepsilon^1$ , can be written as

$$\dot{\boldsymbol{u}}_{p} = \mathbf{i}\omega_{n}\boldsymbol{u}_{p} + \varepsilon \frac{3\mathbf{i}\alpha}{2\omega_{n}}\boldsymbol{u}_{p}^{2}\boldsymbol{u}_{m},$$

$$\dot{\boldsymbol{u}} = \Lambda \boldsymbol{u} + \varepsilon \boldsymbol{n}_{u}^{*}\mathbf{u}^{*} \quad \rightsquigarrow$$

$$\dot{\boldsymbol{u}}_{m} = -\mathbf{i}\omega_{n}\boldsymbol{u}_{m} - \varepsilon \frac{3\mathbf{i}\alpha}{2\omega_{n}}\boldsymbol{u}_{p}\boldsymbol{u}_{m}^{2},$$
(141)

and because of the occurrence of imaginary parts in these expressions, this is called the *complex* normal form format. The near-identity transform (Eq. (117)) becomes

$$\boldsymbol{q} = \boldsymbol{u} + \varepsilon \boldsymbol{b} \mathbf{u}^* \quad \rightsquigarrow \quad \begin{bmatrix} q_p \\ q_m \end{bmatrix} = \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \frac{\varepsilon \alpha}{\omega_n^2} \begin{bmatrix} \frac{u_p^3}{4} - \frac{3u_p u_m^2}{4} - \frac{u_m^3}{8} \\ -\frac{u_p^3}{8} - \frac{3u_p^2 u_m}{4} + \frac{u_m^3}{4} \end{bmatrix}$$
(142)

and because  $x = q_p + q_m$  from Eq. (142) we can express x as

$$x = u_p + u_m + \frac{\varepsilon \alpha}{\omega_n^2} \left[ \frac{u_p^3 + u_m^3}{8} - \frac{3}{4} (u_p^2 u_m + u_p u_m^2) \right] = (U - \varepsilon \frac{3\alpha U^3}{16\omega_n^2}) \cos(\omega_r t) + \varepsilon \frac{\alpha U^3}{32\omega_n^2} \cos(3\omega_r t) + \mathcal{O}(\varepsilon^2),$$
(143)

once the base solutions from Eq. (22) are substituted<sup>49</sup>.

Note that Eq. (141) can be compared with Eq. (20) from Example 2 (Section 1.1.3) in the case where  $\varepsilon = 1$  (note also the sign difference because of the  $-\alpha$  term in Eq. (19)). Likewise, Eq. (143) can be compared with Eq. (21) noting again the  $-\alpha$  sign difference from Eq. (19). Note also the subtle differences between these expressions in terms of when  $\omega_r$  appears compared to  $\omega_n$ . In order to formalise this we will introduce a detuning in Section 4.6.

The zeros in the homological equation, Eq. (140) relate to the resonant terms, that occur in this example. As a result, the resulting set of normal form equations in Eq. (141) is not linear, the terms relating to the resonances must be retained.

In order to actually compute the *x* approximation from Eq. (143), we first need to find the initial condition value of *U*, which we denote  $U_0$  at time t = 0. Substituting t = 0 into Eq. (143), and using Cardano's formula leads to

$$x(0) = U_0 - \varepsilon \frac{5\alpha U^3}{32\omega_n^2}, \quad \forall \quad U_0 = \sqrt[3]{-\frac{\Gamma}{2}} + \sqrt{\frac{\Gamma^2}{4} + \frac{\Pi^3}{27}} + \sqrt[3]{-\frac{\Gamma}{2}} - \sqrt{\frac{\Gamma^2}{4} + \frac{\Pi^3}{27}}, \quad \text{where } \Pi = -\frac{32\omega_n^2}{5\varepsilon\alpha}, \quad (144)$$

and  $\Gamma = -\Pi x(0)$ . Note that Cardano's formula applies when the discriminant is positive, meaning  $4\Pi^3 + 27\Gamma^2 > 0$ . A numerical example is shown in Fig. 5.

Next we consider the approach to near-identity transformations for second-order equations.

<sup>&</sup>lt;sup>49</sup>Note that the base solutions in Eq. (22) are defined in terms of a response frequency  $\omega_r$  which may be taken as equal to or not equal to the natural frequencies  $\omega_n$  and  $\omega_d$  depending on the specific context of the problem at hand.

#### 4.2.3 Second-order equations

Now let us consider second-order differential equations of the form

$$\ddot{\mathbf{q}} = \hat{\mathscr{F}}(\mathbf{q}, \varepsilon) = -\hat{\Lambda}\mathbf{q} - \varepsilon\hat{\mathbf{n}}_1(\mathbf{q}) - \varepsilon^2\hat{\mathbf{n}}_2(\mathbf{q}) + \dots$$
(145)

where  $\hat{\Lambda}$  is defined in Eq. (70) and the  $\hat{\mathbf{n}}_j(\mathbf{q})$  contain the nonlinear terms. In this case, we define the structure of the desired transformed differential equation to be

$$\ddot{\mathbf{u}} = \hat{\mathscr{G}}(\mathbf{u}, \varepsilon) = -\hat{\Lambda}\mathbf{u} - \varepsilon\hat{\mathbf{n}}_{\mathbf{u}1}(\mathbf{u}) - \varepsilon^2\hat{\mathbf{n}}_{\mathbf{u}2}(\mathbf{u}) + \dots$$
(146)

Now substituting (the lower of) Eq. (116) into Eq. (145) gives

$$\ddot{\mathbf{u}} + \varepsilon \frac{\mathrm{d}^2 \mathbf{h}_1}{\mathrm{d}t^2} + \varepsilon^2 \frac{\mathrm{d}^2 \mathbf{h}_2}{\mathrm{d}t^2} = -\hat{\Lambda} \mathbf{u} - \varepsilon \hat{\Lambda} \mathbf{h}_1 - \varepsilon^2 \hat{\Lambda} \mathbf{h}_2 - \varepsilon \hat{\mathbf{n}}_1 (\mathbf{u} + \varepsilon \mathbf{h}_1 + \varepsilon^2 \mathbf{h} + \dots) - \varepsilon^2 \hat{\mathbf{n}}_2 (\mathbf{u} + \varepsilon \mathbf{h}_1 + \varepsilon^2 \mathbf{h} + \dots) + \dots$$
(147)

Then substituting for ü from Eq. (146) gives

$$-\hat{\Lambda}\mathbf{u} - \varepsilon \hat{\mathbf{n}}_{\mathbf{u}\mathbf{1}}(\mathbf{u}) - \varepsilon^{2} \hat{\mathbf{n}}_{\mathbf{u}\mathbf{2}}(\mathbf{u}) + \varepsilon \frac{d^{2} \mathbf{h}_{1}}{dt^{2}} + \varepsilon^{2} \frac{d^{2} \mathbf{h}_{2}}{dt^{2}} = -\hat{\Lambda}\mathbf{u} - \varepsilon \hat{\Lambda}\mathbf{h}_{1} - \varepsilon \hat{\Lambda}\mathbf{h}_{2} - \varepsilon \hat{\mathbf{n}}_{1}(\mathbf{u} + \varepsilon \mathbf{h}_{1} + \varepsilon^{2} \mathbf{h} + ...) - \varepsilon \hat{\mathbf{n}}_{2}(\mathbf{u} + \varepsilon \mathbf{h}_{1} + \varepsilon^{2} \mathbf{h} + ...) + ...$$
(148)

Using a Taylor series expansion in the same way as Eq. (123), we obtain

$$\hat{\mathbf{n}}_i(\mathbf{u} + \varepsilon \mathbf{h}_1(\mathbf{u}) + \dots) = \hat{\mathbf{n}}_i(\mathbf{u}) + \varepsilon D\{\hat{\mathbf{n}}_i(\mathbf{u})\}\mathbf{h}_1(\mathbf{u}) + \dots,$$
(149)

where  $D\{\hat{\mathbf{n}}_i\}$  is the Jacobian matrix (or gradient vector) depending on the dimensions of  $\mathbf{u}$ . Then equating  $\varepsilon$  terms in Eq. (148) gives

$$\varepsilon^{0}: \quad -\hat{\Lambda}\mathbf{u} = -\hat{\Lambda}\mathbf{u} \tag{150}$$

$$\boldsymbol{\varepsilon}^{1}: \quad -\frac{\mathrm{d}^{2}\mathbf{h}_{1}}{\mathrm{d}t^{2}} - \hat{\Lambda}\mathbf{h}_{1} = \hat{\mathbf{n}}_{1} - \hat{\mathbf{n}}_{u1}, \tag{151}$$

$$\varepsilon^{2}: -\frac{\mathrm{d}^{2}\mathbf{h}_{2}}{\mathrm{d}t^{2}} - \hat{\Lambda}\mathbf{h}_{2} = D\{\hat{\mathbf{n}}_{1}\}\mathbf{h}_{1} + \hat{\mathbf{n}}_{2} - \hat{\mathbf{n}}_{\mathbf{u}2},$$
(152)

Obtaining an expression for  $\frac{d^2 \mathbf{h}_j}{dt^2}$  will be achieved by differentiating the first Lie derivative from Eq. (127) a second time. The second Lie derivative of  $\mathbf{u}^*$  to order  $\varepsilon^0$  is then defined by<sup>50</sup>

$$\frac{\mathrm{d}^{2}\mathbf{u}^{*}}{\mathrm{d}t^{2}}\bigg|_{\varepsilon=0} = \frac{\partial}{\partial u} \left(\Lambda^{*}\mathbf{u}^{*}\right) \dot{u}\bigg|_{\varepsilon=0} = \Lambda^{*} \frac{\partial \mathbf{u}^{*}}{\partial u} \Lambda u = (\Lambda^{*})^{2} \mathbf{u}^{*}.$$
(153)

Substituting this, and relations for h from Eq. (117) into the homological equations Eq. (151) and Eq. (152) gives

$$\boldsymbol{\varepsilon}^{1}: \quad -\frac{\partial}{\partial \boldsymbol{u}} \left( \frac{\partial \mathbf{h}_{1}}{\partial \boldsymbol{u}} \Lambda \boldsymbol{u} \right) \Lambda \boldsymbol{u} - \hat{\Lambda} \mathbf{h}_{1} = \hat{\mathbf{n}}_{1} - \hat{\mathbf{n}}_{\mathbf{u}\mathbf{1}}, \qquad \qquad \qquad \qquad -\mathbf{b}_{1} \left( \Lambda^{*} \right)^{2} \mathbf{u}^{*} - \hat{\Lambda} \mathbf{b}_{1} \mathbf{u}^{*} = \hat{\mathbf{n}}_{1} - \hat{\mathbf{n}}_{\mathbf{u}\mathbf{1}}, \tag{154}$$

$$\varepsilon^{2}: -\frac{\partial}{\partial u} \left( \frac{\partial \mathbf{h}_{2}}{\partial u} \Lambda u \right) \Lambda u - \hat{\Lambda} \mathbf{h}_{2} = D\{\mathbf{\hat{n}}_{1}\} \mathbf{h}_{1} + \mathbf{\hat{n}}_{2} - \mathbf{\hat{n}}_{u2}, \quad \rightarrow \quad -\mathbf{b}_{2} \left(\Lambda^{+}\right)^{2} \mathbf{u}^{+} - \hat{\Lambda} \mathbf{b}_{2} \mathbf{u}^{+} = D\{\mathbf{\hat{n}}_{1}\} \mathbf{b}_{1} \mathbf{u}^{*} + \mathbf{\hat{n}}_{2} - \mathbf{\hat{n}}_{u2}, \quad (155)$$

As a result, the order- $\varepsilon^1$  homological equation (without frequency detuning which will be added in Section 4.6) for second-order equations is

$$\hat{\mathbf{n}}_{1} - \hat{\mathbf{n}}_{u1} = -\mathbf{b}_{1} \left(\Lambda^{*}\right)^{2} \mathbf{u}^{*} - \hat{\Lambda} \mathbf{b}_{1} \mathbf{u}^{*} = \hat{\mathscr{H}}_{1}(\mathbf{u}), \tag{156}$$

which is equivalent to the homological equation defined by [128] when they first introduced this type of transformation. An example is considered next.

<sup>&</sup>lt;sup>50</sup>Notice that the derivative is with respect to  $\boldsymbol{u}$  just as for first-order differential equations. The relationship is that for for each coordinate  $u_j$  in the vector  $\boldsymbol{u}$  there is a corresponding  $u_{pj}$  and  $u_{mj}$  in the vector  $\boldsymbol{u}$ , and  $\boldsymbol{u}^*$  is a function of  $u_{pj}$  and  $u_{mj}$ , for j = 1, 2, 3, ...N. In addition, there is a relationship between  $\hat{\Lambda}$  and  $\Lambda$  for undamped oscillators that can be inferred from Eq. (70) and Eq. (77).

## 4.2.4 Example 12: The real normal form for the undamped cubic-quintic oscillator

In this example, we follow the approach proposed by [128] which leads to a type of real normal form transformation<sup>51</sup>. First we note that the undamped cubic-quintic oscillator is the same as the Duffing equation (Eq. (132)) with an additional quintic polynomial term, that we write in the form of Eq. (145), and because this is a single degree-of-freedom example, x = q, therefore

$$\ddot{x} + \omega_n^2 x + \varepsilon \alpha_1 x^3 + \varepsilon \alpha_2 x^5 = 0 \quad \text{with } x = q \rightsquigarrow \qquad \ddot{q} + \omega_n^2 q + \varepsilon \alpha_1 q^3 + \varepsilon \alpha_2 q^5 = 0,$$
  
$$\rightsquigarrow \qquad \ddot{q} = -\omega_n^2 q - \varepsilon \hat{\mathbf{n}}_1(q_p, q_m) = 0,$$
(157)

where

$$\hat{\mathbf{n}}_{1}(q_{p},q_{m}) = \alpha_{1}(q_{p}+q_{m})^{3} + \alpha_{2}(q_{p}+q_{m})^{5} = \alpha_{1}(q_{p}^{3}+3q_{p}^{2}q_{m}+3q_{p}q_{m}^{2}+q_{m}^{3}) + \alpha_{2}(q_{p}^{5}+5q_{p}^{4}q_{m}+10q_{p}^{3}q_{m}^{2}+10q_{p}^{2}q_{m}^{3}+5q_{p}q_{m}^{4}+q_{m}^{5}) = [\alpha_{1} \ 3\alpha_{1} \ 3\alpha_{1} \ \alpha_{1} \ \alpha_{2} \ 5\alpha_{2} \ 10\alpha_{2} \ 5\alpha_{2} \ \alpha_{2}]\mathbf{q}^{*},$$

$$(158)$$

where  $\mathbf{q} = [q_p, q_m]^T$  and  $\mathbf{q}^* = [q_p^3 \ q_p^2 q_m \ q_p q_m^2 \ q_m^3 \ q_p^5 \ q_p^4 q_m \ q_p^3 q_m^2 \ q_p^2 q_m^3 \ q_p q_m^4 \ q_m^5]^T$ . Note that we are treating the combination of both the cubic and quintic nonlinear terms as just a single nonlinear term,  $\mathbf{\hat{n}}_1$ . This is possible in the  $\varepsilon$ -order approach, but not in the *k*-order approach, where instead we would take  $\mathbf{\hat{n}}_1 = \alpha_1 q^3$  and  $\mathbf{\hat{n}}_2 = \alpha_2 q^5$ . However, there are potential drawbacks to be aware of. Specifically, for this example there will be 5th order terms generated when the assumed solution is substituted into the cubic terms in  $\mathbf{\hat{n}}_1$  (in the terminology of Example 10 these are  $n_{(5)}^{(3)}$  terms), and these will only be captured in the  $\varepsilon^2$ -order analysis which we will show in Example 18 (Section 4.8.1) — see also Example 17 (Section 4.7.1).

Using this approach, we can set  $\mathbf{n}_2 = 0, ...$  etc. and so  $\mathbf{n}_1 = \mathbf{n}$ , and correspondingly  $\mathbf{b}_1 = \mathbf{b}$  and  $\mathbf{\hat{n}_{u1}} = \mathbf{\hat{n}_u}$  throughout the rest of this example. Next we apply a near-identity nonlinear transform, Eq. (116), of the form

$$q = u + \varepsilon \mathbf{b} \mathbf{u}^*$$
 where  $\mathbf{b} = [b_1 \ b_2 \ b_3 \ b_4 \ b_5 \ b_6 \ b_7 \ b_8 \ b_9 \ b_{10}],$  (159)

and  $\mathbf{u}^*$  is defined as  $\mathbf{u}^* = [u_p^3 \ u_p^2 u_m \ u_p u_m^2 \ u_m^3 \ u_p^5 \ u_p^4 u_m \ u_p^3 u_m^2 \ u_p^2 u_m^3 \ u_p u_m^4 \ u_m^5]^T$ . The transformed dynamic equation, Eq. (146), is expressed in the new coordinates as

$$\ddot{u} + \omega_n^2 u + \varepsilon \hat{\mathbf{n}}_u(u_p, u_m) = 0 \quad \rightsquigarrow \quad \ddot{u} = -\omega_n^2 u - \varepsilon \hat{\mathbf{n}}_u(u_p, u_m).$$
(160)

Substituting Eq. (159) and Eq. (160) into Eq. (157) leads the order- $\varepsilon^1$  homological equation  $\mathscr{H}_1(\mathbf{u})$  given in Eq. (156), which in this example (remembering that we have dropped the subscript 1 from  $\mathbf{n}$ ,  $\mathbf{b}$  and  $\hat{\mathbf{n}}_{\mathbf{u}}$ ) can be written as

$$\boldsymbol{\varepsilon}^{1}: \quad -\mathbf{b}\frac{\mathrm{d}^{2}\mathbf{u}^{*}}{\mathrm{d}t^{2}} - \boldsymbol{\omega}_{n}^{2}\mathbf{b}\mathbf{u}^{*} = \hat{\mathbf{n}} - \hat{\mathbf{n}}_{\mathbf{u}} = -\mathbf{b}\left(\boldsymbol{\Lambda}^{*}\right)^{2}\mathbf{u}^{*} - \boldsymbol{\omega}_{n}^{2}\mathbf{b}\mathbf{u}^{*}.$$
(161)

The Lie derivative term can be determined from Eq. (153), with  $\boldsymbol{u} = [u_p, u_m]^T$  and  $\Lambda$  as defined in Eq. (133). For this example we obtain

<sup>51</sup> In classifying this as a real normal form, we are following the definitions of [104] who introduced this terminology to distinguish between the *complex* normal form format such as [80] and the *real* normal form formats such as [128, 168, 169].

(162)

**u**\*.

(163)

Then using  $\hat{\mathbf{n}}_{\mathbf{u}} = [\hat{n}_{u1} \ \hat{n}_{u2} \ \hat{n}_{u3} \ \hat{n}_{u4} \ \hat{n}_{u5} \ \hat{n}_{u6} \ \hat{n}_{u7} \ \hat{n}_{u8} \ \hat{n}_{u9} \ \hat{n}_{u10}]\mathbf{u}^*$  and substituting Eq. (158), with  $q_i \rightarrow u_i$  (to order  $\varepsilon^0$ , similar to Eq. (136)) into Eq. (161) gives

 $([\alpha_1 \ 3\alpha_1 \ 3\alpha_1 \ \alpha_1 \ \alpha_2 \ 5\alpha_2 \ 10\alpha_2 \ 10\alpha_2 \ 5\alpha_2 \ \alpha_2] - [\hat{n}_{u1} \ \hat{n}_{u2} \ \hat{n}_{u3} \ \hat{n}_{u4} \ \hat{n}_{u5} \ \hat{n}_{u6} \ \hat{n}_{u7} \ \hat{n}_{u8} \ \hat{n}_{u9} \ \hat{n}_{u10}]) \mathbf{u}^* =$ 

$$\left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2 - \left(-(\mathbf{i}\omega_n)^2[b_1\ b_2\ b_3\ b_4\ b_5\ b_6\ b_7\ b_8\ b_9\ b_{10}]\right)^2$$

Evaluating the right hand side gives four zero values which allows the  $\hat{n}_u$  coefficients to be identified such that

$$([\alpha_1 \ 3\alpha_1 \ \alpha_1 \ \alpha_2 \ 5\alpha_2 \ 10\alpha_2 \ 5\alpha_2 \ \alpha_2] - [0 \ 3\alpha_1 \ 3\alpha_1 \ 0 \ 0 \ 10\alpha_2 \ 10\alpha_2 \ 0 \ 0]) \mathbf{u}^* = [b_1(8\omega_n^2) \ b_2(0) \ b_3(0) \ b_4(8\omega_n^2) \ b_5(24\omega_n^2) \ b_6(8\omega_n^2) \ b_7(0) \ b_8(0) \ b_9(8\omega_n^2) \ b_{10}(24\omega_n^2)] \mathbf{u}^*.$$
(164)

From which the transform coefficient values are  $b_1 = b_4 = \alpha_1/8\omega_n^2$ ,  $b_6 = b_9 = 5\alpha_2/8\omega_n^2$ ,  $b_5 = b_{10} = \alpha_2/24\omega_n^2$ , and  $b_2 = b_3 = b_7 = b_8 = 0$ . Note that  $b_2$ ,  $b_3 b_7$  and  $b_8$  are the free functions in this example, as any value can be selected without affecting the outcome of the homological equation. Here we choose to set the free functions to zero. Other choices will be discussed in Section 4.7. Next, using the fact that

 $\varepsilon \hat{\mathbf{n}}_{\mathbf{u}}(u_p, u_m) = \varepsilon \begin{bmatrix} 0 & 3\alpha_1 & 3\alpha_1 & 0 & 0 & 10\alpha_2 & 10\alpha_2 & 0 & 0 \end{bmatrix} \mathbf{u}^* = \varepsilon 3\alpha_1(u_p^2 u_m + u_p u_m^2) + \varepsilon 10\alpha_2(u_p^3 u_m^2 + u_p^2 u_m^3)$ 

gives the  $\varepsilon^1$  normal form as

$$\ddot{u} + \omega_n^2 u + \varepsilon 3\alpha_1 (u_p^2 u_m + u_p u_m^2) + \varepsilon 10\alpha_2 (u_p^3 u_m^2 + u_p^2 u_m^3) = 0,$$
(165)

where  $u = u_p + u_m$ , and this type of normal form can be expressed in terms of the  $u_p$  and  $u_m$  coordinates separately as

$$\ddot{u}_{p} + \omega_{n}^{2} u_{p} + \varepsilon 3 \alpha_{1} u_{p}^{2} u_{m} + \varepsilon 10 \alpha_{2} (u_{p}^{3} u_{m}^{2} + u_{p}^{2} u_{m}^{3}) = 0,$$

$$\ddot{u}_{m} + \omega_{n}^{2} u_{m} + \varepsilon 3 \alpha_{1} u_{p} u_{m}^{2} + \varepsilon 10 \alpha_{2} (u_{p}^{3} u_{m}^{2} + u_{p}^{2} u_{m}^{3}) = 0.$$
(166)

Note that the analysis needs to be extended to  $\varepsilon^2$  to capture the effect of the 5*th*-order term generated by the cubic nonlinearity — see Example 18 and Eq. (303). The near-identity transform, to order  $\varepsilon^1$  may now be written as

$$x = u + \varepsilon \mathbf{b}\mathbf{u}^* = \frac{U}{2} (\mathbf{e}^{\mathbf{i}(\omega_r t)} + \mathbf{e}^{-\mathbf{i}(\omega_r t)}) + \varepsilon \left[ \frac{\alpha_1}{8\omega_n^2} \ 0 \ 0 \ \frac{\alpha_1}{8\omega_n^2} \ \frac{\alpha_2}{24\omega_n^2} \ \frac{5\alpha_2}{8\omega_n^2} \ 0 \ 0 \ \frac{5\alpha_2}{8\omega_n^2} \ \frac{\alpha_2}{24\omega_n^2} \right] \mathbf{u}^*$$
(167)

using the base solutions from Eq. (22) in the  $u = u_p + u_m$  term. Then substituting the base solutions from Eq. (22) into the  $\mathbf{u}^*$  vector, the solution for *x* becomes

$$x = U\cos(\omega_r t) + \varepsilon \left(\frac{\alpha_1 U^3}{32\omega_n^2} + \frac{5\alpha_2 U^5}{128\omega_n^2}\right)\cos(3\omega_r t) + \varepsilon \frac{\alpha_2 U^5}{384\omega_n^2}\cos(5\omega_r t) + \mathcal{O}(\varepsilon^2).$$
 (168)

172 | doi:10.25518/2684-6500.84



**Fig. 5:** A simulation of the undamped Duffing oscillator showing (a) the solution, *x*, and (b) the phase portrait. The normal form solutions are from Example 12 (complex normal form shown with red crosses), and Example 13 (real normal form shown with green circles). These normal form solutions are compared with a reference solution generated from a fourth-order Runge-Kutta integration (the 'odeint' numerical integration routine from the Python numerical library — blue solid line). The complex normal form solution is computed using Eq. (143) with Eq. (144) (red crosses) and the real normal form solution is computed using Eq. (143) with Eq. (144) (red crosses) and the real normal form solution is computed with Eq. (169) with  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  (green circles). These solutions are differentiated to get  $\dot{x}$  values. Parameter values are: x(0) = 0.01,  $\omega_n = \sqrt{10}$  rad/s, and  $\alpha_1 = \alpha = -10$  N/(kgm<sup>3</sup>)  $\alpha_2 = 0$ .

In order to make a comparison with Example 11, we set  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$ . Then if this solution for *x*, Eq. (168), is compared to Eq. (143) from Example 11, it can be noted that there is no correction to the  $\cos(\omega_n t)$  term (sometimes also called the fundamental term or the  $\varepsilon^0$  term). In this normal form, each order of  $\varepsilon$  adds an additional term in the series without adding a correction to the fundamental term (although corrections are added to other terms), a feature called *killing the fundamental* meaning that the higher-order corrections are *orthogonal* to the fundamental part of the solution [85].

Next, in order to compute the *x* approximation from Eq. (168), we first need to find the initial condition value of *U* (which we denote  $U_0$  at time t = 0). Substituting t = 0,  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  into Eq. (168) and using Cardano's formula leads to

$$x(0) = U_0 + \varepsilon \frac{\alpha U^3}{32\omega_n^2}, \quad \forall \quad U_0 = \sqrt[3]{-\frac{\Gamma}{2} + \sqrt{\frac{\Gamma^2}{4} + \frac{\Pi^3}{27}}} + \sqrt[3]{-\frac{\Gamma}{2} - \sqrt{\frac{\Gamma^2}{4} + \frac{\Pi^3}{27}}}, \quad \text{where } \Pi = \frac{32\omega_n^2}{\varepsilon\alpha}, \quad (169)$$

and  $\Gamma = -\Pi x(0)$ . Note that Cardano's formula applies when the discriminant is positive meaning  $4\Pi^3 + 27\Gamma^2 > 0$ . A numerical example is shown in Fig. 5.

It can be noted from Fig. 5 that the complex and real normal form solutions give very similar results (note that the comparison is made with  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  in Fig. 5). The equivalence of these two solutions in discussed in Section 4.2.5. In addition, the  $\varepsilon^2$ -order analysis for this example (with frequency detuning) is given in Examples 17 (Section 4.7.1) and 18 (Section 4.8.1). For example, the solutions for *x* from this example, Eq. (168), can be compared to Eq. (294) (minimal normal form) and Eq. (305) (non-minimal normal form).

### 4.2.5 Equivalence of complex and real normal form

In order to make a comparison between the complex and real normal form, we consider the example of the Duffing oscillator. In order to do this, we make a comparison between Eq. (166) with  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  (so the result reduces to that of the Duffing oscillator) and the results from Example 11 (Section 4.2.2). To do this, first notice that differentiating the normal form equations for the complex Duffing normal form, (Example 11, Eq. (141)) with respect to time gives

$$\ddot{u}_{p} = \mathbf{i}\omega_{n}\dot{u}_{p} + \varepsilon \frac{3\mathbf{i}\alpha}{2\omega_{n}} \frac{\mathrm{d}}{\mathrm{d}t}(u_{p}^{2}u_{m}),$$

$$\ddot{u}_{m} = -\mathbf{i}\omega_{n}\dot{u}_{m} - \varepsilon \frac{3\mathbf{i}\alpha}{2\omega_{n}} \frac{\mathrm{d}}{\mathrm{d}t}(u_{p}u_{m}^{2}).$$
(170)

173 | doi:10.25518/2684-6500.84
Now eliminating the  $\dot{u}_p$  and  $\dot{u}_m$  by substituting from Eq. (141) (and using operator representation Section 1.1.2) leads to

Finally it can be seen that adding the two expressions on the right of Eq. (171) gives

$$\ddot{u}_p + \ddot{u}_m + \omega_n^2 (u_p + u_m) + \varepsilon 3\alpha (u_p^2 u_m + u_p u_m^2) = 0 \quad \rightsquigarrow \quad \ddot{u} + \omega_n^2 u + \varepsilon 3\alpha (u_p^2 u_m + u_p u_m^2) = 0$$
(172)

which is the same as the real normal form expression from Eq. (165) when  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  are introduced.

The solutions for *x* between the complex and real normal form for the Duffing oscillator can be made the same using a specific choice of the free functions. This will be described in more detail in Section 4.4.2. Having shown the equivalence of the complex and real normal form, we will focus on the real, and Hamiltonian normal form for the remainder of this paper.

### 4.3 Derivation of the homological equations using Lie series

In the notation of this paper<sup>52</sup> the Lie series can be defined in the following way (see [22, 120] for detailed derivations). Let  $\Theta[\mathbf{q}]$  be an arbitrary function of  $\mathbf{q}$ , then the Lie series is

$$\Theta[\mathbf{q}] = \Theta[\mathbf{u}] + \varepsilon(G\Theta) + \frac{\varepsilon^2}{2!}(G(G\Theta)) + \frac{\varepsilon^3}{3!}(G(G(G\Theta))) + \dots$$
(173)

with group operator

$$\mathbf{G} = \frac{\partial \cdot}{\partial \mathbf{u}} \mathbf{h} \Big|_{\boldsymbol{\varepsilon} = 0}.$$
 (174)

Notice that from Eq. (173) that the order- $\varepsilon^1$  terms satisfy

$$(\Theta[\mathbf{q}] - \Theta[\mathbf{u}])_{\varepsilon^1} \approx G\Theta \tag{175}$$

which when  $\mathbf{q}$  is substituted, using the near-identity transformation, Eq. (116), is the order  $\varepsilon^1$  homological equation from the method of normal forms discussed in detail in Section 4.2. Furthermore, the term G $\Theta$  corresponds to either  $\mathscr{H}_1$  for first-order differential equations or  $\mathscr{H}_1$  for second-order differential equations, as will be demonstrated later in this Section.

In order to consider Lie series transformations of both first and second-order differential equations, the functions  $\Theta$  will need to include the variables *t*, *q*, and *q*, in the first-order case, and *t*, *q*, *q* and *q* in the second-order case. The method for *prolonging* the group transformation to include these variables is described in detail by [22]. We now use this analysis to derive the homological equations for both first- and second-order differential equations.

#### 4.3.1 First-order equations

First let us consider first-order equations in the format of Eq. (119), i.e.

$$\frac{\mathrm{d}\boldsymbol{q}}{\mathrm{d}t} = \mathscr{F}(\boldsymbol{q}, \boldsymbol{\varepsilon}) \quad \text{which can be redefined as} \quad \dot{\boldsymbol{q}} - \mathscr{F}(\boldsymbol{q}, \boldsymbol{\varepsilon}) = \Theta_1(\boldsymbol{q}, \dot{\boldsymbol{q}}, \boldsymbol{\varepsilon}) = 0. \tag{176}$$

Now we can consider a transformation of this first-order differential equation using the method of Lie series<sup>53</sup>. To do this we need to revisit the near-identity transformation given in Eq. (116). First we write the differentials of *t* and *q* as

$$dt = dt, \text{ and } d\boldsymbol{q} = \frac{\partial \boldsymbol{T}}{\partial t} dt + \frac{\partial \boldsymbol{T}}{\partial \boldsymbol{u}} d\boldsymbol{u}, \quad \rightsquigarrow \quad \frac{d\boldsymbol{q}}{dt} = \frac{\partial \boldsymbol{T}}{\partial t} + \frac{\partial \boldsymbol{T}}{\partial \boldsymbol{u}} \frac{d\boldsymbol{u}}{dt}$$
(177)

 $<sup>^{52}</sup>$ We use  ${\bf q}$  etc. in this expression, but the same relationships apply for q and so on.  $^{53}$ We follow the analysis from [22].

Noting that T is defined in Eq. (116), and is not an explicit function of t (so that  $\partial T/\partial t = 0$ ) and substituting  $T(u, \varepsilon) = u + \varepsilon h(u)$  means that Eq. (177) becomes<sup>54</sup>

$$\dot{\boldsymbol{q}} = \frac{\partial}{\partial \boldsymbol{u}} (\boldsymbol{u} + \boldsymbol{\varepsilon} \boldsymbol{h}) \dot{\boldsymbol{u}} = \dot{\boldsymbol{u}} + \boldsymbol{\varepsilon} \frac{\partial \boldsymbol{h}}{\partial \boldsymbol{u}} \dot{\boldsymbol{u}} = \dot{\boldsymbol{u}} + \boldsymbol{\varepsilon} \boldsymbol{b} \Lambda^* \mathbf{u}^* = \boldsymbol{T}^{\dagger} (\boldsymbol{u}, \dot{\boldsymbol{u}}, \boldsymbol{\varepsilon}),$$
(178)

where we have substituted  $h = bu^*$  and used the Lie derivative from Eq. (127) to evaluate the partial derivative term. Now Eq. (178) can be added to Eq. (116) as an extension or *prolongation* of the variables being transformed, so that

$$t = t,$$

$$q = T(u,\varepsilon) = u + \varepsilon b \mathbf{u}^*,$$
(179)

$$\dot{\boldsymbol{q}} = \boldsymbol{T}^{\dagger}(\boldsymbol{u}, \dot{\boldsymbol{u}}, \varepsilon) = \dot{\boldsymbol{u}} + \varepsilon \boldsymbol{b} \Lambda^* \mathbf{u}^*.$$

Now we will define new augmented vectors as

$$\tilde{\boldsymbol{q}} = \begin{bmatrix} \boldsymbol{q} \\ \dot{\boldsymbol{q}} \end{bmatrix}, \quad \tilde{\boldsymbol{u}} = \begin{bmatrix} \boldsymbol{u} \\ \dot{\boldsymbol{u}} \end{bmatrix}, \quad \tilde{\boldsymbol{h}} = \begin{bmatrix} \boldsymbol{h} \\ \boldsymbol{h}^{\dagger} \end{bmatrix}, \quad \tilde{\boldsymbol{T}} = \begin{bmatrix} \boldsymbol{T} \\ \boldsymbol{T}^{\dagger} \end{bmatrix}, \quad (180)$$

to give a near-identity transformation in the new variables as

 $\tilde{q} = \tilde{T}(\tilde{u}, \varepsilon) = \tilde{u} + \varepsilon \tilde{h}$ (181)

where  $h = b\mathbf{u}^*$  and from Eq. (127)

$$\boldsymbol{h}^{\dagger} = \frac{\mathrm{d}\boldsymbol{h}}{\mathrm{d}t} = \boldsymbol{b}\frac{\mathrm{d}\mathbf{u}^*}{\mathrm{d}t} = \boldsymbol{b}\Lambda^*\mathbf{u}^*. \tag{182}$$

Next the augmented variables can be substituted into Eq. (174), so that the group operator becomes

$$G_{FO}\Theta_1 = \frac{\partial \Theta_1}{\partial \tilde{\boldsymbol{u}}} \tilde{\boldsymbol{h}} = \frac{\partial \Theta_1}{\partial \boldsymbol{u}} \boldsymbol{h} + \frac{\partial \Theta_1}{\partial \tilde{\boldsymbol{u}}} \boldsymbol{h}^{\dagger},$$
(183)

where  $G_{FO}$  is the Lie group operator for first-order equations and  $\frac{\partial \Theta_1}{\partial u}$  and  $\frac{\partial \Theta_1}{\partial \dot{u}}$  are  $N \times N$  Jacobian matrices. Note also that  $\varepsilon = 0$  is used when evaluating the group operator, so from Eq. (179) we have that q = u and  $\dot{q} = \dot{u}$  in the evaluation of  $G_{FO}$ . This means that  $\Theta_1(q, \dot{q}, 0) = \Theta_1(u, \dot{u}, 0) = \dot{u} - \mathscr{F}(u, 0)$ .

Evaluating the partial derivatives of  $\Theta_1$  for the group operator with  $\varepsilon = 0$  gives

$$\frac{\partial \Theta_1}{\partial \boldsymbol{u}} = -\frac{\partial \mathscr{F}}{\partial \boldsymbol{u}}, \quad \frac{\partial \Theta_1}{\partial \dot{\boldsymbol{u}}} = 1,$$
(184)

so then

$$G_{FO}\Theta_1 = -\frac{\partial \mathscr{F}}{\partial \boldsymbol{u}}\boldsymbol{h} + \boldsymbol{h}^{\dagger} = -\Lambda \boldsymbol{b}\boldsymbol{u}^* + \boldsymbol{b}\Lambda^*\boldsymbol{u}^* = \mathscr{H}_1, \quad \text{where} \quad \mathscr{F}(\boldsymbol{u},0) = \Lambda \boldsymbol{u}$$
(185)

gives the Lie bracket part of the order- $\varepsilon^1$  homological equation. The complete homological equation can then be obtained by substituting Eq. (185) and Eq. (179) into Eq. (175). The resulting expressions can then be compared with Eq. (131) obtained from the previous normal form techniques.

<sup>&</sup>lt;sup>54</sup>Note that throughout this Section we are assuming that  $n_2 = 0, ...$  etc. and so we use  $n_1 = n$ , and correspondingly  $h_1 = h$ ,  $b_1 = b$  and  $\hat{n}_{u1} = \hat{n}_u$ . Similarly for the analysis of second-order equations.

#### 4.3.2 Second-order equations

Now let us consider equations of second-order written in the form of Eq. (145) so that

$$\frac{d^2\mathbf{q}}{dt^2} = \hat{\mathscr{F}}(\mathbf{q}, \dot{\mathbf{q}}, \varepsilon) \quad \text{which can be redefined as} \quad \ddot{\mathbf{q}} - \hat{\mathscr{F}}(\mathbf{q}, \dot{\mathbf{q}}, \varepsilon) = \Theta_2(\mathbf{q}, \dot{\mathbf{q}}, \ddot{\mathbf{q}}, \varepsilon) = 0.$$
(186)

The group can now be prolonged again using the same approach as above (but with  $\mathbf{q}$  instead of  $\mathbf{q}$  etc.). First the differential of  $\dot{\mathbf{q}}$  is written as

$$d\dot{\mathbf{q}} = \frac{\partial \mathbf{T}^{\dagger}}{\partial t} dt + \frac{\partial \mathbf{T}^{\dagger}}{\partial \mathbf{u}} d\mathbf{u} + \frac{\partial \mathbf{T}^{\dagger}}{\partial \dot{\mathbf{u}}} d\dot{\mathbf{u}}.$$
(187)

The function  $\mathbf{T}^{\dagger}$  comes from the previously prolongated term in Eq. (178) and because  $\mathbf{T}^{\dagger}$  is not a function of *t*, then  $\partial \mathbf{T}^{\dagger}/\partial t = 0$ . Now setting  $\partial \mathbf{T}^{\dagger}/\partial t = 0$  and dividing Eq. (187) by d*t* gives

$$\ddot{\mathbf{q}} = \frac{\partial \mathbf{T}^{\dagger}}{\partial \mathbf{u}} \dot{\mathbf{u}} + \frac{\partial \mathbf{T}^{\dagger}}{\partial \dot{\mathbf{u}}} \ddot{\mathbf{u}}.$$
(188)

Substituting  $\mathbf{T}^{\dagger} = \dot{\mathbf{u}} + \varepsilon \mathbf{b} \Lambda^* \mathbf{u}^*$  (i.e. from Eq. (178)) leads to

$$\ddot{\mathbf{q}} = \varepsilon \mathbf{b} \Lambda^* \left( \frac{\partial \mathbf{u}^*}{\partial \mathbf{u}} \right) \dot{\mathbf{u}} + (I+0) \, \ddot{\mathbf{u}} \rightsquigarrow \ddot{\mathbf{q}} = \ddot{\mathbf{u}} + \varepsilon \mathbf{b} \Lambda^{*2} \mathbf{u}^*.$$

So finally the prolonged group is

$$t = t,$$

$$\mathbf{q} = \mathbf{T}(\mathbf{u}, \varepsilon) = \mathbf{u} + \varepsilon \mathbf{b} \mathbf{u}^*,$$

$$\dot{\mathbf{q}} = \mathbf{T}^{\dagger}(\mathbf{u}, \dot{\mathbf{u}}, \varepsilon) = \dot{\mathbf{u}} + \varepsilon \mathbf{b} \Lambda^* \mathbf{u}^*,$$

$$\ddot{\mathbf{q}} = \mathbf{T}^{\ddagger}(\mathbf{u}, \dot{\mathbf{u}}, \ddot{\mathbf{u}}, \varepsilon) = \ddot{\mathbf{u}} + \varepsilon \mathbf{b} \Lambda^{*2} \mathbf{u}^*.$$
(189)

Now define new augmented vectors as

$$\check{\mathbf{q}} = \begin{bmatrix} \mathbf{q} \\ \dot{\mathbf{q}} \\ \dot{\mathbf{q}} \end{bmatrix}, \quad \check{\mathbf{u}} = \begin{bmatrix} \mathbf{u} \\ \dot{\mathbf{u}} \\ \ddot{\mathbf{u}} \end{bmatrix}, \quad \check{\mathbf{h}} = \begin{bmatrix} \mathbf{h} \\ \mathbf{h}^{\dagger} \\ \mathbf{h}^{\ddagger} \end{bmatrix}, \quad \check{\mathbf{T}} = \begin{bmatrix} \mathbf{T} \\ \mathbf{T}^{\dagger} \\ \mathbf{T}^{\ddagger} \end{bmatrix},$$
(190)

such that the second, third and fourth equations in Eq. (189) can be combined to give a near-identity transformation in the new variables as

$$\check{\mathbf{q}} = \mathbf{\hat{T}}(\check{\mathbf{u}}, \varepsilon) = \check{\mathbf{u}} + \varepsilon \check{\mathbf{h}},\tag{191}$$

where from above we know that  $\mathbf{h} = \mathbf{b}\mathbf{u}^*$ ,  $\mathbf{h}^{\dagger} = \mathbf{b}\Lambda^*\mathbf{u}^*$ , and from Eq. (153)

$$\mathbf{h}^{\ddagger} = \frac{d\mathbf{h}^{\dagger}}{dt} = \mathbf{b}\Lambda^* \frac{d\mathbf{u}^*}{dt} = (\Lambda^*)^2 \mathbf{u}^*.$$
(192)

Next the augmented variables can be substituted into Eq. (174), so that the group operator becomes

$$G_{SO}\Theta_2 = \frac{\partial\Theta_2}{\partial\check{\mathbf{u}}}\check{\mathbf{h}} = \frac{\partial\Theta_2}{\partial\mathbf{u}}\mathbf{h} + \frac{\partial\Theta_2}{\partial\dot{\mathbf{u}}}\mathbf{h}^{\dagger} + \frac{\partial\Theta_2}{\partial\ddot{\mathbf{u}}}\mathbf{h}^{\ddagger}.$$
(193)

Evaluating the partial derivatives of  $\Theta_2$  gives

$$\frac{\partial \Theta_2}{\partial \mathbf{u}} = -\frac{\partial \hat{\mathscr{F}}}{\partial \mathbf{u}}, \quad \frac{\partial \Theta_2}{\partial \dot{\mathbf{u}}} = -\frac{\partial \hat{\mathscr{F}}}{\partial \dot{\mathbf{u}}}, \quad \frac{\partial \Theta_2}{\partial \ddot{\mathbf{u}}} = 1,$$
(194)

*Journal of Structural Dynamics*, 1, (pp. 138-216) 2022 Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.

so then

$$G_{SO}\Theta_2 = -\frac{\partial\widehat{\mathscr{F}}}{\partial \mathbf{u}}\mathbf{h} - \frac{\partial\widehat{\mathscr{F}}}{\partial\dot{\mathbf{u}}}\mathbf{h}^{\dagger} + \mathbf{h}^{\ddagger} = \hat{\Lambda}\mathbf{b}\mathbf{u}^* + \mathbf{b}(\Lambda^*)^2\mathbf{u}^* = -\hat{\mathscr{H}}_1,$$
(195)

because

$$\hat{\mathscr{F}}(\mathbf{q},\varepsilon) = \hat{\mathscr{F}}(\mathbf{T}(\mathbf{u},\varepsilon),\varepsilon)|_{\varepsilon=0} = \hat{\Lambda}\mathbf{u} \quad \rightsquigarrow \quad \frac{\partial\hat{\mathscr{F}}}{\partial \mathbf{u}}\Big|_{\varepsilon=0} = -\hat{\Lambda} \quad \text{and} \quad \frac{\partial\hat{\mathscr{F}}}{\partial \dot{\mathbf{u}}}\Big|_{\varepsilon=0} = 0.$$
(196)

Note that this analysis works providing all terms relating to  $\dot{\mathbf{q}}$  and  $\dot{\mathbf{u}}$  are included in the nonlinear vector of the equation of motion. i.e. from Eq. (145)  $\ddot{\mathbf{q}} = \hat{\mathscr{F}}(\mathbf{q}, \dot{\mathbf{q}}, \varepsilon) = -\hat{\Lambda}\mathbf{q} - \varepsilon \hat{\mathbf{n}}(\mathbf{q}, \dot{\mathbf{q}})$ . If damping terms relating to  $\dot{\mathbf{q}}$  were included at  $\varepsilon^0$ , then this would change the homological equation, because  $\frac{\partial \hat{\mathscr{F}}}{\partial \dot{\mathbf{u}}}\Big|_{\varepsilon=0} \neq 0$ . This is the reason that damping terms are recommend to be included in the nonlinear vector in Section 4.5.1 for second-order differential equations. However, this restriction does not apply to first order differential equations.

It's interesting to note that Eq. (195) gives  $-\hat{\mathcal{H}}_1$  whereas in the first-order case we obtained  $\mathcal{H}_1$  directly — see Eq. (185). The reason is that for the definition of  $\Theta_2$  given in Eq. (186), the term  $(\Theta[\mathbf{q}] - \Theta[\mathbf{u}])_{\varepsilon^1} = \mathbf{n}_u - \mathbf{n}$  rather than  $\mathbf{n} - \mathbf{n}_u$  as given in the homological equation for second order equations Eq. (156). If  $\Theta_2$  is redefined as  $\Theta_2 = \hat{\mathcal{F}} - \ddot{\mathbf{q}} = 0$  then then Eq. (195) becomes equal to  $+\hat{\mathcal{H}}_1$ . In that case the homological equation given in Eq. (195) becomes the same as Eq. (156). Next we consider the Hamiltonian Normal form first proposed by Birkhoff [11].

#### 4.4 Hamiltonian normal form

A standard result from classical mechanics (see for example [117, 54], and also [116, 120]) states that<sup>55</sup>: If the Hamiltonian, H, of a nonlinear oscillator system can be written in the form

$$H(\mathbf{q}) = H_0(\mathbf{q}) + \varepsilon H_1(\mathbf{q}) + \mathscr{O}(\varepsilon^2)$$
(197)

then it is possible to define a coordinate transformation from the original coordinate set  $\mathbf{q}$  to a new set of coordinates  $\mathbf{u}$  so that there is a new Hamiltonian function, called the Kamiltonian, given by

$$K(\mathbf{u}) = H_0(\mathbf{u}) + \varepsilon(H_1(\mathbf{u}) + \{H_0, G\}) + \mathscr{O}(\varepsilon^2),$$
(198)

using the near-identity transformation

$$\mathbf{q} = \mathbf{u} + \boldsymbol{\varepsilon} \{ \mathbf{u}, G \} + \mathcal{O}(\boldsymbol{\varepsilon}^2), \tag{199}$$

where *G* is a generator function,  $\varepsilon$  is a series expansion parameter, and  $\{f,g\}$  denotes the Poisson bracket for two arbitrary functions *f* and *g* 

$$\{f,g\} = \frac{\partial f}{\partial q} \frac{\partial g}{\partial p} - \frac{\partial f}{\partial p} \frac{\partial g}{\partial q}$$
 where  $q$  and  $p$  are some generalised coordinates. (200)

In the non-resonant case,  $\{H_0, G\} = 0$ , which is often referred to as the Birkhoff normal form [11, 6]. In contrast, the solution to the resonant case when,  $\{H_0, G\} \neq 0$ , is attributed to Gustavson [57] and as a result is called the Birkhoff-Gustavson normal form [38]. The question of how to choose an appropriate generating function has a long history, and the interested reader can find full details of this in a number of papers, e.g. [74, 34, 117, 37, 24, 38, 95, 174] and the textbooks [116, 120].

#### 4.4.1 Hamiltonian to Kamiltonian transformation

The approach taken here will be to define the Kamiltonian in terms of a residual function  $R(\mathbf{u})$  such that

$$K(\mathbf{u}) = H_0(\mathbf{u}) + \varepsilon R(\mathbf{u}),$$

(201)

<sup>&</sup>lt;sup>55</sup>A demonstration of this result for the Duffing example is given later in Section 4.4.2.

which at order  $\varepsilon^1$  leads to the relationship

$$R(\mathbf{u}) = H_1(\mathbf{u}) + \{H_0, G\} \rightsquigarrow \{H_0, G\} = R(\mathbf{u}) - H_1(\mathbf{u}),$$
(202)

which is the order- $\varepsilon^1$  homological equation in Hamiltonian form. To see how this result arises, we start by substituting Eq. (199) into Eq. (197), so that

$$H(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) = H_0(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) + \varepsilon H_1(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) + \mathcal{O}(\varepsilon^2),$$
(203)

which when the right-hand-side is expanded gives to order  $\varepsilon^1$ 

$$K(\mathbf{u}) = H(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) = H_0(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) + \varepsilon H_1(\mathbf{u}) + \mathcal{O}(\varepsilon^2).$$
(204)

Now consider the  $H_0$  term on the right-hand-side. Birkhoff [11] showed that for a single degree-of-freedom system<sup>56</sup>  $H_0(\mathbf{q}) = \rho q_p q_m$  where  $\rho$  is a constant related to the eigenvalues (a specific example is given in Eq. (212)) and  $\mathbf{q} = [q_p \ q_m]^T$  are the generalised coordinates. As a result, using Eq. (199) with  $\mathbf{u} = [u_p \ u_m]^T$  we have

$$H_0(\mathbf{q}) \implies H_0(\mathbf{u} + \varepsilon \{\mathbf{u}, G\}) = \rho q_p q_m = \rho \left( u_p + \varepsilon \frac{\partial G}{\partial u_m} \right) \left( u_m - \varepsilon \frac{\partial G}{\partial u_p} \right), \tag{205}$$

which gives to order  $\varepsilon^1$ 

$$H_0(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) = \rho \left( u_p u_m + \varepsilon \left( u_m \frac{\partial G}{\partial u_m} - u_p \frac{\partial G}{\partial u_p} \right) \right) + \mathcal{O}(\varepsilon^2).$$
(206)

Then because  $H_0(\mathbf{u}) = \rho u_p u_m$  we can write

$$H_0(\mathbf{u} + \boldsymbol{\varepsilon}\{\mathbf{u}, G\}) = \rho \left( u_p u_m + \frac{\boldsymbol{\varepsilon}}{\rho} \left( \frac{\partial H_0}{\partial u_p} \frac{\partial G}{\partial u_m} - \frac{\partial H_0}{\partial u_m} \frac{\partial G}{\partial u_p} \right) \right) + \mathcal{O}(\boldsymbol{\varepsilon}^2),$$
(207)

which can be written as

$$H_0(\mathbf{u} + \varepsilon\{\mathbf{u}, G\}) = \rho u_p u_m + \varepsilon\{H_0, G\} + \mathscr{O}(\varepsilon^2), = H_0(\mathbf{u}) + \varepsilon\{H_0, G\}) + \mathscr{O}(\varepsilon^2)$$
(208)

which when substituted into Eq. (204) gives Eq. (198).

#### 4.4.2 Example 13 Hamiltonian normal form for the conservative Duffing oscillator

Normally the coordinate displacement q, and momentum  $p = m\dot{q}$  are used for the Hamiltonian, but to avoid confusion with coordinates  $q_p$  and  $q_m$ , here we use coordinates x for the displacement and  $p = m\dot{x}$  for the momentum for the undamped, unforced (i.e. conservative) Duffing oscillator given by Eq. (132) (see Example 11 Section 4.2.2). Then we can rewrite the equations of motion in the Hamiltonian form as

$$\dot{x} = \frac{\partial H}{\partial p} = \frac{p}{m},$$

$$\dot{p} = -\frac{\partial H}{\partial x} = -kx - k_3 x^3,$$
(209)

with the Hamiltonian

$$H(\mathbf{p}) = \frac{p^2}{2m} + \frac{kx^2}{2} + \varepsilon \frac{k_3 x^4}{4} = H_0 + \varepsilon H_1,$$
(210)

where, *m* is mass, *k* linear stiffness,  $k_3$  nonlinear stiffness,  $\varepsilon$  is a small parameter,  $\alpha = k_3/m$ , and  $\omega_n^2$  is the linear natural frequency and  $\mathbf{p} = [x, p]^T$ .

<sup>&</sup>lt;sup>56</sup>This can be generalised to systems of *N*-degrees-of-freedom providing that the *i*<sup>th</sup>  $H_{i0}(\mathbf{q})$  function has the format  $H_{i0}(\mathbf{q}) = \rho_i q_{ip} q_{im}$  for all  $i \in N$ , see [11]. However, and important detail to note is that for each degree-of-freedom, the result relies on  $k = m\omega_n$  which only works in a straightforward way for the case without detuning (shown here). In the case when detuning is added, additional frequency terms will be generated at order  $\varepsilon^1$ . Examples of systems with more than one degree-of-freedom can be found in [38, 174].

Before we can use the result from Section 4.4.1, we need to apply a *diagonalisation transform* to get the equations into the form of Eq. (197). The usual choice for a diagonalisation transform is related directly to the modal transformation defined in Eq. (4)<sup>57</sup>. For consistency with the engineering approach we use the change of coordinates defined by Eq. (4) above. As a result the diagonalisation transformation of the Hamiltonian  $H(\mathbf{p}) \rightarrow H(\mathbf{q})$ , will be such that  $x = q_p + q_m$  and  $p = mi\omega_n(q_p - q_m)$ , which can be written using Eq. (4) as

$$\begin{bmatrix} x \\ p \end{bmatrix} = \begin{bmatrix} 1 & 1 \\ m\lambda_1 & m\lambda_2 \end{bmatrix} \begin{bmatrix} q_p \\ q_m \end{bmatrix}.$$
 (211)

Note that to use the Birkhoff form of  $H_0$  we do not use detuning, and so  $p = mi\omega_n(q_p - q_m)$  is assumed. Substituting these relationships into the Hamiltonian Eq. (210) gives

$$H(\mathbf{q}) = 2m\omega_n^2 q_p q_m + \varepsilon \frac{k_3}{4} (q_p + q_m)^4 = H_0(\mathbf{q}) + \varepsilon H_1(\mathbf{q}),$$
(212)

from which we note that  $\rho = 2m\omega_n^2$  in this example. When the  $H_1$  term is expanded we obtain

$$H_{1}(\mathbf{q}) = \frac{k_{3}}{4} \left( q_{p}^{4} + 4q_{p}^{3}q_{m} + 6q_{p}^{2}q_{m}^{2} + 4q_{p}q_{m}^{3} + q_{m}^{4} \right) = \frac{k_{3}}{4} \begin{bmatrix} 1 & 4 & 6 & 4 & 1 \end{bmatrix} \begin{bmatrix} q_{p}^{4} \\ q_{p}^{3}q_{m} \\ q_{p}^{2}q_{m}^{2} \\ q_{p}q_{m}^{3} \\ q_{m}^{4} \end{bmatrix} = \mathbf{H}_{1}^{*}\mathbf{q}^{\dagger},$$
(213)

where

$$\mathbf{H}_{1}^{*} = \frac{k_{3}}{4} \begin{bmatrix} 1 & 4 & 6 & 4 & 1 \end{bmatrix} \text{ and } \mathbf{q}^{\dagger} = \begin{bmatrix} q_{p}^{4} \\ q_{p}^{3}q_{m} \\ q_{p}^{2}q_{m}^{2} \\ q_{p}q_{m}^{3} \\ q_{m}^{4} \end{bmatrix}.$$
(214)

The next step is to make the coordinate transformation from  $q \rightarrow u$  via Eq. (199).

Now we define *G* and *R* to mirror the structure of  $H_1$  in Eq. (213), so that  $G(\mathbf{u}) = \mathbf{G}^* \mathbf{u}^{\dagger}$ , and  $R(\mathbf{u}) = \mathbf{R}^* \mathbf{u}^{\dagger}$ where both are functions of  $\mathbf{u}$  rather than  $\mathbf{q}$ . Notice that to order  $\varepsilon^1$ ,  $H_1(\mathbf{u}) = \mathbf{H}_1^* \mathbf{u}^{\dagger}$  where  $\mathbf{u}^{\dagger}$  exactly mirrors  $\mathbf{q}^{\dagger}$ but with  $q_p = u_p$  and  $q_m = u_m$ . This means that the homological equation Eq. (202)) can be written as

$$\{H_0, \mathbf{G}^* \mathbf{u}^\dagger\} = \mathbf{R}^* \mathbf{u}^\dagger - \mathbf{H}_1^* \mathbf{u}^\dagger \rightsquigarrow \{H_0, \mathbf{G}^* \mathbf{u}^\dagger\} = (\mathbf{R}^* - \mathbf{H}_1^*) \mathbf{u}^\dagger.$$
(215)

Next we consider the Poisson bracket  $\{H_0, \mathbf{G}^* \mathbf{u}^\dagger\}$  which, using the definition for the Poisson bracket with  $\mathbf{u} = [u_p, u_m]^T$ , gives

$$\{H_0, \mathbf{G}^* \mathbf{u}^{\dagger}\} = \frac{\partial H_0}{\partial u_p} \frac{\partial \mathbf{G}^* \mathbf{u}^{\dagger}}{\partial u_m} - \frac{\partial H_0}{\partial u_m} \frac{\partial \mathbf{G}^* \mathbf{u}^{\dagger}}{\partial u_p}.$$
(216)

This leads to

$$\{H_0, \mathbf{G}^* \mathbf{u}^\dagger\} = 2m\omega_n^2 \left( u_m \mathbf{G}^* \frac{\partial \mathbf{u}^\dagger}{\partial u_m} - u_p \mathbf{G}^* \frac{\partial \mathbf{u}^\dagger}{\partial u_p} \right) = 2m\omega_n^2 \left[ G_1(-4) \ G_2(-2) \ G_3(0) \ G_4(2) \ G_5(4) \right] \mathbf{u}^\dagger, \tag{217}$$

such that the homological equation becomes

$$[G_1(-4) G_2(-2) G_3(0) G_4(2) G_5(4)] \mathbf{u}^{\dagger} = \frac{1}{2m\omega_n^2} ([0 \ 0 \ R_3 \ 0 \ 0] - \frac{k_3}{4} [1 \ 4 \ 6 \ 4 \ 1]) \mathbf{u}^{\dagger},$$
(218)

<sup>&</sup>lt;sup>57</sup>However, most treatments are from a mathematical perspective, and the scaling  $m = \omega_n = 1$  is applied (either directly, or through some form of non-dimensionalisation) such that the transformation to new coordinates  $\hat{x} = [\hat{x}, \hat{p}]^T$ , is carried out such that  $\hat{x} = x + \mathbf{i}p$  and  $\hat{p} = x - \mathbf{i}p$  (see for example [195]). This choice of coordinates will lead to the desired form of the Hamiltonian as defined by Birkhoff [11].

where the  $R_i$  coefficients have been set to zero except where a zero appears in  $\{H_0, G\}$  on the left hand side of Eq. (218). So Eq. (218) can be satisfied by setting

$$R_3 = \frac{3k_3}{2}$$
, and  $\mathbf{G}^* = \frac{\alpha}{8\omega_n^2} \begin{bmatrix} \frac{1}{4} \ 2 \ 0 \ -2 \ -\frac{1}{4} \end{bmatrix}$  where  $\alpha = \frac{k_3}{m}$ , (219)

and  $G^\ast$  is multiplied by  $u^\dagger$  to give the generator function.

Next we consider finding an approximate solution for x using the Poisson bracket  $\{u, G\}$ 

$$\{\mathbf{u},G\} = \frac{\partial \mathbf{u}}{\partial u_p} \frac{\partial G}{\partial u_m} - \frac{\partial \mathbf{u}}{\partial u_m} \frac{\partial G}{\partial u_p}.$$
(220)

Because  $\mathbf{u} = [u_p, u_m]^T$  is a vector this gives

$$\{\mathbf{u}, \mathbf{G}^* \mathbf{u}^{\dagger}\} = \begin{bmatrix} 1\\0 \end{bmatrix} \mathbf{G}^* \frac{\partial \mathbf{u}^{\dagger}}{\partial u_m} - \begin{bmatrix} 0\\1 \end{bmatrix} \mathbf{G}^* \frac{\partial \mathbf{u}^{\dagger}}{\partial u_p} \equiv \frac{\alpha}{8\omega_n^2} \begin{bmatrix} 2 & 0 & -6 & -1\\-1 & -6 & 0 & 2 \end{bmatrix} \mathbf{u}^*,$$
(221)

where  $\mathbf{u}^*$  for this example is

$$\mathbf{u}^* = \begin{bmatrix} u_p^3 \\ u_p^2 u_m \\ u_p u_m^2 \\ u_m^3 \end{bmatrix}.$$
(222)

Then to obtain the approximate expression for x, we use Eq. (221) with Eq. (199) and Eq. (211) to obtain

$$x = q_p + q_m = u_p + u_m + \varepsilon \frac{\alpha}{8\omega_n^2} (u_p^3 + u_m^3 - 6(u_p^2 u_m + u_p u_m^2)) + \mathcal{O}(\varepsilon^2).$$
(223)

Now that the normal form has been obtained, we can substitute an assumed solution, which we take as the two base solutions (without frequency detuning or phase lag) given as

$$u_p = \frac{U}{2} e^{i\omega_n t}$$
 and  $u_m = \frac{U}{2} e^{-i\omega_n t}$ , (224)

which then leads to a final expression for approximate displacement

$$x = (U - \varepsilon \frac{3\alpha U^3}{16\omega_n^2})\cos(\omega_n t) + \varepsilon \frac{\alpha U^3}{32\omega_n^2}\cos(3\omega_n t) + \mathcal{O}(\varepsilon^2).$$
(225)

This can be compared to Eq. (143), which is the result from Example 12 (Section 4.2.2).

As noted in Example 11, one characteristic of the approximation in Eq. (225) is that each order  $\varepsilon$  will update each of the harmonic components in the series. So as more  $\varepsilon$  terms are added, the coefficients of each harmonic components are modified. This is in contrast to the real normal form expression from Example 12, Eq. (168), where the higher-order corrections are orthogonal to the fundamental part of the solution. However, because there are free functions (see for example [85]) in the G<sup>\*</sup> matrix, Eq. (225) can also be made to have orthogonal higher-order correction terms. In this case the coefficient  $G_3$  can be chosen to have any value with no change to the homological equation. In this example, if  $G_3 = 3$  is selected then the series solution becomes

$$x = U\cos(\omega_n t) + \varepsilon \frac{\alpha U^3}{32\omega_n^2} \cos(3\omega_n t) + \mathcal{O}(\varepsilon^2).$$
(226)

This is the same result as that obtained in Example 12, Eq. (168), when  $\alpha_2 = 0$  and  $\alpha_1 = \alpha$  are substituted.

Notice that although we have obtained a solution for x we have not yet defined the Kamiltonian or the equations or motion for this example. To do this we first use Eq. (201) from which the Kamiltonian can be written as

$$K(\mathbf{u}) = 2m\omega_n^2 u_p u_m + \varepsilon \frac{3k_3}{2} u_p^2 u_m^2 + \mathscr{O}(\varepsilon^2),$$
(227)

180 | doi:10.25518/2684-6500.84

 $\Box$ 

which is the normal form of the Hamiltonian up to order  $\varepsilon$ , and can be compared with previous results, for example in [195], which has the same structure, but different scaling, as part of the chosen coefficients and diagonalisation transform. Substituting Eq. (224) into Eq. (227) gives

$$K(\mathbf{u}) = k \frac{U^2}{2} + \varepsilon \frac{3k_3 U^4}{32} + \mathscr{O}(\varepsilon^2),$$
(228)

which implies that  $K(\mathbf{u})$  is independent of mass *m*, and is dependent on just the stiffness parameters *k* and  $k_3$  together with the amplitude value *U*. As such, it represents a nonlinear potential energy function for the Duffing oscillator.

One of the advantages of using the Hamiltonian normal form is that you can work with Hamiltonians rather than equations of motion. However, the transformations introduce a scaling factor, which is related to the nature of the diagonalising transformation. To see this we note that in the original coordinates,  $p = [x, p]^T$  the following standard relationship holds

$$\dot{\boldsymbol{p}} = \{\boldsymbol{p}, H\} \tag{229}$$

which when evaluated gives Eq. (209). We would now like to find the equivalent version of this for the transformed coordinates q and u. First consider the coordinate change from  $p \rightarrow q$ , for which Eq. (229), via the chain rule, becomes

$$\frac{\partial x}{\partial q_p} \dot{q}_p + \frac{\partial x}{\partial q_m} \dot{q}_m = \frac{\partial H}{\partial q_p} \frac{\partial q_p}{\partial p} + \frac{\partial H}{\partial q_m} \frac{\partial q_m}{\partial p},$$

$$\frac{\partial p}{\partial q_p} \dot{q}_p + \frac{\partial p}{\partial q_m} \dot{q}_m = -\left(\frac{\partial H}{\partial q_p} \frac{\partial q_p}{\partial x} + \frac{\partial H}{\partial q_m} \frac{\partial q_m}{\partial x}\right).$$
(230)

To compute the partial derivatives on the left of Eq. (230) we use Eq. (211) to find that

$$\frac{\partial x}{\partial q_p} = \frac{\partial x}{\partial q_m} = 1, \qquad \frac{\partial p}{\partial q_p} = m\lambda_1, \qquad \frac{\partial p}{\partial q_m} = m\lambda_2.$$
(231)

To find the partial derivatives on the right of Eq. (230) it is useful to explicitly define the inverse relationship of the transformation using Eq. (211) giving

$$\begin{bmatrix} q_p \\ q_m \end{bmatrix} = \frac{1}{m(\lambda_2 - \lambda_1)} \begin{bmatrix} m\lambda_2 & -1 \\ & & \\ -m\lambda_1 & 1 \end{bmatrix} \begin{bmatrix} x \\ p \end{bmatrix}.$$
(232)

From this we obtain

$$\frac{\partial q_p}{\partial p} = \frac{-1}{m(\lambda_2 - \lambda_1)}, \qquad \frac{\partial q_m}{\partial p} = \frac{1}{m(\lambda_2 - \lambda_1)}, \qquad \frac{\partial q_p}{\partial x} = \frac{1}{2}, \qquad \frac{\partial q_m}{\partial x} = \frac{1}{2}.$$
(233)

Now, using these definitions, we can rewrite Eq. (230) as

\_

$$\dot{q}_{p} + \dot{q}_{m} = \frac{\partial H}{\partial q_{p}} \left( \frac{-1}{m(\lambda_{2} - \lambda_{1})} \right) + \frac{\partial H}{\partial q_{m}} \left( \frac{1}{m(\lambda_{2} - \lambda_{1})} \right),$$

$$m\lambda_{1}\dot{q}_{p} + m\lambda_{2}\dot{q}_{m} = -\frac{1}{2} \left( \frac{\partial H}{\partial q_{p}} + \frac{\partial H}{\partial q_{m}} \right),$$
(234)

which can be solved as a pair of simultaneous equations. For example, multiplying the first line in Eq. (234) by  $-m\lambda_1$  and adding the two equations results in

$$\dot{q}_m m(\lambda_2 - \lambda_1) = \frac{\partial H}{\partial q_p} \left( -\frac{1}{2} \right) + \frac{\partial H}{\partial q_m} \left( \frac{1}{2} \right) - \frac{1}{2} \left( \frac{\partial H}{\partial q_p} + \frac{\partial H}{\partial q_m} \right), \rightsquigarrow$$

$$\dot{q}_m = -\frac{1}{m(\lambda_2 - \lambda_1)} \frac{\partial H}{\partial q_p} = \frac{1}{2m \mathbf{i} \omega_n} \frac{\partial H}{\partial q_p} = -\mathbf{i} \omega_n q_m + \varepsilon \frac{3k_3}{2m \mathbf{i} \omega_n} (q_p + q_m)^3,$$
(235)

where  $\frac{\partial H}{\partial q_p}$  has been obtained from Eq. (212). Similarly, multiplying the second line in Eq. (234) by  $-m\lambda_2$  and adding the two equations results in

$$\dot{q}_p = \frac{1}{m(\lambda_2 - \lambda_1)} \frac{\partial H}{\partial q_m} = \mathbf{i}\omega_n q_p - \varepsilon \frac{3k_3}{2m\mathbf{i}\omega_n} (q_p + q_m)^3,$$
(236)

where  $\frac{\partial H}{\partial q_m}$  has been obtained from Eq. (212). Note that if we define  $\hat{\rho} = 1/(m(\lambda_2 - \lambda_1))$ , then we have the modified Hamiltonian relationships<sup>58</sup>

$$\dot{q}_p = \frac{1}{\hat{\rho}} \frac{\partial H}{\partial q_m}, \quad \dot{q}_m = -\frac{1}{\hat{\rho}} \frac{\partial H}{\partial q_p}, \quad \rightsquigarrow \quad \dot{\mathbf{q}} = \frac{1}{\hat{\rho}} \{\mathbf{q}, H\}.$$
(237)

Now the transformation from  $\mathbf{q} \rightarrow \mathbf{u}$  leads to the relationship

$$\dot{u}_{p} = \frac{1}{\hat{\rho}} \frac{\partial K}{\partial u_{m}} = \mathbf{i}\omega_{n}u_{p} + \mathbf{i}\varepsilon \frac{3\alpha}{2\omega_{n}}u_{p}^{2}u_{m},$$

$$\dot{u}_{m} = -\frac{1}{\hat{\rho}} \frac{\partial K}{\partial u_{p}} = -\mathbf{i}\omega_{n}u_{m} - \mathbf{i}\varepsilon \frac{3\alpha}{2\mathbf{i}\omega_{n}}u_{p}u_{m}^{2},$$
(238)

where,  $\alpha = k_3/m$ , and so  $\dot{\mathbf{u}} = \frac{1}{\hat{\rho}} \{\mathbf{u}, K\}$ , which is exactly the same result we obtained using the complex normal form in Example 11, as given in Eq. (141). Note that the scaling factor,  $\hat{\rho}$  relates to the term multiplying the inverse matrix in Eq. (232).

#### Damping in normal form transformations 4.5

For the linear systems in Section 2 and Section 3 dealing with damped oscillators is relatively straight forward (assuming viscous damping) and leads to well established results. However, in nonlinear normal form analysis, the inclusion of damping is more complicated. We motivate this discussion, with the following example.

#### 4.5.1 Example 14: the damped, unforced escape equation

The damped, unforced escape equation can be written as

$$\ddot{x} + 2\zeta \omega_n \dot{x} + \omega_n^2 x + \beta x^2 = 0, \quad \rightsquigarrow \quad \dot{x} = A \mathbf{x} + \tilde{\mathcal{N}}(\mathbf{x}): \quad A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & -2\zeta \omega_n \end{bmatrix}, \quad \tilde{\mathcal{N}}(\mathbf{x}) = \begin{cases} 0 \\ -\beta x_1^2 \end{cases}, \quad (239)$$

where the states are  $\mathbf{x} = \{x_1 \ x_2\}^T = \{x \ \dot{x}\}^T$ . Transforming to new coordinates  $\mathbf{q} = \{q_p, q_m\}^T$  using the linear normal modes of the system,  $\mathbf{x} = \Phi \mathbf{q}$  (see Section 3.2) gives modal matrices

$$\Lambda = \begin{bmatrix} \lambda_1 & 0 \\ 0 & \lambda_2 \end{bmatrix}, \quad \Phi = \begin{bmatrix} 1 & 1 \\ \lambda_1 & \lambda_2 \end{bmatrix}, \quad \Phi^{-1} = \frac{1}{\lambda_2 - \lambda_1} \begin{bmatrix} \lambda_2 & -1 \\ -\lambda_1 & 1 \end{bmatrix},$$

where  $\lambda_1$  and  $\lambda_2$  are the eigenvalues of *A*, given by  $\lambda_{1,2} = -\zeta \omega_n \pm i\omega_d$ . The transformed equation of motion is given by

$$\dot{\boldsymbol{q}} = \Lambda \boldsymbol{q} + \varepsilon \boldsymbol{n}_1(\boldsymbol{q}) \quad \text{where} \quad \boldsymbol{n}_1(\boldsymbol{q}) = \Phi^{-1} \tilde{\mathscr{N}}(\Phi \boldsymbol{q}) = \frac{\beta}{\lambda_2 - \lambda_1} \begin{bmatrix} (q_p + q_m)^2 \\ -(q_p + q_m)^2 \end{bmatrix} \quad \text{and} \quad \boldsymbol{n}_2 = 0, \dots$$
(240)

We know from Example 10 (Section 4.1.1) that quadratic terms can be eliminated at order k = 2 or  $\varepsilon^1$  and so we assume that a nonlinear coordinate transformation from coordinate q to a new coordinate set u, is possible obtain a linear system of the form

$$\dot{\boldsymbol{u}} = \Lambda \boldsymbol{u} + \mathscr{O}(\varepsilon^2) \qquad \text{or written in full} \qquad \begin{bmatrix} \dot{\boldsymbol{u}}_p \\ \dot{\boldsymbol{u}}_m \end{bmatrix} = \begin{bmatrix} \lambda_1 & 0 \\ 0 & \lambda_2 \end{bmatrix} \begin{bmatrix} \boldsymbol{u}_p \\ \boldsymbol{u}_m \end{bmatrix} + \mathscr{O}(\varepsilon^2), \tag{241}$$

<sup>&</sup>lt;sup>58</sup>We note that [6] also comments on modified Hamiltonian relationships for the case where  $\hat{\rho} = i$ .

which means completely eliminating the nonlinear term  $n_1(q)$  in Eq. (240)<sup>59</sup>. To do this we make the transformation

$$\boldsymbol{q} = \boldsymbol{u} + \varepsilon \boldsymbol{h}(\boldsymbol{u}) = \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \varepsilon \begin{bmatrix} h_1(u_p, u_m) \\ h_2(u_p, u_m) \end{bmatrix} = \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \varepsilon \begin{bmatrix} b_1 u_p^2 + b_2 u_p u_m + b_3 u_m^2 \\ b_4 u_p^2 + b_5 u_p u_m + b_6 u_m^2 \end{bmatrix} + \mathscr{O}(\varepsilon^2),$$
(242)

where the dots are used to denote that the  $h_j(u_p, u_m)$  series can have higher order terms, and we have included up to quadratic terms because the nonlinear term in Eq. (239) is quadratic. The  $b_j$  are coefficients that are yet to be identified. Substituting the first of Eq. (242) and Eq. (241) into Eq. (240) leads to order- $\varepsilon^1$  homological equation of the form

$$\varepsilon^{1}: \frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{h} - \Lambda \boldsymbol{h} = \boldsymbol{n}(\boldsymbol{u}) \quad \rightsquigarrow \quad \frac{\partial \boldsymbol{h}}{\partial \boldsymbol{u}} \Lambda \boldsymbol{u} - \Lambda \boldsymbol{h} = \boldsymbol{n}(\boldsymbol{u}) \quad \rightsquigarrow \quad \boldsymbol{b} \Lambda^{*} \mathbf{u}^{*} - \Lambda \boldsymbol{b} \mathbf{u}^{*} = \boldsymbol{n}^{*} \mathbf{u}^{*}.$$
(243)

In order to compute the Lie derivative we use

$$\frac{\partial \boldsymbol{h}}{\partial \boldsymbol{u}} \Lambda \boldsymbol{u} = \begin{bmatrix} \frac{\partial h_1}{\partial u_p} & \frac{\partial h_1}{\partial u_m} \\ \frac{\partial h_2}{\partial u_p} & \frac{\partial h_2}{\partial u_m} \end{bmatrix} \begin{bmatrix} \lambda_1 & 0 \\ 0 & \lambda_2 \end{bmatrix} \begin{bmatrix} u_p \\ u_m \end{bmatrix}.$$
(244)

As a result the order- $\varepsilon^1$  homological equation Eq. (243) can be written as

$$\begin{bmatrix} b_1(-\lambda_1) & b_2(-\lambda_2) & b_3(\lambda_1 - 2\lambda_2) \\ b_4(\lambda_2 - 2\lambda_1) & b_5(-\lambda_1) & -b_6(\lambda_2) \end{bmatrix} \mathbf{u}^* = -\frac{\beta}{\lambda_2 - \lambda_1} \begin{bmatrix} 1 & 2 & 1 \\ -1 & -2 & -1 \end{bmatrix} \mathbf{u}^*,$$
(245)

where  $\mathbf{u}^* = [u_p^2 \ u_p u_m \ u_m^2]^2$ . Eq. (245) can be solved for the  $b_j$  coefficients and substituted into Eq. (242) to give a transformation of

$$\boldsymbol{q} = \boldsymbol{u} + \boldsymbol{h}(\boldsymbol{u}) = \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \boldsymbol{\varepsilon} \begin{bmatrix} \frac{\beta}{(\lambda_2 - \lambda_1)\lambda_1} u_p^2 + \frac{2\beta}{(\lambda_2 - \lambda_1)\lambda_2} u_p u_m - \frac{\beta}{(\lambda_2 - \lambda_1)(\lambda_1 - 2\lambda_2)} u_m^2 \\ \frac{\beta}{(\lambda_2 - \lambda_1)(\lambda_2 - 2\lambda_1)} u_p^2 - \frac{2\beta}{(\lambda_2 - \lambda_1)\lambda_1} u_p u_m - \frac{\beta}{(\lambda_2 - \lambda_1)\lambda_2} u_m^2 \end{bmatrix} + \mathcal{O}(\boldsymbol{\varepsilon}^2)$$
(246)

up to quadratic terms.

This analysis relies on the denominator terms such as  $(\lambda_2 - 2\lambda_1)$  that occur in Eq. (245) and Eq. (246) not becoming zero (otherwise the system would have resonances). In this case, the eigenvalues are complex, and therefore they are in the Poincaré domain (see Fig. 4), meaning non-resonant. However, what happens as the damping becomes small and  $\zeta \rightarrow 0$  so that  $\lambda_{1,2} \rightarrow \pm i\omega_n$ ? In that case, we might want to consider the case when the eigenvalues are in the Siegel domain, and therefore the homological equation detects the resonant terms (see Section 4.1.2 for a more detailed discussion of resonance in this context). There are three methods that can do this, and these are:

**Method 1:** A solution to this problem proposed by Jezequel & Lamarque [80] was to reformulate the eigenvalues using the idea of a *detuning* parameter (using a *k*-order methodology). So for Example 14 where the eigenvalues are  $\lambda_{1,2} = -\zeta \omega_n \pm i \omega_d$  we write

$$\lambda_1 = \mathbf{i}\omega_n + \delta$$

$$\lambda_2 = -\mathbf{i}\omega_n + \bar{\delta}$$
(247)

where  $\delta$  is the detuning parameter such that,  $\operatorname{Re}\delta = -\zeta \omega_n$ ,  $\operatorname{Im}\delta = \omega_n(1 - \sqrt{1 - \zeta^2})$  and  $|\delta| \ll 1$  meaning that  $\zeta \ll 1$ . Here  $\operatorname{Im}\delta$  is used to represent the detuning (meaning small difference) between  $\omega_n$  and  $\omega_d$ .

Applying this approach to Example 14, Eq. (244) becomes

$$\frac{\partial \mathbf{h}}{\partial \boldsymbol{u}} \Lambda \boldsymbol{u} = \begin{bmatrix} \frac{\partial h_1}{\partial u_p} & \frac{\partial h_1}{\partial u_m} \\ \frac{\partial h_2}{\partial u_p} & \frac{\partial h_2}{\partial u_m} \end{bmatrix} \begin{bmatrix} \mathbf{i}\omega_n & 0 \\ 0 & -\mathbf{i}\omega_n \end{bmatrix} \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \text{ terms in } \delta,$$

$$\text{and} \quad \frac{1}{\lambda_2 - \lambda_1} = \frac{1}{-2\mathbf{i}\omega_n} + \text{ terms in } \delta$$
(248)

<sup>59</sup>Note that because  $n_2 = 0, ...$  etc. we use  $n_1 = n$ , and correspondingly  $b_1 = b$  and  $\hat{n}_{u1} = \hat{n}_u$ .

where the terms in  $\delta$  are assumed to be relevant only at the next highest order than the one currently being computed (e.g in this example we are computing to quadratic, so  $\delta$  terms are assumed to be included at cubic order). Now substituting these new expressions into Eq. (246) (with  $\varepsilon = 1$ ) gives

$$\mathbf{q} = \mathbf{u} + \mathbf{h}(\mathbf{u}) = \begin{bmatrix} u_p \\ u_m \end{bmatrix} + \frac{\beta}{\omega_n^2} \begin{bmatrix} \frac{u_p^2}{2} & -u_p u_m & -\frac{u_m^2}{6} \\ & & \\ -\frac{u_p^2}{6} & -u_p u_m & \frac{u_m^2}{2} \end{bmatrix},$$
(249)

which once the base solutions are substituted, and using  $x = x_1 = q_p + q_m$ , gives the same solution for *x* as Eq. (105) (the undamped normal form). The normal form equations, Eq. (241), become

$$\dot{u}_p = +\mathbf{i}\omega_n u_p + \delta u_p,\tag{250}$$

$$\dot{u}_m = -\mathbf{i}\omega_n u_m + ar{\delta} u_m,$$

which is a linear damped system. We note at this order, damping appears in Eq. (250), but not in the series solution for  $x_1$  (e.g. Eq. (105)) because the damping terms have been shifted up to the next order. To see the effect of damping in the solution for  $x_1$  it would be necessary to compute the next order of the normal form.

The concept of detuning is a very important one for nonlinear normal forms and we will return to it in Section 4.6. However, to continue the discussion on including damping in normal form transformations of nonlinear systems we mention two further methods.

**Method 2:** This is to use the method of reduction of order presented in Section 2.3. Let's consider how this might work for the damped escape equation from Example 14 (Section 4.5.1). The damped, unforced escape equation (e.g. Eq. (239)) can be expressed as

$$\ddot{x} + \varepsilon \dot{x} + \omega_n^2 x + \varepsilon \hat{\beta} x^2 = 0, \tag{251}$$

where we have rescaled the  $\beta$  coefficient such that  $\beta = \varepsilon \hat{\beta}$  and  $\varepsilon = 2\zeta \omega_n$ . This rescaling means that we are assuming that both the damping and nonlinear coefficient are of the order  $\varepsilon^1$  small, which is a reasonable assumption for many weakly nonlinear oscillators, but will restrict the parameter values for which this method can be applied. Next make the reduction of order substitution  $x = q(t)e^{-\frac{\varepsilon}{2}t}$  so that the escape equation becomes

$$\ddot{q} + \omega_d^2 x + \varepsilon \mathrm{e}^{-\frac{\varepsilon}{2}t} \hat{\beta} q^2 = 0.$$
<sup>(252)</sup>

Now, assuming  $\varepsilon$  is a small quantity we can apply an approximation of the exponential function  $e^{-\frac{\varepsilon}{2}t} = 1 - \frac{\varepsilon}{2}t + ... \mathscr{O}(\varepsilon^2 t^2)$  we obtain

$$\ddot{q} + \omega_d^2 x + \varepsilon (1 - \frac{\varepsilon}{2}t)\hat{\beta}q^2 = 0 \rightsquigarrow \ddot{q} + \omega_d^2 x + \varepsilon \hat{\beta}q^2 + \dots \mathcal{O}(\varepsilon^2) = 0,$$
(253)

which is valid when  $\varepsilon$  is small, and is equivalent to the case when damping is small,  $\zeta \ll 1$ . Now Eq. (253) has the same structure as the *undamped* escape equation, except now the damping is included in the damped natural frequency,  $\omega_d$ . Applying the normal form transformation to Eq. (253) gives a normal form equation (e.g. Eq. (83) with  $n_u^{(2)} = n_u^{(3)} = 0$  and  $\omega_d$  replacing  $\omega_n$ ) of

$$\ddot{u} + \omega_d^2 u = 0. \tag{254}$$

and a solution for x (e.g. Eq. (105) with  $\omega_d$  replacing  $\omega_n$ ) of

$$x = U\cos(\omega_d t) + \frac{\beta U^2}{6\omega_d^2}\cos(2\omega_d t) - \frac{\beta U^2}{2\omega_d^2}.$$
(255)

Notice that now the solution for *x* in Eq. (255) is in terms of  $\omega_d$ , but there is no exponential decay, and just like the Jezequel & Lamarque [80] solution, higher order terms need to be computed to include these effects, which in this case are the  $\mathscr{O}(\varepsilon^2)$  terms<sup>60</sup>. In fact computing the higher order terms in both the above methods is non-trivial, and so we present a third method which is used in preference to both previous methods.

<sup>&</sup>lt;sup>60</sup>The interested reader might want to study the related method proposed by Burton [17] which avoids the expansion of the exponential term.

**Method 3:** In this method we treat the damping terms as part of the nonlinear vector such that the unforced version of Eq. (79) becomes

$$M\ddot{\mathbf{x}} + K\mathbf{x} + \mathcal{N}_d(\mathbf{x}, \dot{\mathbf{x}}) = 0, \tag{256}$$

where the  $\mathcal{N}_d$  vector now also contains the damping terms. The particular logic for doing this for systems of equations written in second-order form, has been described in Section 4.3.2, and for such systems this is the most straightforward method to implement. Further details can be found in [183]. An example will be shown in Section 4.6.1

## 4.6 Frequency detuning

We now consider an important modelling technique that will allow us to distinguish between a response frequency,  $\omega_r$ , that is different from the linear damped or undamped natural frequencies,  $\omega_d$  and  $\omega_n$  respectively. This will enable us to develop the idea of amplitude-dependent *backbone* curves in Section 4.8, although it should be noted that backbone curves (and other types of parameter variation) can be obtained using many other techniques — see for example [163, 112, 148, 149, 150, 90, 50, 53, 119, 173, 101, 119]. We consider the multi-frequency case (i.e. more than a single-degree-of-freedom) where

$$\omega_{ni}^2 = \omega_{ri}^2 + \varepsilon \delta_i + \mathscr{O}(\varepsilon^2) \quad \text{for} \quad i = 1, 2, 3...N.$$
(257)

Then we can form a matrix equation

$$\hat{\Lambda} = \Upsilon + \varepsilon \Delta + \mathscr{O}(\varepsilon^{2}) \quad \text{where} \quad \Upsilon = \begin{bmatrix} \omega_{r1}^{2} & 0 & \dots & 0 & 0 \\ 0 & \omega_{r2}^{2} & \dots & 0 & 0 \\ \vdots & \vdots & \vdots & \ddots & 0 \\ 0 & 0 & \dots & 0 & \omega_{rN}^{2} \end{bmatrix},$$
(258)

the matrix  $\hat{\Lambda}$  is given by Eq. (70) and  $\Delta$  is a diagonal matrix of the  $\delta_i$  values.

Now consider the case when the homological equation is derived up to second order (Eq. (154) and Eq. (155) combined), which gives

$$-\varepsilon \hat{\mathbf{n}}_{\mathbf{u}\mathbf{1}}(\mathbf{u}) - \varepsilon^2 \hat{\mathbf{n}}_{\mathbf{u}\mathbf{2}}(\mathbf{u}) + \varepsilon \ddot{\mathbf{h}}_1 + \varepsilon^2 \ddot{\mathbf{h}}_2 = -\hat{\Lambda}(\varepsilon \mathbf{h}_1 + \varepsilon^2 \mathbf{h}_2) - \varepsilon \hat{\mathbf{n}}_1(\mathbf{u}) - \varepsilon^2 \frac{\partial \hat{\mathbf{n}}_1(\mathbf{u})}{\partial \mathbf{u}} \mathbf{h}_1 - \varepsilon^2 \hat{\mathbf{n}}_2(\mathbf{u}),$$
(259)

where  $\frac{\partial \hat{\mathbf{n}}}{\partial \mathbf{u}}$  is the Jacobian matrix (or gradient vector) depending on the dimensions of  $\mathbf{u}$ , and an overdot represents  $\frac{d}{dt}$ . Now we introduce detuning using Eq. (258) to give

$$-\varepsilon \hat{\mathbf{n}}_{u1}(\mathbf{u}) - \varepsilon^2 \hat{\mathbf{n}}_{u2}(\mathbf{u}) + \varepsilon \ddot{\mathbf{h}}_1 + \varepsilon^2 \ddot{\mathbf{h}}_2 = (\Upsilon + \varepsilon \Delta)(\varepsilon \mathbf{h}_1 + \varepsilon^2 \mathbf{h}_2) - \varepsilon \hat{\mathbf{n}}_1(\mathbf{u}) - \varepsilon^2 \frac{\partial \hat{\mathbf{n}}_1(\mathbf{u})}{\partial \mathbf{u}} \mathbf{h}_1 - \varepsilon^2 \hat{\mathbf{n}}_2(\mathbf{u}).$$
(260)

Now it can be seen from Eq. (260) that the terms at orders  $\varepsilon^1$  and  $\varepsilon^2$  can be equated separately such that

$$\varepsilon^{1}: \quad \mathbf{\hat{n}}_{1} - \mathbf{\hat{n}}_{u1} = -\mathbf{\ddot{h}}_{1} - \mathbf{\Upsilon}\mathbf{h}_{1} = -\mathbf{b}_{1} (\mathbf{\Lambda}^{*}_{d})^{2} \mathbf{u}^{*} - \mathbf{\Upsilon}\mathbf{b}_{1} \mathbf{u}^{*},$$
  

$$\varepsilon^{2}: \quad \Delta \mathbf{h}_{1} + \frac{\partial \mathbf{\hat{n}}_{1}}{\partial \mathbf{u}} \mathbf{h}_{1} + \mathbf{\hat{n}}_{2} - \mathbf{\hat{n}}_{u2} = -\mathbf{\ddot{h}}_{2} - \mathbf{\Upsilon}\mathbf{h}_{2} = -\mathbf{b}_{2}\mathbf{\widetilde{\Upsilon}}^{2}\mathbf{u}^{+} - \mathbf{\Upsilon}\mathbf{b}_{2}\mathbf{u}^{+}.$$
(261)

The first line in Eq. (261) is the detuned order- $\varepsilon^1$  homological equation (compare with in Eq. (156)), where  $\Lambda^*_d$  represents the Lie derivative coefficient matrix obtained from Eq. (127) (and Eq. (153)) when the detuned system is being considered. This type of detuned formulation will be used for the rest of the examples in this paper, starting with Examples 15 and 16 (Section 4.6.1 and Section 4.6.2 respectively).

The second line in Eq. (261) gives the detuned order  $\varepsilon^2$  homological equation which needs to be solved if the normal form is to be computed to order  $\varepsilon^2$  or higher. In this case the basis function is  $\mathbf{u}^+$  such that  $\mathbf{h}_2 = \mathbf{b}_2 \mathbf{u}^+$ . Notice also that evaluating the  $\mathbf{\ddot{h}}_2$  term leads to a  $\tilde{\Upsilon}^2$  term. This term is equivalent to  $(\Lambda^*_d)^2$ , and is computed as the Lie derivative coefficient matrix obtained from Eq. (127) (and Eq. (153)) when the detuned system is being considered. This will be demonstrated in Example 18, Section 4.8.1. Now we consider two detuned examples computed to order  $\varepsilon^1$ .

#### 4.6.1 Example 15 the unforced Van der Pol oscillator

The unforced Van der Pol oscillator can be written as

$$\ddot{x} + \varepsilon(\mu x^2 - \gamma)\dot{x} + \omega_n^2 x = 0, \text{ with } x = q \rightsquigarrow \quad \ddot{q} + \omega_n^2 q + \varepsilon(\mu q^2 - \gamma)\dot{q} = 0, \quad \rightsquigarrow \quad \ddot{q} + \omega_n^2 q + \varepsilon \hat{\mathbf{n}}_1(q_p, q_m) = 0, \quad (262)$$

where by using  $q = q_p + q_m$  and  $\dot{q} = \mathbf{i}\omega_r(q_p - q_m)$  we can obtain

$$\hat{\mathbf{n}}_1(q_p, q_m) = \mathbf{i}\omega_r(\mu q_p^3 + \mu q_p^2 q_m - \mu q_p q_m^2 - \mu q_m^3 - \gamma q_p + \gamma q_m) = \mathbf{i}\omega_r[\mu \ \mu - \mu - \mu - \gamma \ \gamma]\mathbf{q}^*,$$
(263)

where  $\mathbf{q}^* = [q_p^3 \ q_p^2 q_m \ q_p q_m^2 \ q_m^3 \ q_p \ q_m]^T$  and  $\mathbf{n}_2 = 0^{61}$ . Notice that, as described in Section 4.5.1, we are including the linear damping term in the nonlinear vector  $\hat{\mathbf{n}}(q_p, q_m)$ .

Now apply a near-identity nonlinear transform, Eq. (116), of the form

$$q = u + \varepsilon \mathbf{b} \mathbf{u}^* \qquad \text{where} \qquad \mathbf{b} = [b_1 \ b_2 \ b_3 \ b_4 \ b_5 \ b_6], \tag{264}$$

and  $\mathbf{u}^* = [u_p^3 \ u_p^2 u_m \ u_p u_m^2 \ u_m^3 \ u_p \ u_m]^T$ . The transformed dynamic equation, Eq. (146), is expressed in the new coordinates as

$$\ddot{u} + \omega_n^2 u + \varepsilon \hat{\mathbf{n}}_u(u_p, u_m) = 0.$$
(265)

Substituting Eq. (264) and Eq. (265) into Eq. (262) (with detuning) leads to the  $\varepsilon^1$  homological equation  $\hat{\mathscr{H}}_1(\mathbf{u})$  given by

$$\varepsilon^{1}: -\mathbf{b}\frac{\mathrm{d}^{2}\mathbf{u}^{*}}{\mathrm{d}t^{2}} - \omega_{r}^{2}\mathbf{b}\mathbf{u}^{*} = \hat{\mathbf{n}} - \hat{\mathbf{n}}_{\mathbf{u}} = -\mathbf{b}\Lambda_{d}^{*2}\mathbf{u}^{*} - \Upsilon\mathbf{b}\mathbf{u}^{*}.$$
(266)

Then using  $\hat{\mathbf{n}}_{\mathbf{u}} = [\hat{n}_{u1} \ \hat{n}_{u2} \ \hat{n}_{u3} \ \hat{n}_{u4} \ \hat{n}_{u5} \ \hat{n}_{u6}]\mathbf{u}^*$ , the second Lie derivative Eq. (153), with  $q_i \rightarrow u_i$  (to order  $\varepsilon^0$ ) gives

Evaluating the right hand side gives four zero values which allows the  $\hat{\mathbf{n}}_{\mathbf{u}}$  coefficients to be identified such that

$$(\mathbf{i}\omega_r[\mu \ \mu - \mu - \mu - \gamma \ \gamma] - \mathbf{i}\omega_r[0 \ \mu \ -\mu \ 0 \ -\gamma \ \gamma])\mathbf{u}^* = [b_1(8\omega_r^2) \ b_2(0) \ b_3(0) \ b_4(8\omega_n^2) \ b_5(0) \ b_6(0)]\mathbf{u}^*.$$
(268)

From this, the transform coefficient values are  $b_1 = i\mu/8\omega_r$ ,  $b_4 = -i\mu/8\omega_r$ , and  $b_2 = b_3 = b_5 = b_6 = 0$ . Note that  $b_2$ ,  $b_3$ ,  $b_5$  and  $b_6$  are the free functions in this example, as any value can be selected without affecting the outcome of the homological equation. Other choices will be discussed in Section 4.7. Next, using the fact that

$$\boldsymbol{\varepsilon}\hat{\mathbf{n}}_{\mathbf{u}}(u_p, u_m) = \boldsymbol{\varepsilon}\mathbf{i}\omega_r[0 \ \mu \ -\mu \ 0 \ -\gamma \ \gamma]\mathbf{u}^* = \boldsymbol{\varepsilon}\mathbf{i}\omega_r(\mu u_p^2 u_m - \mu u_p u_m^2 - \gamma u_p + \gamma u_m) = \boldsymbol{\varepsilon}\mathbf{i}\omega_r(\mu u_p u_m - \gamma)(u_p - u_m),$$

and noting that  $\mathbf{i}\omega_r(u_p - u_m) = \dot{u}$  gives the  $\varepsilon^1$  real normal form as

$$\ddot{u} + \omega_n^2 u + \varepsilon (\mu u_p u_m - \gamma) \dot{u} = 0, \quad \rightsquigarrow \quad \ddot{u} + \omega_n^2 u + \varepsilon (\mu \frac{U^2}{4} - \gamma) \dot{u} = 0,$$
(269)

when the base solutions are used in the  $u_p u_m$  term. In fact, substituting the base solutions into Eq. (269) gives

$$(\mathbf{i}\omega_r)^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \omega_n^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \mathbf{i}\omega_r \varepsilon (\mu \frac{U^2}{4} - \gamma) \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} - \mathbf{e}^{-\mathbf{i}\omega_r t}) = 0.$$
(270)

<sup>&</sup>lt;sup>61</sup>As  $\mathbf{n}_2 = 0, ...$  etc. we set  $\mathbf{n}_1 = \mathbf{n}$ , and correspondingly,  $\mathbf{h}_1 = \mathbf{h}$ ,  $\mathbf{b}_1 = \mathbf{b}$  and  $\mathbf{\hat{n}_{u1}} = \mathbf{\hat{n}_u}$  throughout this example.



**Fig. 6:** Two simulations of the undamped Van der Pol oscillator showing the solution, *x*, from Example 15 (real normal form shown with red crosses). The normal form solution is compared with a reference solution generated from a fourth-order Runge-Kutta integration (the 'odeint' numerical integration routine from the Python numerical library — blue solid line). The real normal form solution is computed with Eq. (273) (red crosses). Parameter values are: x(0) = 0.025,  $\gamma = 0.2$ , and  $\omega_n = 1$ . In panel (a) the nonlinearity is relatively small  $\mu = 1$ , so that U = 0.89, and in panel (b) the nonlinearity is five times larger,  $\mu = 5$ , so that U = 0.4. The initial condition for the odeint simulation has been deliberately set to be less than the steady state amplitude of the limit cycle (i.e. periodic solution), in order to demonstrate how the transient solution converges to the limit cycle as time increases.

The real and imaginary parts of the  $e^{i\omega_r t}$  terms (or the  $e^{-i\omega_r t}$ ) can be balanced to give

Re: 
$$\omega_r^2 = \omega_n^2$$
,  $\rightsquigarrow \quad \omega_r = \omega_n$ ,  
Im:  $\omega_r \varepsilon (\mu \frac{U^2}{4} - \gamma) = 0$ ,  $\rightsquigarrow \quad U = \sqrt{4\frac{\gamma}{\mu}}$ ,
(271)

which corresponds to the classical solution to the unforced Van der Pol oscillator with a single limit cycle solution. Using  $\omega_r = \omega_n$ , the near-identity transform, to order  $\varepsilon^1$  may now be written as

$$x = u + \varepsilon \mathbf{b} \mathbf{u}^* = \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_n t} + \mathbf{e}^{-\mathbf{i}\omega_n t}) + \varepsilon \left[ \frac{\mathbf{i}\mu}{8\omega_n} \ 0 \ 0 \ \frac{-\mathbf{i}\mu}{8\omega_n} \ 0 \ 0 \right] \mathbf{u}^*, \tag{272}$$

which gives the solution for x as

$$x = \sqrt{4\frac{\gamma}{\mu}}\cos(\omega_n t) - \varepsilon \frac{\mu}{32\omega_n} \left(\sqrt{4\frac{\gamma}{\mu}}\right)^3 \sin(3\omega_n t) + \mathcal{O}(\varepsilon^2). \qquad \Box$$
(273)

A numerically computed example is shown in Fig. 6.

#### 4.6.2 Example 16 the unforced Rayleigh oscillator

The unforced Rayleigh oscillator can be written as<sup>62</sup>

$$\ddot{x} + \varepsilon (\frac{\dot{x}^2}{3} - 1)\dot{x} + \omega_n^2 x = 0, \text{ with } x = q \rightsquigarrow \qquad \ddot{q} + \omega_n^2 q + \varepsilon (\frac{\dot{q}^2}{3} - 1)\dot{q} = 0, \quad \rightsquigarrow \quad \ddot{q} + \omega_n^2 q + \varepsilon \hat{\mathbf{n}}(q_p, q_m) = 0,$$
(274)

where by using  $q = q_p + q_m$  and  $\dot{q} = \mathbf{i}\omega_r(q_p - q_m)$  we can obtain

$$\hat{\mathbf{n}}(q_p, q_m) = -\mathbf{i}\omega_r(\frac{\omega_r^2}{3}(q_p^3 - 3q_p^2 q_m + 3q_p q_m^2 - q_m^3) + q_p - q_m) = \mathbf{i}\omega_r[-\frac{\omega_r^2}{3} \quad \omega_r^2 \quad -\omega_r^2 \quad \frac{\omega_r^2}{3} \quad -1 \quad 1]\mathbf{q}^*,$$
(275)

where  $\mathbf{q}^* = [q_p^3 \ q_p^2 q_m \ q_p q_m^2 \ q_m^3 \ q_p \ q_m]^T$ . Notice that, as described in Section 4.5.1, we are including the linear damping term in the nonlinear vector  $\hat{\mathbf{n}}(q_p, q_m)$ .

 $<sup>^{62}\</sup>text{As}$  before  $\mathbf{n}_2=0,...$  etc. so we set  $\mathbf{n}_1=\mathbf{n},$  and correspondingly,  $\mathbf{h}_1=\mathbf{h},$   $\mathbf{b}_1=\mathbf{b}$  and  $\hat{\mathbf{n}}_{u1}=\hat{\mathbf{n}}_u$  throughout this example.

Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.

Now apply a near-identity nonlinear transform, Eq. (116), of the form

$$q = u + \varepsilon \mathbf{b} \mathbf{u}^*$$
 where  $\mathbf{b} = [b_1 \ b_2 \ b_3 \ b_4 \ b_5 \ b_6],$  (276)

and  $\mathbf{u}^* = [u_p^3 \ u_p^2 u_m \ u_p u_m^2 \ u_m^3 \ u_p \ u_m]^T$ . The transformed dynamic equation, Eq. (146), is expressed in the new coordinates as

$$\ddot{u} + \omega_n^2 u + \varepsilon \hat{\mathbf{n}}_u(u_p, u_m) = 0.$$
(277)

Substituting Eq. (276) and Eq. (277) into Eq. (274) (with detuning) leads to the  $\varepsilon^1$  homological equation  $\hat{\mathscr{H}}_1(\mathbf{u})$  given by

$$\varepsilon^{1}: -\mathbf{b}\frac{\mathrm{d}^{2}\mathbf{u}^{*}}{\mathrm{d}t^{2}} - \omega_{r}^{2}\mathbf{b}\mathbf{u}^{*} = \hat{\mathbf{n}} - \hat{\mathbf{n}}_{\mathbf{u}} = -\mathbf{b}\Lambda_{d}^{*2}\mathbf{u}^{*} - \Upsilon\mathbf{b}\mathbf{u}^{*}.$$
(278)

Then using  $\hat{\mathbf{n}}_{\mathbf{u}} = [\hat{n}_{u1} \ \hat{n}_{u2} \ \hat{n}_{u3} \ \hat{n}_{u4} \ \hat{n}_{u5} \ \hat{n}_{u6}]\mathbf{u}^*$ , the second Lie derivative Eq. (153), with  $q_i \rightarrow u_i$  (to order  $\varepsilon^0$ ) gives

Evaluating the right hand side gives four zero values which allows the  $\hat{\mathbf{n}}_{\mathbf{u}}$  coefficients to be identified such that

$$\left(\mathbf{i}\omega_{r}\left[-\frac{\omega_{r}^{2}}{3} \ \omega_{r}^{2} \ -\omega_{r}^{2} \ \frac{\omega_{r}^{2}}{3} \ -1 \ 1\right] - \mathbf{i}\omega_{r}\left[0 \ \omega_{r}^{2} \ -\omega_{r}^{2} \ 0 \ -1 \ 1\right]\right)\mathbf{u}^{*} = \left[b_{1}(8\omega_{r}^{2}) \ b_{2}(0) \ b_{3}(0) \ b_{4}(8\omega_{n}^{2}) \ b_{5}(0) \ b_{6}(0)\right]\mathbf{u}^{*}$$
(280)

From this the transform coefficient values are  $b_1 = i\omega_r/24$ ,  $b_4 = -i\omega_r/24$ , and  $b_2 = b_3 = b_5 = b_6 = 0$ . Next, using the fact that

$$\varepsilon \hat{\mathbf{n}}_{\mathbf{u}}(u_p, u_m) = \varepsilon \mathbf{i} \omega_r [0 \ \omega_r^2 \ -\omega_r^2 \ 0 \ -1 \ 1] \mathbf{u}^* = \varepsilon \mathbf{i} \omega_r (\omega_r^2 u_p^2 u_m - \omega_r^2 u_p u_m^2 - u_p + u_m) = \varepsilon \mathbf{i} \omega_r (\omega_r^2 u_p u_m - 1) (u_p - u_m),$$

and noting that  $\mathbf{i}\omega_r(u_p - u_m) = \dot{u}$  gives the  $\varepsilon^1$  real normal form as

$$\ddot{u} + \omega_n^2 u + \varepsilon (\omega_r^2 u_p u_m - 1) \dot{u} = 0, \tag{281}$$

Substituting the base solutions into Eq. (281) gives

$$(\mathbf{i}\omega_r)^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \omega_n^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \mathbf{i}\omega_r \varepsilon (\omega_r^2 \frac{U^2}{4} - 1) \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} - \mathbf{e}^{-\mathbf{i}\omega_r t}) = 0.$$
(282)

The real and imaginary parts of the  $e^{i\omega_r t}$  terms (or the  $e^{-i\omega_r t}$ ) can be balanced to give

Re: 
$$\omega_r^2 = \omega_n^2$$
,  $\rightsquigarrow \quad \omega_r = \omega_n$ ,  
Im:  $\omega_r \varepsilon (\omega_r^2 \frac{U^2}{4} - 1) = 0$ ,  $\rightsquigarrow \quad U = \sqrt{\frac{4}{\omega_r^2}}$ ,
(283)

which is the same as the unforced Van der Pol oscillator with parameters  $\mu = \omega_r^2$  and  $\gamma = 1$ . Using  $\omega_r = \omega_n$ , the near-identity transform, to order  $\varepsilon^1$  may now be written as

$$x = u + \varepsilon \mathbf{b} \mathbf{u}^* = \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_n t} + \mathbf{e}^{-\mathbf{i}\omega_n t}) + \varepsilon \left[ \frac{\mathbf{i}\omega_n}{24} \ 0 \ 0 \ \frac{-\mathbf{i}\omega_n}{24} \ 0 \ 0 \right] \mathbf{u}^*,$$
(284)

which gives the solution for x as

$$x = \sqrt{\frac{4}{\omega_n^2}} \cos(\omega_n t) - \varepsilon \frac{\omega_n}{96} \left(\sqrt{\frac{4}{\omega_n^2}}\right)^3 \sin(3\omega_n t) + \mathcal{O}(\varepsilon^2).$$
 (285)

Using this type of real normal form approach, the Rayleigh and Van der Pol oscillators therefore have the same normal form structure, and only differ in terms of the constants in the nonlinear term. Next we consider how the choice of the free functions can be used.

### 4.7 Choice of free functions

As was previously noted, if zeros occur in the homological equation, then the  $b_i$  coefficient that corresponds to the zero can be chosen to have any value without changing the homological equation. We follow [85] in calling the  $b_i$  coefficients the 'free functions'. The key point to note is that although the free functions do not change the homological equation they do change the solution for x. In most of the examples shown so far,  $b_i = 0$  is chosen whenever there was a free function choice, except in Example 13, Section 4.4.2, where we showed how a specific choice of the free function could be used to alter the solution for x (see Eq. (226) and the related discussion).

Another concept, called the *minimal normal form*, uses the choice of free functions to try an eliminate terms at higher order in the solution for *x*. For example, in the case where we are truncating at order  $\varepsilon^1$ , then the idea is to try and eliminate terms at order  $\varepsilon^2$  and above. In order to carry out a minimal normal form, we need to compute the homological equation truncated at second order (or higher), which is taken as the second line in Eq. (261).

Now consider eliminating the  $\varepsilon^2$  terms (and higher if required) in the solution for *x*. In this context it means setting  $\mathbf{h}_2 = 0$  (i.e setting all the  $b_i$  coefficients to zero for  $\mathbf{h}_2$ ). It can be seen that if this is done for the second line in Eq. (261), the result is

$$\Delta \mathbf{h}_1 + \frac{\partial \hat{\mathbf{n}}_1}{\partial \mathbf{u}} \mathbf{h}_1 + \hat{\mathbf{n}}_2 - \hat{\mathbf{n}}_{\mathbf{u}2} = 0.$$
(286)

So, to obtain the minimal normal form to order  $\varepsilon^2$  step one is to solve the  $\varepsilon^1$  homological equation (first line in Eq. (261)) to find  $\mathbf{h}_1$ , then use Eq. (286) to solve for  $\hat{\mathbf{n}}_{u2}$ . An example will be considered next.

#### 4.7.1 Example 17: The minimal normal form for the undamped cubic-quintic oscillator

We consider the cubic-quintic oscillator from Example 12 (Section 4.2.4) written as

$$\ddot{x} + \omega_n^2 x + \varepsilon \alpha_1 x^3 + \varepsilon \alpha_2 x^5 = 0 \quad \text{with } x = q \rightsquigarrow \quad \ddot{q} + \omega_n^2 q + \varepsilon \alpha_1 q^3 + \varepsilon \alpha_2 q^5 = 0, \quad \rightsquigarrow \\ \ddot{q} = -\omega_n^2 q - \varepsilon \hat{\mathbf{n}}_1(q_p, q_m) = 0, \tag{287}$$

from which the  $\varepsilon^1$  homological equation gives (see Eq. (164))

$$([\alpha_1 \ 3\alpha_1 \ \alpha_2 \ 1\alpha_2 \ 5\alpha_2 \ 10\alpha_2 \ 10\alpha_2 \ 5\alpha_2 \ \alpha_2] - [0 \ 3\alpha_1 \ 3\alpha_1 \ 0 \ 0 \ 10\alpha_2 \ 10\alpha_2 \ 0 \ 0]) \mathbf{u}^* = [b_1(8\omega_r^2) \ b_2(0) \ b_3(0) \ b_4(8\omega_r^2) \ b_5(24\omega_r^2) \ b_6(8\omega_r^2) \ b_7(0) \ b_8(0) \ b_9(8\omega_r^2) \ b_{10}(24\omega_r^2)] \mathbf{u}^*,$$

$$(288)$$

although here we apply detuning using the analysis in Section 4.6 so that the expression has  $\omega_r$  rather than  $\omega_n$  as in Eq. (164). In this case the coefficient values are  $b_1 = b_4 = \alpha_1/8\omega_r^2$ ,  $b_6 = b_9 = 5\alpha_2/8\omega_r^2$ ,  $b_5 = b_{10} = \alpha_2/24\omega_r^2$ , and  $b_2 = b_3 = b_7 = b_8 = 0$  which means that the  $\varepsilon^1$  term in Eq. (116) will be given by

$$\mathbf{h}_{1} = \mathbf{b}_{1}\mathbf{u}^{*} = \frac{\alpha_{1}}{8\omega_{r}^{2}}u_{p}^{3} + \frac{\alpha_{1}}{8\omega_{r}^{2}}u_{m}^{3} + \frac{\alpha_{2}}{24\omega_{r}^{2}}u_{p}^{5} + \frac{5\alpha_{2}}{8\omega_{r}^{2}}u_{p}^{4}u_{m} + \frac{5\alpha_{2}}{8\omega_{r}^{2}}u_{p}u_{m}^{4} + \frac{\alpha_{2}}{24\omega_{r}^{2}}u_{m}^{5}.$$
(289)

We can use this to find the minimal normal form for the cubic-quintic oscillator using Eq. (286). To do this note that for the single-degree-of-freedom oscillator such as this  $\mathbf{u} = u$ , so from Eq. (287),

$$\hat{\mathbf{n}}_{1}(q) = \alpha_{1}q^{3} + \alpha_{3}q^{5} \quad \rightsquigarrow \quad \alpha_{1}(u + \varepsilon \mathbf{h}_{1})^{3} + \alpha(u + \varepsilon \mathbf{h}_{1})^{5} \quad \rightsquigarrow \quad (\alpha_{1}(u)^{3} + \alpha_{2}(u)^{5})|_{\varepsilon=0} \quad \rightsquigarrow \\ \frac{\partial \hat{\mathbf{n}}_{1}(u)}{\partial u} = 3\alpha_{1}(u)^{2} + 5\alpha_{2}(u)^{4} = 3\alpha_{1}(u_{p} + u_{m})^{2} + 5\alpha_{2}(u_{p} + u_{m})^{4}$$
(290)

where  $\varepsilon = 0$  has been used in the partial derivative, so that when the resulting expression are substituted into the second line of Eq. (261) then only terms up to order- $\varepsilon^2$  are retained.

Now we can write Eq. (286) for the cubic-quintic oscillator (noting that  $\hat{n}_2 = 0$  from Eq. (287)) as

$$(\Delta + 3\alpha_1(u_p + u_m)^2 + 5\alpha_2(u_p + u_m)^4) \left(\frac{\alpha_1}{8\omega_r^2}u_p^3 + \frac{\alpha_1}{8\omega_r^2}u_m^3 + \frac{\alpha_2}{24\omega_r^2}u_p^5 + \frac{5\alpha_2}{8\omega_r^2}u_p^4u_m + \frac{5\alpha_2}{8\omega_r^2}u_pu_m^4 + \frac{\alpha_2}{24\omega_r^2}u_m^5\right) = \hat{\mathbf{n}}_{\mathbf{u}2}$$

$$\Rightarrow \hat{\mathbf{n}}_{\mathbf{u}2} = A_1u_p^3 + A_2u_m^3 + A_3u_p^5 + A_4u_p^4u_m + A_5u_p^3u_m^2 + A_6u_p^2u_m^3 + A_7u_pu_m^4 + A_8u_m^5 + \mathcal{O}(k = 7)$$
(291)

where  $\Delta = \omega_n^2 - \omega_r^2$  and

$$A_1 = A_2 = \frac{\Delta \alpha_1}{8\omega_r^2}, \quad A_3 = A_8 = \frac{9\alpha_1^2 + \Delta \alpha_2}{24\omega_r^2}, \quad A_4 = A_7 = \frac{6\alpha_1^2 + 5\Delta \alpha_2}{8\omega_r^2}, \quad A_5 = A_6 = \frac{3\alpha_1^2}{8\omega_r^2}, \tag{292}$$

and we have included terms up to k = 5 order in Eq. (291). Then by adding the  $\varepsilon^2$  (e.g up to k = 5) terms to the  $\varepsilon$  normal form, Eq. (165), (e.g. to obtain Eq. (146)), the minimal normal form becomes

$$\ddot{u} + \omega_n^2 u + \varepsilon 3\alpha (u_p^2 u_m + u_p u_m^2) + \varepsilon 10\alpha_2 (u_p^3 u_m^2 + u_p^2 u_m^3) + \varepsilon^2 (A_1 u_p^3 + A_2 u_m^3 + A_3 u_p^5 + A_4 u_p^4 u_m + A_5 u_p^3 u_m^2 + A_6 u_p^2 u_m^3 + A_7 u_p u_m^4 + A_8 u_m^5) + \mathscr{O}(\varepsilon^3, k = 7) = 0.$$
(293)

Then the assumed base solutions (from Eq. (22)), are substituted into Eq. (116) (lower) to give the solution for x as

$$x = U\cos(\omega_r t) + \varepsilon \left(\frac{\alpha_1 U^3}{32\omega_r^2} + \frac{5\alpha_2 U^5}{128\omega_r^2}\right)\cos(3\omega_r t) + \varepsilon \frac{\alpha_2 U^5}{384\omega_r^2}\cos(5\omega_r t) + \mathcal{O}(\varepsilon^3). \qquad \Box$$
(294)

Note that due to the fact that this is a single-degree-of-freedom oscillator and  $\mathbf{h}_2 = 0$ , Eq. (294) is given by  $x = q = u + \varepsilon \mathbf{h}_1 + \varepsilon^2(0)$ . When Eq. (294) is compared with Eq. (168) from Example 12, it can be seen that the expressions are almost the same except Eq. (294) has the detuned frequency,  $\omega_r$ , and is computed to order  $\mathcal{O}(\varepsilon^3)$  rather than  $\mathcal{O}(\varepsilon^2)$ . As can be seen from this example, this type of manipulation is just forcing all the terms that would appear in the solution for x, Eq. (294) into the normal form Eq. (293). The complexity of Eq. (293) is significantly increased, which may be a disadvantage for some applications. If instead of this we set  $\mathbf{h}_2 \neq 0$  in the  $\varepsilon^2$  homological equation (second line in Eq. (261)) then we can obtain the real normal form to order  $\varepsilon^2$ . This will be demonstrated in Example 18 (Section 4.8.1).

#### 4.8 Backbone curves

An important application that can be achieved using normal form is the computation of backbone curves. The idea of a backbone curve was introduced in Section 1.1.3, and an example was shown in Fig. 2. To compute backbone curves using the types of normal form transformations described in this paper, detuning needs to be included. This is because, the backbone curve represents (an approximation) of the nonlinear amplitude-frequency variation which is a fundamental characteristic of many nonlinear oscillators. For example, the backbone curve relationship shown in Fig. 2, can be derived from the detuned version of Example 12 (Section 4.2.4). From Example 12, substituting the (detuned) base solutions (Eq. (22)) into Eq. (165),

$$(\mathbf{i}\omega_r)^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \omega_n^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \varepsilon 3\alpha \frac{U^3}{8} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) = 0.$$
(295)

The coefficients of the  $e^{i\omega_r t}$  terms (or the  $e^{-i\omega_r t}$ ) can be balanced to give

$$\omega_r^2 = \omega_n^2 + \varepsilon \frac{3\alpha U^2}{4} + \mathscr{O}(\varepsilon^2), \tag{296}$$

from which, when  $\varepsilon = 1$ , we can obtain

$$\omega_r = \sqrt{\omega_n^2 + \alpha \frac{3U^2}{4} + \mathcal{O}(2)}.$$
(297)



**Fig. 7:** Conservative backbone curves, showing (a) examples of conservative backbone curves for the Duffing oscillator computed using Eq. (297) for different values of  $\alpha$  when  $\omega_n = 1$ , and (b) a schematic diagram of the conservative backbone concept. Here the conservative backbone curve is defined as the curve that joins all the points of maximum displacement for a family of periodic orbits generated by the nonlinear oscillator under investigation. For clarity, just two periodic orbits are shown here (blue curves). The maximum (and minimum) points of displacement amplitude are shown as black dots, and both the positive and negative backbone curves are shown (red curves). Typically just the upper half of this diagram is shown (as in panel (a)), and symmetry is assumed for the lower half. Note that units of frequency are in radians per second, and units of displacement amplitude, U are in mm or m.

It should be noted that the example shown in Fig. 2 had negative  $\alpha$  and so substituting  $-\alpha$  into Eq. (296) gives the same relationship as Eq. (26) from Example 2. Backbone curves computed for undamped, unforced systems are called *conservative* backbone curves. A schematic diagram of the concept and example curves for Eq. (297) are shown in Fig. 7. We now consider computing the backbone curve to order  $\varepsilon^2$  for the undamped cubic-quintic oscillator

### 4.8.1 Example 18: the $\varepsilon^2$ real normal form of the undamped cubic-quintic oscillator

The  $\varepsilon^2$  real normal form of the undamped cubic-quintic oscillator can be found by extending the analysis from the order- $\varepsilon^1$  approach shown in Example 12, Section 4.2.4 (noting that  $\hat{\mathbf{n}}_2 = 0$  from Eq. (287), and that the quintic nonlinear terms are included in  $\hat{\mathbf{n}}_1$  in these examples). To do this, we first express Eq. (291) as

$$\Delta \mathbf{h}_{1} + \frac{\partial \hat{\mathbf{n}}}{\partial \mathbf{u}} \mathbf{h}_{1} = A_{1} u_{p}^{3} + A_{2} u_{m}^{3} + A_{3} u_{p}^{5} + A_{4} u_{p}^{4} u_{m} + A_{5} u_{p}^{3} u_{m}^{2} + A_{6} u_{p}^{2} u_{m}^{3} + A_{7} u_{p} u_{m}^{4} + A_{8} u_{m}^{5} = \hat{\mathbf{n}}^{+} \mathbf{u}^{+},$$
(298)

where

$$\hat{\mathbf{n}}^{+} = \begin{bmatrix} A_{1} \ A_{2} \ A_{3} \ A_{4} \ A_{5} \ A_{6} \ A_{7} \ A_{8} \end{bmatrix} \text{ and } \mathbf{u}^{+} = \begin{bmatrix} u_{p}^{3} \\ u_{m}^{3} \\ u_{p}^{5} \\ u_{p}^{4} u_{m} \\ u_{p}^{2} u_{m}^{3} \\ u_{p} u_{m}^{4} \\ u_{p}^{5} \end{bmatrix},$$

$$(299)$$

such that if we use  $\mathbf{u}^+$  as the basis for all the terms in the  $\varepsilon^2$  homological equation, and the  $A_j$  coefficients are given in Eq. (292). Then (the second line in) Eq. (261) becomes

$$\hat{\mathbf{n}}^+ \mathbf{u}^+ - \hat{\mathbf{n}}_{\mathbf{u}2}^+ \mathbf{u}^+ = -\mathbf{b}_2 \tilde{\Upsilon}^2 \mathbf{u}^+ - \Upsilon \mathbf{b}_2 \mathbf{u}^+, \tag{300}$$

where we have used the second Lie derivative (Eq. (153)) in order to evaluate  $\ddot{\mathbf{h}}_2$  and thus find  $\tilde{\Upsilon}^2$ . In addition,  $\hat{\mathbf{n}}_{\mathbf{n}^2}^+$  and  $\mathbf{b}_2$  are coefficient matrices of the same dimension as  $\hat{\mathbf{n}}^+$ .

Computing the  $\varepsilon^2$  homological equation for the undamped cubic-quintic oscillator gives

$$([A_1 \ A_2 \ A_3 \ A_4 \ A_5 \ A_6 \ A_7 \ A_8] - [\hat{n}_{u21} \ \hat{n}_{u22} \ \hat{n}_{u23} \ \hat{n}_{u24} \ \hat{n}_{u25} \ \hat{n}_{u26} \ \hat{n}_{u27} \ \hat{n}_{u28}]) \mathbf{u}^+ =$$

$$(\mathbf{u}_{\mathbf$$

Evaluating the right hand side gives two zero values which allows the  $\hat{n}_{u2}^+$  coefficients to be identified such that

$$([A_1 \ A_2 \ A_3 \ A_4 \ A_5 \ A_6 \ A_7 \ A_8] - [0 \ 0 \ 0 \ A_5 \ A_6 \ 0 \ 0]) \mathbf{u}^+ = [b_1(8\omega_r^2) \ b_2(8\omega_r^2) \ b_3(24\omega_r^2) \ b_4(8\omega_n^2) \ b_5(0) \ b_6(0) \ b_7(8\omega_r^2) \ b_8(24\omega_r^2)] \mathbf{u}^+.$$

$$(302)$$

Next we substitute for  $A_5$  and  $A_6$  from Eq. (292) and then use the fact that

$$\varepsilon^{2} \hat{\mathbf{n}}_{\mathbf{u2}}(u_{p}, u_{m}) = \varepsilon^{2} \begin{bmatrix} 0 & 0 & 0 & \frac{3\alpha_{1}^{2}}{8\omega_{r}^{2}} & \frac{3\alpha_{1}^{2}}{8\omega_{r}^{2}} & 0 & 0 \end{bmatrix} \mathbf{u}^{+} = \varepsilon^{2} \frac{3\alpha_{1}^{2}}{8\omega_{r}^{2}} (u_{p}^{3}u_{m}^{2} + u_{p}^{2}u_{m}^{3}),$$

which is up to k = 5, as for Example 17. Then by adding the  $\varepsilon^2$  (e.g up to k = 5) terms to the  $\varepsilon$  normal form, Eq. (165), (e.g. to obtain Eq. (146)), the  $\varepsilon^2$  normal form becomes

$$\ddot{u} + \omega_n^2 u + \varepsilon 3\alpha_1 (u_p^2 u_m + u_p u_m^2) + \left(\varepsilon 10\alpha_2 + \varepsilon^2 \frac{3\alpha_1^2}{8\omega_r^2}\right) (u_p^3 u_m^2 + u_p^2 u_m^3) = 0.$$
(303)

Using the values for  $\mathbf{b}_1$  from Eq. (289), and  $\mathbf{b}_2$  from solving Eq. (302), the near-identity transform, to order  $\varepsilon^2$  may now be written as

$$x = u + \varepsilon \mathbf{b}_{1} \mathbf{u}^{*} + \varepsilon^{2} \mathbf{b}_{2} \mathbf{u}^{+} = \frac{U}{2} \left( e^{\mathbf{i}(\omega_{r}t)} + e^{-\mathbf{i}(\omega_{r}t)} \right) + \varepsilon \left[ \frac{\alpha_{1}}{8\omega_{r}^{2}} \ 0 \ 0 \ \frac{\alpha_{1}}{8\omega_{r}^{2}} \ \frac{\alpha_{2}}{24\omega_{r}^{2}} \ \frac{5\alpha_{2}}{8\omega_{r}^{2}} \ 0 \ 0 \ \frac{5\alpha_{2}}{8\omega_{r}^{2}} \ \frac{\alpha_{2}}{24\omega_{r}^{2}} \right] \mathbf{u}^{*} + \varepsilon^{2} \left[ \frac{\Delta\alpha_{1}}{64\omega_{r}^{4}} \ \frac{\Delta\alpha_{1}}{64\omega_{r}^{4}} \ \frac{9\alpha_{1}^{2} + \Delta\alpha_{2}}{576\omega_{r}^{4}} \ \frac{6\alpha_{1}^{2} + 5\Delta\alpha_{2}}{64\omega_{r}^{4}} \ 0 \ 0 \ \frac{6\alpha_{1}^{2} + 5\Delta\alpha_{2}}{64\omega_{r}^{4}} \ \frac{9\alpha_{1}^{2} + \Delta\alpha_{2}}{576\omega_{r}^{4}} \right] \mathbf{u}^{+},$$
(304)

where  $\mathbf{u}^*$  is given by Eq. (162). Then the assumed base solutions (from Eq. (22)), are substituted into Eq. (304) to give the solution for *x* as  $x = q = u + \varepsilon \mathbf{h}_1 + \varepsilon^2 \mathbf{h}_2$  which leads to

$$x = U\cos(\omega_{r}t) + \left[\varepsilon\frac{\alpha_{1}U^{3}}{32\omega_{r}^{2}} + \varepsilon\frac{5\alpha_{2}U^{5}}{128\omega_{r}^{2}} + \varepsilon^{2}\frac{\Delta\alpha_{1}U^{3}}{256\omega_{r}^{4}} + \varepsilon^{2}\frac{3\alpha_{1}^{2}U^{5}}{512\omega_{r}^{4}} + \varepsilon^{2}\frac{\Delta\alpha_{2}U^{5}}{1024\omega_{r}^{4}}\right]\cos(3\omega_{r}t) \\ + \left[\varepsilon\frac{\alpha_{2}U^{5}}{384\omega_{r}^{2}} + \varepsilon^{2}\frac{\alpha_{1}^{2}U^{5}}{1024\omega_{r}^{4}} + \varepsilon^{2}\frac{\Delta\alpha_{2}U^{5}}{9216\omega_{r}^{4}}\right]\cos(5\omega_{r}t) + \mathcal{O}(\varepsilon^{3}, k = 7).$$
(305)

To find an expression for  $\omega_r^2$  the (detuned) base solutions (Eq. (22)) are taken to be the assumed solution, and are substituted into Eq. (303) to give

$$(\mathbf{i}\omega_r)^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \omega_n^2 \frac{U}{2} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \varepsilon 3\alpha_1 \frac{U^3}{8} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) + \left(\varepsilon 10\alpha_2 + \varepsilon^2 \frac{3\alpha_1^2}{8\omega_r^2}\right) \frac{U^5}{32} (\mathbf{e}^{\mathbf{i}\omega_r t} + \mathbf{e}^{-\mathbf{i}\omega_r t}) = 0.$$

*Journal of Structural Dynamics*, 1, (pp. 138-216) 2022 Normal form transformations for structural dynamics: An introduction for linear and nonlinear systems.



**Fig. 8:** Numerically computed conservative backbone curves for the oscillator in Example 18 with  $\omega_n = \sqrt{10}$  rad/s, and  $\varepsilon = 1$ . Showing (a) the case for  $\alpha_1 = -0.5$  &  $\alpha_2 = 0.1$ , and (b) the case when  $\alpha_1 = 0.5$  &  $\alpha_2 = -0.1$ . The blue dots are numerically computed maximum values of the displacement amplitude, and are taken as the reference solution. The blue line is the  $\varepsilon^1$  backbone curve solution computed by substituting the assumed solution into Eq. (165), and the purple line is the  $\varepsilon^2$  backbone curve solution computed from Eq. (307). It can be noted from both plots that as *U* increases, there is a combination of both softening and hardening frequency behaviour. In (a) the backbone curve first softens and then hardens as *U* increases, whereas in (b) (where the coefficients  $\alpha_1 \& \alpha_2$  have the opposite signs) the backbone curve first hardens and then softens as *U* increases. From a perturbation viewpoint, it could be expected that the solutions will give better agreement with the reference solution for "small" *U* (and  $\alpha_1, \alpha_2$ ) values. In this interpretation, for values U > 1 we should be cautious about how accurate the predictions from these types of approximate solutions might be.

(306)

The coefficients of the  $e^{i\omega_r t}$  terms (or the  $e^{-i\omega_r t}$ ) can be balanced to give

$$\omega_r^2 = \omega_n^2 + \varepsilon \frac{3\alpha_1 U^2}{4} + \varepsilon \frac{5\alpha_2 U^4}{8} + \varepsilon^2 \frac{3\alpha_1^2 U^4}{128\omega_r^2} + \mathcal{O}(\varepsilon^3, k = 7).$$
 (307)

A numerically computed example is shown in Fig. 8 for the case when  $\alpha_2 = 0$ . Further details of this type of analysis is given in Nasir et al. [122].

### 4.9 Normal form transformations of bifurcation phenomena

An important use of normal form transformation in the field of nonlinear dynamics is to simplify models of important dynamic phenomena, such as bifurcations. From a structural dynamics perspective, we can start with model containing multiple terms and coefficients, and use the normal form process to eliminate as many of them as possible at a given order, resulting in a simpler system with fewer coefficients [100, 78, 61]. In this use of normal form transformations, it is not necessarily expected that the process will provide an approximate solution of the equations. Instead, a systematic reduction of terms (and equations) is carried out, until there is no further reduction possible. Ultimately, this can result in just a single equation and parameter remaining that determines the behaviour of the bifurcation. In other words, a reduction to the simplest non-trivial case is sometimes possible. We consider the example of the Hopf bifurcation.

#### 4.9.1 Example 19 a normal form analysis for a Hopf bifurcation

We revisit the unforced Van der Pol oscillator from Example 15, and start by writing it in a slightly modified form as

$$\ddot{x} + (\mu x^2 - 2\zeta \omega_n) \dot{x} + \omega_n^2 x = 0, \quad \rightsquigarrow \quad \dot{\mathbf{x}} = A\mathbf{x} + \tilde{\mathcal{N}}(\mathbf{x}): \quad A = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & 2\zeta \omega_n \end{bmatrix}, \quad \tilde{\mathcal{N}}(\mathbf{x}) = \begin{cases} 0 \\ -\mu x_1^2 x_2 \end{cases}, \quad (308)$$

where the states are  $\mathbf{x} = \{x_1 \ x_2\}^T = \{x \ \dot{x}\}^T$ . Next we will use the Jordan normal form technique from Section 2.2, specifically Example 6, to transform the systems so that the linear matrix A has the real form for complex eigenvalues.

To do this we first note that the eigenvalues of A in this example are  $\lambda_{1,2} = \zeta \omega_n \pm i\omega_d$ , where  $\omega_d = \omega_n \sqrt{1-\zeta^2}$ is the damped natural frequency. Therefore, for the linear part of the problem we have a similar situation to that in Example 6, and we can make the coordinate transformation x = Pq where in this example

$$P = \begin{bmatrix} 0 & 1\\ \omega_d & \zeta \omega_n \end{bmatrix} , \quad P^{-1} = \begin{bmatrix} \frac{-\zeta \omega_n}{\omega_d} & \frac{1}{\omega_d}\\ 1 & 0 \end{bmatrix} \text{ and } \tilde{\mathscr{N}}(P\boldsymbol{q}) = \begin{cases} 0\\ -\mu q_m^2(\omega_d q_p + \zeta \omega_n q_m) \end{cases} , \tag{309}$$

where  $\boldsymbol{q} = \{q_p | q_m\}^T$ . Next we can write down the transformed system as

$$\dot{\boldsymbol{q}} = P^{-1}AP\boldsymbol{q} + P^{-1}\tilde{\mathscr{N}}(P\boldsymbol{q}) \quad \rightsquigarrow \quad \begin{bmatrix} \dot{q}_p \\ \dot{q}_m \end{bmatrix} = \begin{bmatrix} \zeta \omega_n & -\omega_d \\ \omega_d & \zeta \omega_n \end{bmatrix} \begin{bmatrix} q_p \\ q_m \end{bmatrix} + \begin{cases} -\mu q_m^2 q_p - \mu \gamma q_m^3 \\ 0 \end{cases} \right\}. \tag{310}$$

where  $\gamma = \frac{\zeta \omega_n}{\omega_d}$ . Next we use a process of *complexification* to obtain new coordinates of the form  $z = q_p + \mathbf{i}q_m$  and  $\bar{z} = q_p - \mathbf{i}q_m$ , so that  $q_p = \frac{z+\bar{z}}{2}$ ,  $q_m = \frac{z-\bar{z}}{2\mathbf{i}}$ ,  $\dot{z} = \dot{q}_p + \mathbf{i}\dot{q}_m$  and  $\dot{\bar{z}} = \dot{q}_p - \mathbf{i}\dot{q}_m$ . Using these relationships with Eq. (310) we then obtain

$$\dot{z} = \lambda z - \mu \left(\frac{z - \bar{z}}{2\mathbf{i}}\right)^2 \left(\frac{z + \bar{z}}{2}\right) - \mu \gamma \left(\frac{z - \bar{z}}{2\mathbf{i}}\right)^3 \quad \text{and} \quad \dot{\bar{z}} = \bar{\lambda} \bar{z} - \mu \left(\frac{z - \bar{z}}{2\mathbf{i}}\right)^2 \left(\frac{z + \bar{z}}{2}\right) - \mu \gamma \left(\frac{z - \bar{z}}{2\mathbf{i}}\right)^3, \tag{311}$$

where  $\lambda = \lambda_1 = \zeta \omega_n + i\omega_d$  and  $\lambda = \lambda_2 = \zeta \omega_n - i\omega_d$ .

Now expanding the nonlinear terms and projecting them onto the nonlinear basis vector  $\mathbf{z}^* = [z^3 \ z^2 \overline{z} \ \overline{z}^3]^T$ we obtain the system

$$\begin{bmatrix} \dot{z} \\ \dot{\bar{z}} \end{bmatrix} = \begin{bmatrix} \lambda & 0 \\ 0 & \bar{\lambda} \end{bmatrix} \begin{bmatrix} z \\ \bar{z} \end{bmatrix} + \frac{\mu}{8} \begin{bmatrix} 1 - \mathbf{i}\gamma & -1 + \mathbf{i}3\gamma & -1 - \mathbf{i}3\gamma & 1 + \mathbf{i}\gamma \\ 1 - \mathbf{i}\gamma & -1 + \mathbf{i}3\gamma & -1 - \mathbf{i}3\gamma & 1 + \mathbf{i}\gamma \end{bmatrix} \mathbf{z}^* \quad \text{or} \quad \dot{z} = \Lambda z + N_z \mathbf{z}^*, \tag{312}$$

where 
$$\mathbf{z} = [z, \bar{z}]^T$$
,  $\Lambda = \begin{bmatrix} \lambda & 0 \\ 0 & \bar{\lambda} \end{bmatrix}$  and  $N_z = \frac{\mu}{8} \begin{bmatrix} 1 - \mathbf{i}\gamma & -1 + \mathbf{i}3\gamma & 1 - \mathbf{i}3\gamma & 1 + \mathbf{i}\gamma \\ 1 - \mathbf{i}\gamma & -1 - \mathbf{i}3\gamma & 1 - \mathbf{i}\gamma \end{bmatrix}$ . (313)

We now seek a near-identity normal form transformation from z to a new coordinate set  $w = [w, \bar{w}]^T$ , such that  $z = w + bw^*$  where b is a coefficient matrix,  $w^* = [w^3 \ w^2 \bar{w} \ w \bar{w}^2 \ \bar{w}^3]^T$ , and the transformed system is governed by

$$\dot{\boldsymbol{w}} = \Lambda \boldsymbol{w} + N_{\boldsymbol{w}} \boldsymbol{w}^*, \tag{314}$$

where  $N_w$  is the new (simplified) coefficient matrix. Substituting for z in Eq. (312) and using Eq. (314) leads to a homological equation of the form

$$N_z - N_w = \boldsymbol{b} \frac{\partial \mathbf{w}^*}{\partial \boldsymbol{w}} \Delta \boldsymbol{w} - \Delta \boldsymbol{b} \mathbf{w}^*, \tag{315}$$

which can be written as

$$\begin{pmatrix} \mu \\ \overline{8} \begin{bmatrix} 1 - \mathbf{i}\gamma & -1 + \mathbf{i}3\gamma & -1 - \mathbf{i}3\gamma & 1 + \mathbf{i}\gamma \\ 1 - \mathbf{i}\gamma & -1 + \mathbf{i}3\gamma & -1 - \mathbf{i}3\gamma & 1 + \mathbf{i}\gamma \end{bmatrix} - \begin{bmatrix} n_{w1}^* & n_{w2}^* & n_{w3}^* & n_{w4}^* \\ n_{w5}^* & n_{w6}^* & n_{w7}^* & n_{w8}^* \end{bmatrix} \end{pmatrix} \mathbf{w}^* = \begin{pmatrix} b_1 & b_2 & b_3 & b_4 \\ b_5 & b_6 & b_7 & b_8 \end{bmatrix} \begin{bmatrix} 3\lambda & 0 & 0 & 0 \\ 0 & 2\lambda + \overline{\lambda} & 0 & 0 \\ 0 & 0 & \lambda + 2\overline{\lambda} & 0 \\ 0 & 0 & 0 & 3\overline{\lambda} \end{bmatrix} - \begin{bmatrix} \lambda & 0 \\ 0 & \overline{\lambda} \end{bmatrix} \begin{bmatrix} b_1 & b_2 & b_3 & b_4 \\ b_5 & b_6 & b_7 & b_8 \end{bmatrix} \end{bmatrix} \mathbf{w}^* =$$
(316)
$$\begin{bmatrix} b_1(2\lambda) & b_2(\lambda + \overline{\lambda}) & b_3(2\overline{\lambda}) & b_4(3\overline{\lambda} - \lambda) \\ b_5(3\lambda - \overline{\lambda}) & b_6(2\lambda) & b_7(\lambda + \overline{\lambda}) & b_8(2\overline{\lambda}) \end{bmatrix} \mathbf{w}^*.$$

Now we wish to consider what happens very close to the Hopf bifurcation point. To do this we need to consider what happens when the damping ratio,  $\zeta$ , is allowed to vary. The Hopf bifurcation will occur when  $\zeta = 0$ , and to understand the process, three cases are defined in Table 3. Case iii) is the initial case for this example, with the additional condition that  $\zeta > 0$ . Case ii) is the bifurcation point itself, and Case i) can be thought of as the case before the bifurcation happens. It should also be understood that Cases i) and iii) are very close to the bifurcation point<sup>63</sup>

Case i) $\zeta < 0$	Case ii) $\zeta=0$	Case iii) $\zeta > 0$
$\lambda = (-\zeta)\omega_n + \mathbf{i}\omega_d$	$\lambda = +\mathbf{i}\omega_n$	$\lambda = \zeta \omega_n + \mathbf{i} \omega_d$
$ar{\lambda} = (-\zeta)\omega_n - \mathbf{i}\omega_d$	$ar{\lambda}=-\mathbf{i}\omega_n$	$ar{\lambda} = \zeta \omega_n - \mathbf{i} \omega_d$
$\operatorname{Re}(\lambda, \overline{\lambda}) < 0$	$\operatorname{Re}(\lambda,\bar{\lambda})=0$	$\operatorname{Re}(\lambda,\bar{\lambda}) > 0$
$\lambda + ar{\lambda} = -2\zeta \omega_n$	$\lambda + ar{\lambda} = 0$	$\lambda + ar{\lambda} = 2 \zeta \omega_n$
$\lambda - ar{\lambda} = 2 \mathbf{i} \omega_d$	$\lambda - ar{\lambda} = 2\mathbf{i}\omega_n$	$\lambda - ar{\lambda} = 2\mathbf{i}\omega_d$
$\gamma = \frac{(-\zeta)\omega_n}{\omega_d}$	$\gamma = 1$	$\gamma = \frac{\zeta \omega_n}{\omega_d}$

Table 3: Three cases in the neighbourhood of the Hopf bifurcation for the Van der Pol oscillator.

It can be seen from Table 3 that  $\lambda + \overline{\lambda} = 0$  in Case ii) when  $\zeta = 0$ , and so we treat these terms as resonant and using Eq. (316), set all  $n_{wi}^*$  terms to zero, except for  $n_{w2}^* = \frac{\mu}{8}(-1+i3)$  and  $n_{w2}^* = \frac{\mu}{8}(-1-i3)$  such that using Eq. (314) the normal form can be written

$$\begin{bmatrix} \dot{w} \\ \dot{\bar{w}} \end{bmatrix} = \begin{bmatrix} \lambda & 0 \\ 0 & \bar{\lambda} \end{bmatrix} \begin{bmatrix} w \\ \bar{w} \end{bmatrix} + \frac{\mu}{8} \begin{bmatrix} 0 & -1 + \mathbf{i}3\gamma & 0 & 0 \\ 0 & 0 & -1 - \mathbf{i}3\gamma & 0 \end{bmatrix} \mathbf{w}^* \quad \text{or} \quad \begin{aligned} \dot{w} = \lambda w + n_{w2}^* w^2 \bar{w}, \\ \dot{\bar{w}} = \bar{\lambda} \bar{w} + n_{w7}^* w \bar{w}^2. \end{aligned}$$
(317)

Note that the eight coefficients in  $N_z$  from Eq. (313) has now been reduced to just two. Even so, further reduction and insight can be gained by changing to polar coordinates such that  $w = r(t)e^{i\theta(t)}$  and  $\bar{w} = r(t)e^{-i\theta(t)}$ . Substituting these relationships into the right-hand expression of Eq. (317) gives

$$\dot{r}e^{\mathbf{i}\theta} + r\mathbf{i}\dot{\theta}e^{\mathbf{i}\theta} = \lambda re^{\mathbf{i}\theta} + n_{w2}^*r^3e^{\mathbf{i}\theta}, \qquad \dot{r} + r\mathbf{i}\dot{\theta} = \lambda r + n_{w2}^*r^3,$$

$$\dot{r}e^{-\mathbf{i}\theta} - r\mathbf{i}\dot{\theta}e^{-\mathbf{i}\theta} = \bar{\lambda}re^{-\mathbf{i}\theta} + n_{w7}^*r^3e^{-\mathbf{i}\theta}, \qquad \overset{\sim}{\sim} \qquad \dot{r} - r\mathbf{i}\dot{\theta} = \bar{\lambda}r + \bar{n}_{w2}^*r^3,$$
(318)

where we have used the fact that  $n_{w7}^* = \bar{n}_{w2}^*$  in the right-hand expressions. Now treating the right-hand expression of Eq. (318) as a pair of simultaneous equations, and adding, then subtracting them leads to

where  $\operatorname{Re}(n_{w2}^*) = -\frac{\mu}{8}$ ,  $\operatorname{Im}(n_{w2}^*) = \frac{3\mu\gamma}{8}$  and  $\operatorname{Re}(\lambda) \& \operatorname{Im}(\lambda)$  can be obtained from Table 3. Now consider the equation for  $\dot{r}$  on the right-hand side of Eq. (319). The right-hand side of this equation is zero when

$$\operatorname{Re}(\lambda)r + \operatorname{Re}(n_{w2}^*)r^3 = 0 \quad \rightsquigarrow \quad \text{three possible solutions} \quad r = 0 \quad \text{and} \quad r = \pm \sqrt{-\frac{\operatorname{Re}(\lambda)}{\operatorname{Re}(n_{w2}^*)}}.$$
 (320)

The solution r = 0 exists for all three cases listed in Table 3, but the solutions  $r = \pm \sqrt{-\operatorname{Re}(\lambda)/(\operatorname{Re}(n_{w2}^*))}$  only exist for Case iii) when  $\zeta > 0$ , because  $\operatorname{Re}(n_{w2}^*) = -\frac{\mu}{8} < 0$  (assuming  $\mu > 0$ ) and so  $-\operatorname{Re}(\lambda)/(\operatorname{Re}(n_{w2}^*)) > 0$ .  $\Box$ 

<sup>&</sup>lt;sup>63</sup>A more rigorous mathematical approach is normally applied by defining the size of this region more specifically — see for example [77, 52]. Note also that the phrase 'in the neighbourhood of' is taken to mean 'sufficiently close to'.



**Fig. 9:** A schematic representation of the limit cycle (i.e. periodic orbit) behaviour obtained from the normal form of a (supercritical) Hopf bifurcation, as derived in Example 20 (Section 4.9.1). For  $\text{Re}(\lambda) < 0$  there is just one solution, r = 0, and the dynamics consists of stable spirals converging to the r = 0 equilibrium point. At  $\text{Re}(\lambda) = 0$  the Hopf bifurcation occurs and a limit cycle is created. When  $\text{Re}(\lambda) > 0$  there are now two further solutions that are shown as red vectors (in polar coordinates) in the figure. These solutions relate to the limit cycle (stable in this example) where the radius of the limit cycle corresponds to *r* and the angle,  $\theta$ , denotes the rotation (and  $-\theta$  the counter-rotation) of the two solution vectors. Furthermore when  $\text{Re}(\lambda) > 0$  the solution at r = 0 becomes unstable, and therefore there are now unstable spirals emanating from the equilibrium point, see discussions in [77, 55, 52, 162].

Of course, in the case when  $\mu < 0$ , then the sign of  $\operatorname{Re}(n_{w2}^*)$  changes, meaning that the additional solutions will occur instead for Case i), when  $\zeta < 0^{64}$ . So it can be seen that these successive coordinate changes, i.e. normal form transformations, have reduced the system to the point where the sign of just a single coefficient can be used to determine the type of Hopf bifurcation (supercritical or subcritical). A schematic representation of the Hopf bifurcation normal form is shown in Fig. 9.

It is important to note that this is a relatively simple example. However, even if we had all possible quadratic and cubic terms in the nonlinear vector on the righthand side of Eq. (310), then these could still be projected onto the  $z^*$  basis in the form of  $N_z z^*$  (with different entries in  $N_z$  from those shown above). In this way, *all* the terms in such as system can be reduced to a normal form controlled by just a single parameter.

# 4.10 Summary

In this Section, the normal form transformations of nonlinear systems have been considered. Several important topics have been introduced, including the use of near-identity transformations, derivation of the homological equations using Lie series, Hamiltonian normal form, frequency detuning, choice of free functions, backbone curves and bifurcations. Although the analysis has been developed for multi-degree-of-freedom systems, all the examples in this Section have been single-degree-of-freedom. The complexity of applying normal form techniques to nonlinear oscillators with more than one degree-of-freedom is significant, as will be seen in the next Section. In particular it is the topic of backbone curves that will be of most interest as we consider coupled nonlinear oscillators next.

# 5 Normal form transformations for coupled nonlinear oscillators

In this final Section we will consider how normal form transformations can be applied to systems of coupled nonlinear oscillators. All the analysis developed in Section 4 can be applied to systems with more than a single-degree-of-freedom. However, one important addition is needed. In the base solutions for Example 1, (see Eq. (13)) we included a phase lag term, which we have (deliberately) neglected throughout the discussion of the nonlinear oscillators so far. Although this might be acceptable in many cases for single-degree-of-freedom oscillators, it needs to be included in the case of coupled (e.g. multi-degree-of-freedom) oscillators. Therefore, in this Section we define the assumed base solutions for the *i*<sup>th</sup> degree-of-freedom as

$$u_{pi} = \frac{U}{2} \mathbf{e}^{\mathbf{i}(\omega_{ri}t - \phi_i)}, \quad \text{and} \quad u_{mi} = \frac{U}{2} \mathbf{e}^{-\mathbf{i}(\omega_{ri}t - \phi_i)}.$$
(321)

where  $\omega_{ri}$  and  $\phi_i$  are the response frequency and phase lag for the *i*th degree-of-freedom respectively.

<sup>&</sup>lt;sup>64</sup>We do not go into further detail here, but this will correspond to a subcritical rather than supercritical type of Hopf bifurcation. Much more in depth treatments can be found in many texts on nonlinear dynamics, see for example the discussions in [77, 55, 52, 162].

It should be noted that the assumptions regarding these solution types will capture only periodic solutions with a frequency  $\omega_{ri}$ . Coupled nonlinear systems typically exhibit far more complex behaviours, such as quasiperiodic and chaotic motions — see for example [2, 59, 40, 53] and references therein. The type of analysis presented here will only capture the most basic underlying dynamical behaviour. We now consider an example.

#### 5.0.1 Example 20: Two degree-of-freedom undamped system with cubic spring nonlinearities

We consider the two-degree-of-freedom lumped mass system<sup>65</sup>, as shown in Fig. 10 (a), in which the spring forces contain linear and cubic terms with coefficients k, $k_2$  and  $\kappa$ ,  $\kappa_2$  respectively for masses, m. The equations of motion may be written as

$$\begin{bmatrix} m & 0 \\ 0 & m \end{bmatrix} \ddot{\mathbf{x}} + \begin{bmatrix} k+k_2 & -k_2 \\ -k_2 & k+k_2 \end{bmatrix} \mathbf{x} + \begin{bmatrix} \kappa x_1^3 - \kappa_2 (x_2 - x_1)^3 \\ \kappa x_2^3 + \kappa_2 (x_2 - x_1)^3 \end{bmatrix} = 0, \quad \rightsquigarrow \quad M\ddot{\mathbf{x}} + K\mathbf{x} + \mathcal{N}(\mathbf{x}) = 0, \quad (322)$$

where  $\mathbf{x} = [x_1 \ x_2]^T$  in this case. The linearised version of this system, where  $\kappa = \kappa_2 = 0$ , results in the modal transform

$$\mathbf{x} = \Phi \mathbf{q} \quad \text{where} \quad \Phi = \begin{bmatrix} 1 & 1 \\ 1 & -1 \end{bmatrix}, \quad \rightsquigarrow \quad \ddot{\mathbf{q}} + \begin{bmatrix} \omega_{n1}^2 & 0 \\ 0 & \omega_{n2}^2 \end{bmatrix} \mathbf{q} + \frac{\kappa}{m} \begin{bmatrix} q_1^3 + 3q_1q_2^2 \\ \gamma q_2^3 + 3q_1^2q_2 \end{bmatrix} = 0, \tag{323}$$

where  $\gamma = 1 + 8\kappa_2/\kappa$ ,  $\omega_{n1}^2 = \frac{k}{m}$  and  $\omega_{n2}^2 = \frac{k+2k_2}{m}$  are the linear natural frequencies. Now we define<sup>66</sup>

$$\Lambda = \begin{bmatrix} \omega_{n1}^2 & 0\\ 0 & \omega_{n2}^2 \end{bmatrix}, \text{ and } \varepsilon \hat{\mathbf{n}}(\mathbf{q}) = \varepsilon \frac{\hat{\kappa}}{m} \begin{bmatrix} q_1^3 + 3q_1q_2^2\\ \gamma q_2^3 + 3q_1^2q_2 \end{bmatrix}$$
(324)

where  $\kappa = \varepsilon \hat{\kappa}$  has been used<sup>67</sup>. Now we are transforming from  $q_i$  to  $u_i$ , using  $T_{SO}$  from Eq. (116). Note that  $u_i = u_{ip} + u_{im}$  which can also be expressed as the vector relationship  $\mathbf{u} = \mathbf{u}_p + \mathbf{u}_m$ . Therefore we obtain

<sup>&</sup>lt;sup>65</sup>This is example is taken from Chapter 5 of [128] and [68], and further details can be found there. See also the nonlinear normal mode analysis given by Vakakis et al. [175].

<sup>&</sup>lt;sup>66</sup>As before  $\mathbf{n}_2 = 0, ...$  etc. so we set  $\mathbf{n}_1 = \mathbf{n}$ , and correspondingly,  $\mathbf{h}_1 = \mathbf{h}$ ,  $\mathbf{b}_1 = \mathbf{b}$  and  $\hat{\mathbf{n}}_{u1} = \hat{\mathbf{n}}_u$  throughout this example.

<sup>&</sup>lt;sup>67</sup>Note that the specific structure in the nonlinear vector,  $\hat{\mathbf{n}}(\mathbf{q})$ , of Eq. (324) is of a relatively simple form because of the structure of the original nonlinear vector  $\mathscr{N}(\mathbf{x})$  in this example. In general for problems of this type each row would have terms of  $q_1^3$ ,  $q_1q_2^2$ ,  $q_1^2q_2$  and  $q_2^3$  such that there would be eight coefficients in total in  $\hat{\mathbf{n}}(\mathbf{q})$ .





where  $r = \omega_{r2}/\omega_{r1}$ , and  $\Lambda_d$  is the detuned frequency matrix obtained from  $\dot{u} = \Lambda_d u$  using Eq. (321). Next the detuned Lie derivative is computed by substituting  $\Lambda = \Lambda_d$  into Eq. (127) to give the coefficient matrix  $\Lambda^*_d$ . This

# is repeated in Eq. (153) to give the following

	3		9		
$\Lambda^{*}_{d} = \mathbf{i}\omega_{r1}\mathbf{I}_{20}$ $1$ $-1$ $-3$ $1+2$ $1$ $1-2$ $-1+2$ $-1$ $-1-2$ $2+r$ $2-r$ $r$ $-r$ $-2+$ $-2-$ $3r$ $r$	1	$,  \rightsquigarrow  (\Lambda^*{}_d)^2 = -\omega_{r1}^2 \mathbf{I}_{20}$	1		
	-1		1		
	-3		9		
	1 + 2r		$(1+2r)^2$		
	1		1		
	1 - 2r		$(1-2r)^2$		
	-1 + 2r		$(-1+2r)^2$		
	-1		1		
	-1 - 2r		$(-1-2r)^2$		(207)
	2 + r		$(2+r)^2$	,	(327)
	2-r		$(2-r)^2$		
	r		$r^2$		
	-r		$r^2$		
	-2 + r		$(-2+r)^2$		
	-2 - r		$(-2-r)^2$		
	3 <i>r</i>		$9r^{2}$		
	r		$r^2$		
	-r		$r^2$		
	-3r		9 <i>r</i> <sup>2</sup>		

where  $I_{20}$  is a  $20 \times 20$  identity matrix. These relationships are substituted into the detuned homological equation, Eq. (261), to give

$$-\mathbf{b}\Lambda^{*2}_{\ d}\mathbf{u}^{*} - \mathbf{Y}\mathbf{b}\mathbf{u}^{*} = \omega_{r1}^{2} \begin{bmatrix} b_{1}(8) & b_{21}(9-r^{2}) \\ b_{2}(0) & b_{22}(1-r^{2}) \\ b_{3}(0) & b_{23}(1-r^{2}) \\ b_{4}(8) & b_{24}(9-r^{2}) \\ b_{5}(4(r^{2}+r)) & b_{25}(1-4r+3r^{2}) \\ b_{6}(0) & b_{26}(1-r^{2}) \\ b_{7}(4(r^{2}-r)) & b_{27}(1-4r+3r^{2}) \\ b_{9}(0) & b_{29}(1-r^{2}) \\ b_{10}(4(r^{2}+r)) & b_{30}(1+4r+3r^{2}) \\ b_{11}(3+4r+r^{2}) & b_{31}(4(1+r)) \\ b_{12}(3-4r+r^{2}) & b_{32}(4(1-r)) \\ b_{13}(r^{2}-1) & b_{33}(0) \\ b_{14}(r^{2}-1) & b_{35}(4(1-r)) \\ b_{15}(r^{2}-2r) & b_{35}(4(1-r)) \\ b_{16}(3-4r+r^{2}) & b_{36}(4(1+r)) \\ b_{17}(r^{2}-1) & b_{37}(8r^{2}) \\ b_{18}(r^{2}-1) & b_{39}(0) \\ b_{19}(r^{2}-1) & b_{39}(0) \\ b_{19}(r^{2}-1) & b_{39}(0) \\ b_{20}(r^{2}-1) & b_{40}(8r^{2}) \end{bmatrix}$$

$$= \mathbf{n}^{*}\mathbf{u}^{*} - \mathbf{n}_{\mathbf{u}}^{*}\mathbf{u}^{*}, \quad (328)$$

where **b** and  $\mathbf{n}_{\mathbf{u}}^*$  are coefficient matrices of the same dimension as  $\mathbf{n}^*$  defined in Eq. (325). As with previous examples there is now a choice of free functions for  $b_i$  and i = 1, 2, 3...20. Notice that some terms have a zero associated with the  $b_i$  (as we saw previously), and these are the *unconditional resonances* (as defined in Section 4.1.2). In addition, now that we have multiple degrees-of-freedom, and therefore more than one response frequency, there are other  $b_i$  values which could have a zero term associated with them depending on the frequency ratio value, *r*. These are the *conditional resonances*, and as we discussed in Section 4.1.2 are also called *internal resonances*.

Notice also that the simplest way to satisfy Eq. (328) is to set  $b_i = n_{ui}^* = 0$  whenever  $n_i^* = 0$ . From Eq. (325) it can be seen that this applies to i = 11, 12, 13...28, 29, 30. After setting the associated  $b_i = 0$  for i = 11, 12, 13...28, 29, 30 we are left with unconditionally resonance terms for  $b_2, b_3, b_6, b_9, b_{33}, b_{34}, b_{38}, b_{39}$  and conditionally resonant terms for  $b_7, b_8, b_{32}, b_{35}$  based on the condition r = 1. This is called a one-to-one resonant interaction, because it occurs when  $\omega_{r1} = \omega_{r2}$ .

Therefore to satisfy Eq. (328) for both the conditional and unconditional resonances we set the coefficients of  $\mathbf{n}_{\mathbf{u}}^*$  to be

where  $\delta_{1r}$  is the Kronecker delta function defined such that  $\delta_{1r} = 1$  (i.e. when r = 1) and  $\delta_{1r} = 0$  for all other

values (i.e. when  $r \neq 1$ ). Substituting Eq. (329) into Eq. (146) gives the  $\varepsilon^1$  normal form as

$$\ddot{\mathbf{u}} + \begin{bmatrix} \omega_{n1}^2 & 0\\ 0 & \omega_{n2}^2 \end{bmatrix} \mathbf{u} + \frac{3\kappa}{m} \begin{bmatrix} u_{p1}^2 u_{m1} + u_{p1} u_{m1}^2 + 2u_{p1} u_{p2} u_{m2} + 2u_{m1} u_{p2} u_{m2} + \delta_{1r} (u_{p1} u_{m2}^2 + u_{m1} u_{p2}^2) \\ 2u_{p2} u_{p1} u_{m1} + 2u_{m2} u_{p1} u_{m1} + \gamma u_{p2}^2 u_{m2} + \gamma u_{p2} u_{m2}^2 + \delta_{1r} (u_{p1}^2 u_{m2} + u_{m1}^2 u_{p2}) \\ \end{bmatrix} = 0,$$
(330)

where  $\mathbf{u} = [u_1 \ u_2]^T$ . Furthermore, using Eq. (116), we find that the modal displacements are given by

$$\begin{bmatrix} q_1 \\ q_2 \end{bmatrix} = \begin{bmatrix} u_1 \\ u_2 \end{bmatrix} + \frac{\kappa}{m\omega_{r1}^2} \begin{bmatrix} \frac{1}{8}(u_{p1}^3 + u_{m1}^3) + \frac{3}{4(r^2 + r)}(u_{p1}u_{p2}^2 + u_{m1}u_{m2}^2) + \frac{3}{4(r^2 - r)}(u_{p1}u_{m2}^2 + u_{m1}u_{p2}^2) \\ \frac{\gamma}{8r^2}(u_{m2}^3 + u_{p2}^3) + \frac{3}{4(1 + r)}(u_{m2}u_{m1}^2 + u_{p2}u_{p1}^2) + \frac{3}{4(1 - r)}(u_{p1}^2 u_{m2} + u_{m1}^2 u_{p2}) \end{bmatrix},$$
(331)

when  $r \neq 1$ , and

$$\begin{bmatrix} q_1 \\ q_2 \end{bmatrix} = \begin{bmatrix} u_1 \\ u_2 \end{bmatrix} + \frac{\kappa}{m\omega_{r1}^2} \begin{bmatrix} \frac{1}{8}(u_{p1}^3 + u_{m1}^3) + \frac{3}{4(r^2 + r)}(u_{p1}u_{p2}^2 + u_{m1}u_{m2}^2) \\ \frac{\gamma}{8r^2}(u_{m2}^3 + u_{p2}^3) + \frac{3}{4(1 + r)}(u_{m2}u_{m1}^2 + u_{p2}u_{p1}^2) \end{bmatrix},$$
(332)

for the case when r = 1.

Expressions for the backbone curves can be found by taking the base solutions from Eq. (321) as the assumed solutions and substituting them into Eq. (330). For example, carrying this out for the first line of Eq. (330) gives

$$\left( \omega_{n1}^{2} - \omega_{r1}^{2} \right) U_{1} \left( e^{\mathbf{i}(\omega_{r1}t - \phi_{1})} + e^{-\mathbf{i}(\omega_{r1}t - \phi_{1})} \right) + \frac{3\kappa}{4m} U_{1} \left[ (U_{1}^{2} + 2U_{2}^{2}) \left( e^{\mathbf{i}(\omega_{r1}t - \phi_{1})} + e^{-\mathbf{i}(\omega_{r1}t - \phi_{1})} \right) + \delta_{1r} U_{2}^{2} \left( e^{-\mathbf{i}(\omega_{r1}t - \phi_{1})} e^{-\mathbf{i}2(\phi_{1} - \phi_{2})} + e^{\mathbf{i}(\omega_{r1}t - \phi_{1})} e^{\mathbf{i}2(\phi_{1} - \phi_{2})} \right) \right] = 0.$$

$$(333)$$

Using the same approach for the second line of Eq. (330) and then balancing the  $e^{i(\omega_{r_1}t-\phi_1)}$  terms gives<sup>68</sup>

$$\left\{\omega_{n1}^{2} - \omega_{r1}^{2} + \frac{3\kappa}{4m} \left[U_{1}^{2} + U_{2}^{2} \left(2 + \delta_{1r} e^{i2(\phi_{1} - \phi_{2})}\right)\right]\right\} U_{1} = 0,$$

$$\left\{\omega_{n2}^{2} - \omega_{r2}^{2} + \frac{3\kappa}{4m} \left[\gamma U_{2}^{2} + U_{1}^{2} \left(2 + \delta_{1r} e^{-i2(\phi_{1} - \phi_{2})}\right)\right]\right\} U_{2} = 0.$$
(334)

There are several cases of backbone curves, depending on the parameters in Eq. (334). The first case is for *single mode backbone curves* where we set either  $U_1 = 0$  or  $U_2 = 0$  to give

S1: when 
$$U_2 = 0$$
,  $\omega_{r1}^2 = \omega_{n1}^2 + \frac{3\kappa}{4m}U_1^2$ ,  
S2: when  $U_1 = 0$ ,  $\omega_{r2}^2 = \omega_{n2}^2 + \frac{3\kappa\gamma}{4m}U_2^2$ . (335)

These expression relate to the non-resonant case,  $r \neq 1$  (therefore  $\delta_{1r} = 0$ ), when there is no *phase coupling* between the modal coordinates  $u_1$  and  $u_2$  (and then by definition  $q_1$  and  $q_2$ ) via the  $e^{-i2(\phi_1 - \phi_2)}$  term in Eq. (334).

To consider the resonant case, when r = 1 (and therefore  $\delta_{1r} = 1$ ), we first introduce a general frequency response parameter  $\Omega$  and then rewrite Eq. (334) such that we consider the response of both equations at a specific  $\Omega$  value which gives

$$\Omega^{2} = \omega_{n1}^{2} + \frac{3\kappa}{4m} \left[ U_{1}^{2} + U_{2}^{2}(2+p) \right] = \omega_{n2}^{2} + \frac{3\kappa}{4m} \left[ \gamma U_{2}^{2} + U_{1}^{2}(2+p) \right], \quad \text{where} \quad p = e^{i2(\phi_{1} - \phi_{2})}. \tag{336}$$

In order to maintain real solutions we require that  $p = \pm 1$  such that<sup>69</sup>

p = +1 corresponds to  $|\phi_1 - \phi_2| = 0, \pm n\pi$ , for n = 1, 2, 3... This corresponds to the *in-unison case*, with either in-phase (0) or out-of-phase ( $\pm n\pi$ ) cases,

 $\square$ 

<sup>&</sup>lt;sup>68</sup>As in previous examples, balancing the  $e^{-i(\omega_{r1}t-\phi_1)}$  terms gives equivalent expressions.

<sup>&</sup>lt;sup>69</sup>This analysis was first developed in [114], where these two solutions were referred to as the *normal mode* and *elliptic mode* due to their shapes in the configuration space.

p = -1 corresponds to  $|\phi_1 - \phi_2| = \pm n\pi/2$ , for n = 1, 2, 3... This is the *out-of-unison case*, with  $\pm n\pi/2$  phase differences in the response behaviour.

Here we consider just the p = +1 case. Details of the out-of-unison case can be found for example in [114, 68]. Setting p = +1 and n = 1 yields two backbone curves, labelled  $D12_i^+$  and  $D12_i^-$ , with the phase differences

$$D12_i^+: \quad |\phi_1 - \phi_2| = 0, \qquad D12_i^-: \quad |\phi_1 - \phi_2| = \pi.$$
(337)

Note that the notation is *D* for double (i.e. two mode) interaction, followed by the modes that interact (1 and 2 in this case), the subscript *i* means in-unison, and plus (in-phase) or minus (out-of-phase)<sup>70</sup> Substituting p = +1 in Eq. (336) leads to

$$D12_{i}^{\pm}: \qquad U_{1}^{2} = \left(1 - 4\frac{\kappa_{2}}{\kappa}\right)U_{2}^{2} - \frac{2m}{3\kappa}\left(\omega_{n2}^{2} - \omega_{n1}^{2}\right) \qquad \text{and} \qquad \Omega^{2} = \frac{3\omega_{n1}^{2} - \omega_{n2}^{2}}{2} + \frac{3(\kappa - \kappa_{2})}{m}U_{2}^{2}. \tag{338}$$

From the first expression of Eq. (338) it can be seen that to ensure real solutions the following conditions can be imposed

$$U_2^2 > \frac{2m}{3(\kappa - 4\kappa_2)} (\omega_{n2}^2 - \omega_{n1}^2), \text{ and } \kappa \ge 4\kappa_2.$$
 (339)

Numerically computed results are shown in Fig. 10 (b), (c) and (d). The numerical results are for the case where the central spring connecting the two masses is much less stiff than the springs connecting the masses to the rigid walls at each end, e.g.  $k_2 \ll k$  and  $\kappa_2 \ll \kappa$ . We call this type of system *weakly coupled*, and for very small amplitudes, the two masses both oscillate at a frequency very close to one, on the *S*1 and *S*2 curves. Then as the  $U_i$  or  $X_i$  amplitudes increase, it can be seen that the *S*2 backbone curve has a bifurcation at which point the stable  $D12_i^{\pm}$  curves coalesce with *S*2. The bifurcation is caused by the underlying periodic orbit undergoing a secondary-Hopf type bifurcation. This manifests itself in terms of the backbone curve, as a supercritical pitchfork bifurcation event. After the bifurcation point, the *S*2 becomes unstable, and the stable solutions are given by the  $D12_i^{\pm}$  backbone curves. The type of modal coordinate motion that occurs for each backbone curve is shown in the inset diagrams in Fig. 10 (b). Further details and discussion of this example is given in [128].

Note also that these type of backbone curves can be interpreted as the underlying "skeleton" which govern some of the key characteristics of the forced-damped responses. For example the curvature of the forced-damped resonance curve, as was shown in Fig. 2 (c). In this example, the structure of the backbone curves is considerably more complex than those shown in Fig. 2 (c), but the same principle holds true. A detailed discussion of the forced-damped responses relevant to this example is given in [67] — see also [21, 68, 73].

## 5.1 Summary

The example considered in this Section demonstrates how normal form methods can be used to help understand the behaviour of coupled nonlinear oscillators. We have limited the examples in this Section to just one, because (i) we are now going beyond the introductory remit of this paper, and (ii) there are multiple published examples for the interested reader to find in the literature. For example, normal form analysis of structural elements such as beams, plates, shells and cables can be found in references [124, 194, 171, 68, 128, 44, 33, 50], to highlight just a few. Perhaps the most important features highlighted in this Section is the role of the conditional resonances, and how they can be interpreted using backbone curves.

To demonstrate the approach with as much simplicity as possible, this example has not included either damping or forcing, which are typically needed for real applications. That said, forced-damped cases have also been treated using normal form methods — see for example [128, 15] and references therein. An alternative is to capture the forced-damped behaviour using numerical continuation packages, such as COCO [153], MATCONT [35], MANLAB [56], and AUTO [36], and the conservative backbone curves are found using normal form techniques — for example as shown in Fig. 2<sup>71</sup>. Of course backbone curves are independent of normal forms, and can be found using multiple other methods as well, see for example [128, 41, 42] and references therein. Further potential developments will be discussed in Section 6.3.

<sup>&</sup>lt;sup>70</sup>In [128] and [68] these curves are referred to as  $S3^{\pm}$  for the p = 1 case and  $S4^{\pm}$  for the p = -1 case.

<sup>&</sup>lt;sup>71</sup> Interesting recent studies on the stability of the forced-damped curve and the relationship to the backbone curve have been undertaken by Cenedese and Haller [27, 28].



**Fig. 10:** The two-degree-of-freedom nonlinear oscillator considered in Example 20, Section 5.0.1. Showing (a) the schematic diagram of the two-degree-of-freedom nonlinear oscillator, (b) plot showing the backbone curves in the projection of the displacement amplitude  $X_1$  against  $\Omega$ , where ( $X_i = ||x_i||$  for i = 1, 2). This shows the in-phase backbone, *S*1, the out-of-phase backbone, *S*2, and the conditionally resonant double mode (also called "mixed mode") backbones  $D12_i^{\pm}$ . The inset diagrams show the behaviour in modal coordinates  $q_1$ ,  $q_2$  (noting that  $q_i = u_i$  to order  $\varepsilon^0$ ). Green dots represent the points where the double backbone curves  $D12_i^{\pm}$  bifurcate from the single mode curve *S*2. Black dots represent the birth of a nonlinear normal mode family of periodic solutions via a Hamiltonian Hopf bifurcation. (c) The backbone curves in the projection of  $U_1$  against  $\Omega$ , where red dots represent the start of a non-oscillatory curve e.g. when either  $U_1 = 0$  or  $U_2 = 0$ . (d) Projection of  $U_2$  against  $\Omega$ , where the dash red line represents an unstable part of the backbone curve. Parameter values are m = 1 kg, k = 0.98 N/m,  $k_2 = 0.0202$  N/m,  $\kappa = 0.2$  N/m<sup>3</sup> and  $\kappa_2 = 0.02$  N/m<sup>3</sup>. Note that  $\kappa > 4\kappa_2$  in this example, and that parameters are chosen to show the phenomena rather than relate to a physical example.

# 6 Conclusions and future directions for research

## 6.1 Summary and conclusions

In this paper an introduction to using normal form transformations for linear and nonlinear structural dynamics has been given. After some motivating examples, the paper investigated linear single-degree-of-freedom systems, which is a topic that is fundamental to structural dynamics. In that discussion, we outlined the importance of exponential base solutions for both damped and undamped oscillators. However, it was also highlighted how simplifying the matrix structure of the oscillator model could significantly simplify the solution procedure. Coordinate transformations were considered via Jordan normal form and the method of reduction of order.

Simplifying matrix structures leads naturally to coordinate transformations, and structural dynamists are generally very familiar with this concept via the idea of modal analysis. This was the starting point for the discussion of multi-degree-of-freedom linear systems. Modal transformations were demonstrated both for differential equations written in state-space form and classical coupled ordinary differential equations in second-order form. This type of modal coordinate transformation was then used as a first step when analysing nonlinear oscillators.

We began the discussion of nonlinear systems, by highlighting some of the key issues. Most importantly the concept of nonlinear resonance. Following that, the idea of near-identity transformations was described in detail

for both first-order and second-order sets of equations. This led to expressions for the homological equation, the solution of which is the central operation of a normal form method. Another important detail considered at this point was to show the equivalence of the real and complex normal forms using a Duffing oscillator example.

Normal form transformations can be expressed in terms of Lie series, and to give an insight into this connection we described a derivation of the homological equations using Lie series. This required prolongation of the near-identity transformation, and gives additional insight into the structures that underly the complex and real normal form transformations. This was followed by a description of the Hamiltonian normal form. This type of normal form is not used very often in structural dynamics, and is more commonly used in maths and physics. However, for completeness, it has been included to show the relationship with both the complex and real normal form variants described previously.

In order to describe amplitude-frequency nonlinear effects, the concept of frequency detuning was introduced. Following this, the choice of free functions in the homological equation was discussed, and an example of a minimal normal form included. Next the topic of backbone curves was described, although as pointed out at the time, this is an independent concept that can be used for nonlinear oscillators without normal form methods.

Finally, we considered the case of a coupled nonlinear oscillator system. This final example showed the concept of conditional resonances, a phenomena that is very naturally modelled via the homological equation. It also demonstrated how the real normal form method could be used to find approximate analytical expressions for backbone curves. The conditional (nonlinear) resonance was represented as a bifurcation of one of the backbone curves into a double backbone curve solution.

In Section 1.1.4 we outlined five useful properties of normal form transformations for structural dynamicists. We now revisit each of these in turn:

- I. Obtaining approximate solutions for x: This was demonstrated for most of the examples throughout the paper. For nonlinear systems certain conditions are required for this to be possible; specifically the eigenvalues of the underlying linear problem need to be semisimple see [120]) for details. For multi-degree-of-freedom nonlinear systems, the algebraic complexity of trying to obtain an approximate solutions for x quickly becomes prohibitive.
- **II.** Obtain information on the nonlinear resonances: This is one of the most important aspects, particularly for nonlinear multi-degree-of-freedom oscillators, as we saw in Example 20. It will also be discussed in the next Section.
- **III.** Obtain frequency amplitude relationships known as backbone curves: Again this is particularly relevant for nonlinear multi-degree-of-freedom oscillators (e.g. Example 20) and will also be discussed in the next Section.
- **IV.** *Simplify models of important dynamic phenomena, such as bifurcations.* An example was shown for the Hopf bifurcation in Section 4.9. This is one of the most important dynamic phenomena for vibration problems, and using normal form transformations it can be reduced to a model with just a single parameter that determines the key behaviour.
- V. Normal form transformations used to achieve model-order reduction or system identification. In fact, (and related to point IV above) normal forms can be used not just to reduce the number of states, but also the number of parameters. This comes from the ethos of normal forms, which is to systematically produce the simplest model that captures the important dynamics, both in terms of the number of states and the number of parameters. Model-order reduction or system identification were not discussed in detail in this paper, but they are mentioned in Section 6.3 where we will discuss future research directions.

Other concluding points regarding the normal form techniques introduced in this paper are:

- a) Coordinate transformations can simplify, but not without work! Both linear and nonlinear systems can be simplified using coordinate transformations, as demonstrated in this paper. However, this is not without work. The algebraic intensity, and work required in some of the methods presented here is a significant limitation of both their application and popularity.
- b) Formats and styles are important. As has been shown in this paper, normal form transformations are not unique, and are reliant on many choices. This gives a lot of flexibility in how they can be tailored to suit a particular application. However, it can also be confusing when there are so many different variants, and small variations in the outcomes. That said, the final results should be equivalent, regardless of which style was used, as was shown for example in Section 4.2.5 with the equivalence between the complex and real normal form. For linear systems, there is less to be chosen, but the scaling of eigenvectors is one example where choices made in the method can affect the outcome.

- c) The homological equation is central to the method. For nonlinear systems, this offers the key advantage of the normal form technique over other methods. By developing a homological equation, that is linear to solve, and also identifies the conditional and unconditional resonances, the normal form approach can give insight into the likely behaviour of the nonlinear system. This insight is explicit, in terms of structures like backbone curves and their bifurcations.
- d) Normal forms transformations can be used as part of a reduced order modelling process. In general, a normal form transformation will result in a system that has the same number of degrees-of-freedom as the original system. The examples in this paper have all been either single, two or three degree-of-freedom. As the number of degrees-of-freedom increases beyond this, it makes sense to combine the normal form method with a reduction in the order of the system, in order to capture just the essential dynamics. There have been many investigations into combining normal form and order reduction methods (mentioned below), and this is a rapidly expanding area of future research<sup>72</sup>.

# 6.2 Comparison with other methods; some advantages and disadvantages

In the context of structural dynamics, the method of normal forms has been compared to other techniques that are used to carry out similar tasks, for example to compute nonlinear normal modes or backbone curves [66, 41, 42, 180]. As was described in Section 4.1.3 there are subtle but important distinctions between normal form methods and perturbation techniques, and so we will outline some of the potential advantages of using normal form techniques compared to these and other methods. Choosing a particular technique to use will depend on multiple factors, but the typical advantages of normal form methods are:

- There is no explicit requirement of "smallness" of either the nonlinear terms or amplitude of response<sup>73</sup> in the same way that a perturbation method inherently has — see for example the study carried out by [104]. Observations by [66, 41, 42] also show examples of how normal form approximations to backbone curves can be extended to relatively large amplitudes without significant loss if accuracy, in comparison to harmonic balance and the method of multiple scales.
- 2. During the normal form procedure, no secular terms are generated, and as a result there are no "solvability conditions" (to use the terminology of [84]) that occur, for example, in the method of multiple scales<sup>74</sup>. Advocates of other methods point out that secular terms can be used to infer that resonances are present in the system being considered. However, in a normal form approach, the resonance's are revealed explicitly in the homological equation. Furthermore, the homological equation also gives information on more subtle features of the resonance, such as whether they are conditional or unconditional, without needing to carry out additional work.
- **3.** The resonant terms can be dealt with in a systematic way via the homological equation, whereas in perturbation methods this process is ad-hoc and (depending on the exact details) can require additional terms to "fix" the solutions depending on the exact resonant conditions, as mentioned above.
- 4. For reduced-order models of high-dimensional systems, normal form techniques provide a very strong modelling framework, particularly in terms of *intrusive* methods see for example the comparison made by [180]. That said, normal forms is not the only approach to this problem. For example, harmonic balance methods have also been developed in combination with reduction of order techniques to create reduced-order models see for example [176].

In terms of disadvantages of the normal form technique, Kodama and Mikhailov [92] pointed out some issues that occur when normal form transformations are applied to certain partial differential equations. This subject

<sup>&</sup>lt;sup>72</sup>It should be noted that care needs to be taken with these types of reduction methods, as terms from neglected modes can affect the normal form of the reduced model.

<sup>&</sup>lt;sup>73</sup>The exception is where it is deliberately introduced, as discussed in Section 4.1.3.

<sup>&</sup>lt;sup>74</sup>These issues have been known about for some time. For example, they are described in [141], but first became apparent in a structural dynamics context in the work of [147], where the authors tried to apply the method of multiple scales to systems with internal resonances. In their conclusions the authors stated: *"This indicates a yet unclear mathematical drawback in the specific application of the method of multiple scales to this two time-scale-dependent problem."* The underlying reason for these issues are explained in detail in [84] (which references the theoretical work of [91]). In single-degree-of-freedom systems, the solvability conditions are (typically) benign, and do not restrict choices too much (see [98] for an example of how this restriction can be effectively removed by adding an additional time scale). However, as the system becomes more complex, particularly for systems with more degrees-of-freedom, solvability conditions start to become restrictive, and to get the method of multiple scales to work (without adding additional time scales) the solutions are "fixed" by adjusting both free functions and initial conditions — see the two-degree-of-freedom examples shown in [125] (Chapter 10 page 267) and [41] where it is observed that the initial conditions cannot be set to zero, and instead they need to be used to satisfy the solvability conditions.

has been subsequently further explored by other authors [71, 93, 191]. In addition, normal forms techniques are often regarded as being algebraically intense, and some authors have developed symbolic computation methods (particularly using Maple) in order to try and mitigate this effect [10, 190, 193, 122].

# 6.3 Future directions for research

In terms of future research, it seems likely that normal form methods will continue to be an important class of methods for studying the behaviour of nonlinear oscillator systems. From the authors perspective, there are several key areas that would be of interest for future research development:

- 1. Investigation of nonlinear resonances. This longstanding topic continues to be of significant interest, and there are still several aspects that can be investigated. From a structural dynamics perspective, the understanding of the phenomena has been progressed in several different ways. Firstly from a geometric perspective, and with particular application to the geometric definition of nonlinear normal modes (NNMs) as invariant manifolds — first put forward by Shaw and Pierre [155] — and formalised using normal forms by Touzé [168, 169] (see also the overview given by Volvert and Kerschen [182]). From a normal form perspective this study of resonance in coupled oscillators is mainly being advanced via the analysis of backbone curves - see for example [21, 68, 15, 50, 53]. However, other methods are also relevant in this area. Particularly following the work of Haller and Ponsioen [59] in defining spectral submanifolds, which are now able to capture some of the more subtle and interesting [139] phenomena in this domain. In particular, the spectral submanifold technique can be applied in a normal form style (or as an alternative in a graph style), and an interesting comparison of the spectral submanifold method and other normal form techniques has been carried out by Breunung and Haller [15]. Also related to this topic is the relationship between NNMs (or backbone curves) and the forced-damped responses of coupled oscillator systems [170, 169, 67, 70, 15]. Recent developments have included analysis of rotor stator contact [154], analyses using a Melnikov function approach [28, 27], parametric antiresonance [86], and vibration of nancomposite shells [118] to name just a few. This will continue to be a key area of future research.
- 2. Identification methods from experimental data. This is also a longstanding topic of interest, that has not been approached using normal forms until relatively recently see [50, 33]. In contrast there are a selection of other approaches that are typically used for this techniques, see for example [135, 39, 69, 148, 33, 150] and references therein. Other recent developments include analysis of wheel-set dynamics [187] and power grids [94]. It is likely that this research topic will grow as an area of interest. In particular, there is considerable potential cross-over between using normal form transformations to create reduced-order models for engineering structures, and identification methods. Specifically the reduced-order models of structural elements, such as beams, cables plates and shells, lead to much simpler models with a number of key parameters that can potentially be identified. See for example [169, 179, 173, 133].
- 3. Nonlinear model reduction. Reducing large scale systems (for example from finite element models) down to smaller models that capture the key dynamical behaviour is another area of great importance in the structural dynamics field see for example [172, 19, 131, 20, 62, 138, 177]. As mentioned above, normal form methods have been used extensively in combination with methods to reduce the order of the system see for example [110, 180, 156, 173], and it is likely that this topic will grow considerably in the future. In particular, the connection of invariant manifold computation techniques (described in point 1.) has recently been developed to work for structures with complex geometries, such as those represented in finite element models [180, 181, 79]. A comparison has also been recently made between reduced-order normal form methods, and other similar methods, such as modal derivatives [180]. This is an important development for the application of these methods to more complex engineering problems.
- 4. Hamiltonian normal form. This is mostly used in physics, astronomy and celestial mechanics, as described for example by [63, 49] and references therein. That said, there are also some applications in maths and physics that relate to structural dynamics, such as resonances in chains of oscillators [178], triangular connected spring-mass systems [87], resonance-assisted tunnelling [46], and this is an open area for structural dynamists in the future.

# 7 Author's Contributions

The author wrote the paper, including generating the simulations for all figures.

# 8 Acknowledgements

The author would like to thank all his collaborators and coworkers on this research topic. In particular Simon Neild, Andrea Cammarano, Tom Hill, Alan Champneys, Zhengfang Xin, Alex Shaw, Alexander Elliott, Xuanang Liu, Neil Sims and Ayman Nasir. Thanks also to Ayman Nasir for proof reading the manuscript, and to Cyril Touzé and Yair Zarmi for useful discussions regarding the manuscript. Thanks also to the reviewers of this paper, who's comments helped to improve the final published version. The author would also like to thank the Engineering and Physical Science Research Council for both Phd studentship and grant funding that supported this work, specifically grants EP/K003836/1 & EP/K003836/2.

# References

- [1] S. Adhikari. *Structural dynamic analysis with generalized damping models: analysis.* John Wiley & Sons, 2013.
- [2] M. Amabili and C. Touzé. Reduced-order models for nonlinear vibrations of fluid-filled circular cylindrical shells: Comparison of POD and asymptotic nonlinear normal modes methods. *Journal of fluids and structures*, 23(6):885–903, 2007.
- [3] P. Apiwattanalunggarn, S. W. Shaw, and C. Pierre. Component mode synthesis using nonlinear normal modes. *Nonlinear Dynamics*, 41(1-3):17–46, 2005.
- [4] V. I. Arnold. Geometrical Methods in the Theory of Ordinary Differential Equations. Springer, 1988.
- [5] N. N. Balaji and M. R. W. Brake. A quasi-static non-linear modal analysis procedure extending Rayleigh quotient stationarity for non-conservative dynamical systems. *Computers & Structures*, 230:106184, 2020.
- [6] D. Bambusi. An introduction to Birkhoff normal form, 2014. Unpublished.
- [7] J. Barrow-Green. *Poincaré and the three body problem*. Number 11. American Mathematical Soc., 1997.
- [8] S. Bellizzi and R. Bouc. A new formulation for the existence and calculation of nonlinear normal modes. *Journal of Sound and Vibration*, 287(3):545–569, Oct. 2005.
- [9] C. M. Bender and S. A. Orszag. Advanced mathematical methods for scientists and engineers I: Asymptotic methods and perturbation theory. Springer Science & Business Media, 1999.
- [10] Q. S. Bi and P. Yu. Symbolic computation of normal forms for semi-simple cases. *Journal of Computational and Applied Mathematics*, 102(2):195–220, FEB 26 1999. ISSN 0377-0427. doi: {10.1016/S0377-0427(98) 00222-2}.
- [11] G. D. Birkhoff. Dynamical systems, volume 9. American Mathematical Soc., 1927.
- [12] R. E. D. Bishop and D. C. Johnson. The mechanics of vibration. Cambridge University Press, 1960.
- [13] R. E. D. Bishop, G. M. Gladwell, and S. Michaelson. *The matrix analysis of vibration*. Cambridge University Press, 1965.
- [14] N. Boivin, C. Pierre, and S. W. Shaw. Non-linear normal modes, invariance, and modal dynamics approximations of non-linear systems. *Nonlinear Dynamics*, 8:315–346, 1995.
- [15] T. Breunung and G. Haller. Explicit backbone curves from spectral submanifolds of forced-damped nonlinear mechanical systems. *Proc. R. Soc. A*, 474(2213):20180083, 2018.
- [16] A. D. Bruno. Local methods in nonlinear differential equations. Springer, 1989.
- [17] T. D. Burton. On the amplitude decay of strongly non-linear damped oscillators. *Journal of Sound and Vibration*, 87(4):535–541, 1983.

- [18] T. D. Burton. Numerical calculation of nonlinear normal modes in structural systems. *Nonlinear Dynamics*, 49(3):425–441, Aug. 2007.
- [19] G. Buza, S. Jain, and G. Haller. Using spectral submanifolds for optimal mode selection in nonlinear model reduction. *Proceedings of the Royal Society A*, 477(2246):20200725, 2021.
- [20] X. Cabré, E. Fontich, and R. De La Llave. The parameterization method for invariant manifolds iii: overview and applications. *Journal of Differential Equations*, 218(2):444–515, 2005.
- [21] A. Cammarano, T. L. Hill, S. A. Neild, and D. J. Wagg. Bifurcations of backbone curves for systems of coupled nonlinear two mass oscillator. *Nonlinear Dynamics*, 77(1-2):311–320, 2014.
- [22] B. J. Cantwell. Introduction to symmetry analysis. Cambridge University Press, 2002.
- [23] W. Carnegie and Z. F. Reif. Ultraharmonic resonance of a system with an asymmetrical restoring force characteristic. *Journal of Mechanical Engineering Science*, 11(6):592–597, 1969.
- [24] J. R. Cary. Lie transform perturbation theory for Hamiltonian systems. *Physics Reports*, 79(2):129–159, 1981.
- [25] F. Casas, A. Murua, and M. Nadinic. Efficient computation of the Zassenhaus formula. *Computer Physics Communications*, 183(11):2386–2391, 2012.
- [26] T. K. Caughey and M. E. J. O'Kelly. Classical normal modes in damped linear dynamics systems. Journal of Applied Mechanics: Transactions of the ASME, 32:583–588, 1965.
- [27] M. Cenedese and G. Haller. How do conservative backbone curves perturb into forced responses? a Melnikov function analysis. *Proceedings of the Royal Society A*, 476(2234):20190494, 2020.
- [28] M. Cenedese and G. Haller. Stability of forced–damped response in mechanical systems from a Melnikov analysis. *Chaos: An Interdisciplinary Journal of Nonlinear Science*, 30(8):083103, 2020.
- [29] S.-N. Chow, C. Li, and D. Wang. *Normal forms and bifurcation of planar vector fields*. Cambridge University Press, 1994.
- [30] R. W. Clough and J. Penzien. *Dynamics of Structures*. McGraw-Hill, 1993. second edition.
- [31] D. A. Czaplewski, S. Strachan, O. Shoshani, S. W. Shaw, and D. López. Bifurcation diagram and dynamic response of a MEMS resonator with a 1: 3 internal resonance. *Applied Physics Letters*, 114(25):254104, 2019.
- [32] J. P. Den Hartog. Mechanical Vibrations. McGraw-Hill: New York, 1934.
- [33] V. Denis, M. Jossic, C. Giraud-Audine, B. Chomette, A. Renault, and O. Thomas. Identification of nonlinear modes using phase-locked-loop experimental continuation and normal form. *Mechanical Systems and Signal Processing*, 106:430–452, 2018.
- [34] A. Deprit. Canonical transformations depending on a small parameter. *Celestial mechanics*, 1(1):12–30, 1969.
- [35] A. Dhooge, W. Govaerts, Y. A. Kuznetsov, W. Mestrom, A. M. Riet, and B. Sautois. MATCONT and CL MATCONT: Continuation toolboxes in Matlab. Universiteit Gent, Belgium and Utrecht University, The Netherlands, 2006.
- [36] E. J. Doedel, A. R. Champneys, T. F. Fairgrieve, Y. A. Kuznetsov, B. Sandstede, and X. Wang. Auto97. *Continuation and bifurcation software for ordinary differential equations*, 1998.
- [37] A. J. Dragt and J. M. Finn. Lie series and invariant functions for analytic symplectic maps. *Journal of Mathematical Physics*, 17(12):2215–2227, 1976.
- [38] B. Eckhardt. Birkhoff-Gustavson normal form in classical and quantum mechanics. *Journal of Physics A: Mathematical and General*, 19(15):2961, 1986.

- [39] D. A. Ehrhardt and M. S. Allen. Measurement of nonlinear normal modes using multi-harmonic stepped force appropriation and free decay. *Mechanical Systems and Signal Processing*, 76:612–633, 2016.
- [40] D. A. Ehrhardt, T. L. Hill, S. A. Neild, and J. E. Cooper. Veering and nonlinear interactions of a clamped beam in bending and torsion. *Journal of Sound and Vibration*, 416:1–16, 2018.
- [41] A. J. Elliott, A. Cammarano, S. A. Neild, T. L. Hill, and D. J. Wagg. Comparing the direct normal form and multiple scales methods through frequency detuning. *Nonlinear dynamics*, 94(4):2919–2935, 2018.
- [42] A. J. Elliott, A. Cammarano, S. A. Neild, T. L. Hill, and D. J. Wagg. Using frequency detuning to compare analytical approximations for forced responses. *Nonlinear Dynamics*, 98(4):2795–2809, 2019.
- [43] C. Elphick, E. Tirapegui, M. E. Brachet, P. Coullet, and G. looss. A simple global characterization for normal forms of singular vector fields. *Physica D*, 29:95–127, 1987.
- [44] M. Eugeni, F. Mastroddi, and E. H. Dowell. Normal form analysis of a forced aeroelastic plate. *Journal of sound and vibration*, 390:141–163, 2017.
- [45] D. J. Ewins. *Modal Testing*. Research Studies Press, 2000.
- [46] M. Firmbach, F. Fritzsch, R. Ketzmerick, and A. Bäcker. Resonance-assisted tunneling in four-dimensional normal-form Hamiltonians. *Physical Review E*, 99(4):042213, 2019.
- [47] A. P. Gabale and S. C. Sinha. A direct analysis of nonlinear systems with external periodic excitations via normal forms. *Nonlinear Dynamics*, 55(1):79–93, 2009.
- [48] M. Géradin and D. Rixen. *Mechanical Vibrations: Theory and Application to Structural Dynamics*. Wiley Blackwell, 1997.
- [49] A. Giorgilli. Perturbation methods in celestial mechanics. In *Satellite dynamics and space missions*, pages 51–114. Springer, 2019.
- [50] A. Givois, J.-J. Tan, C. Touzé, and O. Thomas. Backbone curves of coupled cubic oscillators in one-to-one internal resonance: bifurcation scenario, measurements and parameter identification. *Meccanica*, 55(3): 481–503, 2020.
- [51] G. M. L. Gladwell. Inverse problems in vibration. Kluwer: Dordrecht, 1986.
- [52] P. Glendinning. Stability, instability and chaos. Cambridge University Press, 1994.
- [53] G. Gobat, L. Guillot, A. Frangi, B. Cochelin, and C. Touzé. Backbone curves, Neimark-Sacker boundaries and appearance of quasi-periodicity in nonlinear oscillators: Application to 1: 2 internal resonance and frequency combs in MEMS. *Meccanica*, pages 1–33, 2021.
- [54] H. Goldstein, J. Safko, and C. Poole. *Classical mechanics*. Pearson Education, 2013.
- [55] J. Guckenheimer and P. Holmes. *Nonlinear oscillations, dynamical systems, and bifurcations of vector fields*. Springer-Verlag: New York, 1983.
- [56] L. Guillot, B. Cochelin, and C. Vergez. A generic and efficient Taylor series-based continuation method using a quadratic recast of smooth nonlinear systems. *International Journal for numerical methods in Engineering*, 119(4):261–280, 2019.
- [57] F. G. Gustavson. On constructing formal integrals of a Hamiltonian system near ail equilibrium point. *The Astronomical Journal*, 71:670, 1966.
- [58] P. Hagedorn and A. DasGupta. Vibrations and waves in continuous mechanical systems. Wiley, 2007.
- [59] G. Haller and S. Ponsioen. Nonlinear normal modes and spectral submanifolds: existence, uniqueness and use in model reduction. *Nonlinear dynamics*, 86(3):1493–1534, 2016.
- [60] G. Haller and S. Ponsioen. Exact model reduction by a slow–fast decomposition of nonlinear mechanical systems. *Nonlinear Dynamics*, 90(1):617–647, 2017.
- [61] M. Haragus and G. looss. *Local bifurcations, center manifolds, and normal forms in infinite-dimensional dynamical systems.* Springer Science & Business Media, 2010.
- [62] A. Haro, M. Canadell, F. J. L., L. A., and M. J. M. *The parameterization method for invariant manifolds. From rigorous results to effective computations.* Springer International Publishing, 2016.
- [63] M. Harsoula, C. Efthymiopoulos, and G. Contopoulos. Analytical forms of chaotic spiral arms. *Monthly Notices of the Royal Astronomical Society*, 459(4):3419–3431, 2016.
- [64] J. He and Z.-F. Fu. Modal Analysis. Oxford: Butterworth-Heinemann, 2001.
- [65] T. Hill, S. Neild, A. Cammarano, and D. Wagg. The influence of phase-locking on internal resonance from a nonlinear normal mode perspective. *Journal of Sound and Vibration*, 379:135–149, 2016.
- [66] T. Hill, S. Neild, and D. Wagg. Comparing the direct normal form method with harmonic balance and the method of multiple scales. *Procedia engineering*, 199:869–874, 2017.
- [67] T. L. Hill, A. Cammarano, S. A. Neild, and D. J. Wagg. Interpreting the forced responses of a two-degreeof-freedom nonlinear oscillator using backbone curves. *Journal of Sound and Vibration*, 349:276–288, 2015.
- [68] T. L. Hill, A. Cammarano, S. A. Neild, and D. J. Wagg. Out-of-unison resonance in weakly nonlinear coupled oscillators. In *Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences*, volume 471, page 20140659. The Royal Society, 2015.
- [69] T. L. Hill, P. L. Green, A. Cammarano, and S. A. Neild. Fast Bayesian identification of a class of elastic weakly nonlinear systems using backbone curves. *Journal of sound and vibration*, 360:156–170, 2016.
- [70] T. L. Hill, A. Cammarano, S. A. Neild, and D. A. W. Barton. Identifying the significance of nonlinear normal modes. Proc. R. Soc. A, 473(2199):20160789, 2017.
- [71] Y. Hiraoka and Y. Kodama. Normal form and solitions. Lecture notes, Euro Summer School, 2001.
- [72] P. Holmes. Poincaré, celestial mechanics, dynamical-systems theory and "chaos". *Physics Reports*, 193 (3):137–163, 1990.
- [73] D. Hong, T. L. Hill, and S. A. Neild. Conditions for the existence of isolated backbone curves. Proceedings of the Royal Society A, 475(2232):20190374, 2019.
- [74] G. I. Hori. Theory of general perturbation with unspecified canonical variable. *Publications of the Astronomical Society of Japan*, 18:287, 1966.
- [75] K. Huseyin and W. Zhang. On the normal forms associated with high dimensional systems. *J. Vib. Acoust.*, 123(2):157–169, 2001.
- [76] D. J. Inman. Engineering vibration. Prentice Hall, 2007.
- [77] G. looss and M. Adelmeyer. Topics in bifurcation theory. World scientific: New-York, 1998.
- [78] G. looss and M. Adelmeyer. *Topics in bifurcation theory and applications*, volume 3. World Scientific, 1998.
- [79] S. Jain and G. Haller. How to compute invariant manifolds and their reduced dynamics in high-dimensional finite-element models? *arXiv preprint arXiv:2103.10264*, 2021.
- [80] L. Jezequel and C. H. Lamarque. Analysis of nonlinear dynamic systems by the normal form theory. *Journal of Sound and Vibration*, 149(3):429–459, SEP 22 1991. ISSN 0022-460X.
- [81] D. Jiang, C. Pierre, and S. W. Shaw. Nonlinear normal modes for vibratory systems under harmonic excitation. *Journal of Sound and Vibration*, 288(4-5):791–812, Dec. 2005.
- [82] D. I. G. Jones. Handbook of viscoelastic vibration damping. Wiley Blackwell, 2001.

- [83] P. B. Kahn and Y. Zarmi. Nonlinear dynamics: a tutorial on the method of normal forms. *American Journal of Physics*, 68(10):907–919, 2000.
- [84] P. B. Kahn and Y. Zarmi. Consistent application of the method of multiple time scales to nonlinear systems. In C. W. Wang, editor, *Nonlinear Phenomena Research Perspectives*, pages 103–130. Nova Science Publishers, Hauppauge NY, 2007.
- [85] P. B. Kahn and Y. Zarmi. *Nonlinear Dynamics: Exploration Through Normal Forms*. Dover Books on Physics, 2014.
- [86] A. Karev and P. Hagedorn. Simultaneous resonance and anti-resonance in dynamical systems under asynchronous parametric excitation. *Journal of Computational and Nonlinear Dynamics*, 15(9):091001, 2020.
- [87] O. S. Katz and E. Efrati. Regular regimes of the harmonic three-mass system. *Physical Review E*, 101(3): 032211, 2020.
- [88] G. Kerschen, M. Peeters, J.-C. Golinval, and A. F. Vakakis. Nonlinear normal modes, Part I: A useful framework for the structural dynamicist. *Mechanical Systems and Signal Processing*, 23(1):170–194, 2009.
- [89] A. C. King, J. Billingham, and S. R. Otto. *Differential equations*. Cambridge, 2003.
- [90] G. Kleyman, M. Paehr, and S. Tatzko. Application of control-based-continuation for characterization of dynamic systems with stiffness and friction nonlinearities. *Mechanics Research Communications*, 106: 103520, 2020.
- [91] Y. Kodama. Normal form, symmetry and infinite dimensional Lie algebra for system of ODE's. *Physics Letters A*, 191(3-4):223–228, 1994.
- [92] Y. Kodama and A. Mikhailov. Obstacles to asymptotic integrability. In *Algebraic Aspects of Integrable Systems*, pages 173–204. Springer, 1997.
- [93] Y. Kodama and A. V. Mikhailov. Symmetry and perturbation of the vector nonlinear Schrödinger equation. *Physica D: Nonlinear Phenomena*, 152:171–177, 2001.
- [94] R. Kogler, A. Plietzsch, P. Schultz, and F. Hellmann. A normal form for grid forming power grid components. *arXiv preprint arXiv:2106.00644*, 2021.
- [95] P.-V. Koseleff. Comparison between Deprit and Dragt-Finn perturbation methods. *Celestial Mechanics and Dynamical Astronomy*, 58(1):17–36, 1994.
- [96] I. Kovacic and M. J. Brennan. *The Duffing equation: nonlinear oscillators and their behaviour*. John Wiley & Sons, 2011.
- [97] M. Krack. Nonlinear modal analysis of nonconservative systems: extension of the periodic motion concept. *Computers & Structures*, 154:59–71, 2015.
- [98] P. Kramer, A. Khan, P. Stathos, and R. E. Lee DeVille. Method of multiple scales with three time scales. In PAMM: Proceedings in Applied Mathematics and Mechanics, volume 7, pages 2040031–2040032. Wiley Online Library, 2007.
- [99] R. J. Kuether and M. S. Allen. A numerical approach to directly compute nonlinear normal modes of geometrically nonlinear finite element models. *Mechanical Systems and Signal Processing*, 46(1):1–15, 2014.
- [100] Y. A. Kuznetsov. *Elements of Applied Bifurcation Theory*. Springer, 2004.
- [101] M. Kwarta and M. S. Allen. Nonlinear normal mode backbone estimation with near-resonant steady state inputs. *Mechanical Systems and Signal Processing*, 162:108046, 2022.
- [102] W. Lacarbonara. Nonlinear Structural Mechanics: Theory, Dynamical Phenomena and Modeling. Springer, 2013.

- [103] W. Lacarbonara and R. Camillacci. Nonlinear normal modes of structural systems via asymptotic approach. International Journal of Solids and Structures, 41(20):5565–5594, Oct. 2004.
- [104] C.-H. Lamarque, C. Touzé, and O. Thomas. An upper bound for validity limits of asymptotic analytical approaches based on normal form theory. *Nonlinear Dynamics*, 70(3):1931–1949, 2012.
- [105] P. Lancaster. Lambda-matrices and vibrating systems. Courier Corporation, 2002.
- [106] B. J. Lazan. Damping of materials and members in structural mechanisms. Pergamon, 1968.
- [107] A. Leung and Q. Zhang. Normal form computation without central manifold reduction. *Journal of Sound & Vibration*, 266:261–279, 2003.
- [108] A. Y. T. Leung and Q. C. Zhang. Complex normal form for strongly non-linear vibration systems exemplified by Duffing-van der Pol equation. *Journal Of Sound and Vibration*, 213(5):907–914, June 1998.
- [109] A. Y. T. Leung and Q. C. Zhang. Higher order normal form and period averaging. *Journal of sound and vibration*, 217(5):795–806, 1998.
- [110] X. Liu and D. J. Wagg. Simultaneous normal form transformation and model-order reduction for systems of coupled nonlinear oscillators. *Proceedings of the Royal Society A*, 475(2228):20190042, 2019.
- [111] X. Liu and D. J. Wagg.  $\varepsilon^2$ -order normal form analysis for a two-degree-of-freedom nonlinear coupled oscillator. In *Nonlinear Dynamics of Structures, Systems and Devices*, pages 25–33. Springer, 2020.
- [112] J. M. Londoño, J. E. Cooper, and S. A. Neild. Identification of systems containing nonlinear stiffnesses using backbone curves. *Mechanical Systems and Signal Processing*, 84:116–135, 2017.
- [113] W. Magnus. On the exponential solution of differential equations for a linear operator. *Communications* on pure and applied mathematics, 7(4):649–673, 1954.
- [114] A. I. Manevitch and L. I. Manevitch. Free oscillations in conservative and dissipative symmetric cubic two-degree-of-freedom systems with closed natural frequencies. *Meccanica*, 38(3):335–348, 2003.
- [115] L. Meirovitch. Fundamentals of vibration. McGraw-Hill: New York, 2001.
- [116] K. Meyer, G. Hall, and D. Offin. *Introduction to Hamiltonian Dynamical Systems and the N-Body Problems*. Springer, 2008.
- [117] K. R. Meyer. Normal forms for Hamiltonian systems. *Celestial mechanics*, 9:517–522, 1974.
- [118] A. Mohamadi, F. A. Ghasemi, and M. Shahgholi. Forced nonlinear vibration and bifurcation analysis of circular cylindrical nanocomposite shells using the normal form. *International Journal of Non-Linear Mechanics*, 134:103733, 2021.
- [119] F. Müller, G. Abeloos, E. Ferhatoglu, M. Scheel, M. R. W. Brake, P. Tiso, L. Renson, and M. Krack. Comparison between control-based continuation and phase-locked loop methods for the identification of backbone curves and nonlinear frequency responses. In *Nonlinear Structures & Systems, Volume 1*, pages 75–78. Springer, 2021.
- [120] J. Murdock. Normal Forms and Unfoldings for Local Dynamical Systems. Springer, 2002.
- [121] A. D. Nashif, D. I. G. Jones, and J. P. Henderson. Vibration damping. John Wiley:New York, 1985.
- [122] A. Nasir, N. Sims, and D. Wagg. Direct normal form analysis of oscillators with different combinations of geometric nonlinear stiffness terms. *Journal of Applied and Computational Mechanics*, 7:1167–1182, 2021.
- [123] A. Nayfeh and P. Pai. Linear and Nonlinear Structural Mechanics. Wiley, 2004.
- [124] A. H. Nayfeh. Method of normal forms. Wiley, 1993.
- [125] A. H. Nayfeh. Method of normal forms. Wiley, 2011.

- [126] A. H. Nayfeh and D. T. Mook. Nonlinear oscillations. John Wiley: New York, 1995.
- [127] A. H. Nayfeh, W. Lacabonara, and C.-M. Chin. Nonlinear normal modes of buckled beams: Three-to-one and one-to-one internal resonances. *Nonlinear Dynamics*, 18:253–273, 1999.
- [128] S. A. Neild and D. J. Wagg. Applying the method of normal forms to second-order nonlinear vibration problems. *Proceedings of the Royal Society of London A*, 467(2128):1141–1163, 2011.
- [129] S. A. Neild, A. R. Champneys, D. J. Wagg, T. L. Hill, and A. Cammarano. The use of normal forms for analysing nonlinear mechanical vibrations. *Phil. Trans. R. Soc. A*, 373(2051):20140404, 2015.
- [130] D. E. Newland. Mechanical vibration analysis and computation. Longman, 1989.
- [131] E. Nicolaidou, V. R. Melanthuru, T. L. Hill, and S. A. Neild. Accounting for quasi-static coupling in nonlinear dynamic reduced-order models. *Journal of Computational and Nonlinear Dynamics*, 15(7), 2020.
- [132] K. Ochs. A comprehensive analytical solution of the nonlinear pendulum. *European Journal of Physics*, 32(2):479, 2011.
- [133] A. Opreni, A. Vizzaccaro, A. Frangi, and C. Touzé. Model order reduction based on direct normal form: Application to large finite element MEMS structures featuring internal resonance. *Nonlinear Dynamics*, 105:1237–1272, 2021.
- [134] M. Peeters, R. Viguié, G. Sérandour, G. Kerschen, and J.-C. Golinval. Nonlinear normal modes, Part II: Toward a practical computation using numerical continuation techniques. *Mechanical systems and signal processing*, 23(1):195–216, 2009.
- [135] M. Peeters, G. Kerschen, and J.-C. Golinval. Dynamic testing of nonlinear vibrating structures using nonlinear normal modes. *Journal of Sound and Vibration*, 330(3):486–509, 2011.
- [136] C. Pierre, D. Y. Jiang, and S. Shaw. Nonlinear normal modes and their application in structural dynamics. *Mathematical Problems in Engineering*, 2006:10847, 2006.
- [137] H. Poincaré. New methods of celestial mechanics (English translation of: Les méthodes nouvelles do la méchanique celeste, Vols 1–3, 1892,1893,1899, edited by D. Goroff). American Institute of Physics, United States, 1993.
- [138] S. Ponsioen, T. Pedergnana, and G. Haller. Automated computation of autonomous spectral submanifolds for nonlinear modal analysis. *Journal of Sound and Vibration*, 420:269–295, 2018.
- [139] S. Ponsioen, S. Jain, and G. Haller. Model reduction to spectral submanifolds and forced-response calculation in high-dimensional mechanical systems. *Journal of Sound and Vibration*, 488:115640, 2020.
- [140] E. J. Putzer. Avoiding the Jordan canonical form in the discussion of linear systems with constant coefficients. *The American Mathematical Monthly*, 73(1):2–7, 1966.
- [141] Z. Rahman and T. D. Burton. On higher order methods of multiple scales in non-linear oscillations-periodic steady state response. *Journal of Sound and Vibration*, 133(3):369–379, 1989.
- [142] R. H. Rand. Nonlinear normal modes in two-degree-of-freedom systems. Journal of Applied Mechanics, 38(2):561–561, 1971.
- [143] S. S. Rao and F. F. Yap. *Mechanical vibrations*, volume 4. Prentice Hall Upper Saddle River, 2011.
- [144] M. Rauscher. Steady oscillations of systems with non-linear and unsymmetrical elasticity. PhD thesis, Massachusetts Institute of Technology, 1937.
- [145] J. W. S. Rayleigh. Theory of sound: Volume 1. Macmillan and Co: London, 1894.
- [146] J. W. S. Rayleigh. Theory of sound: Volume 2. Macmillan and Co: London, 1894.
- [147] G. Rega, W. Lacarbonara, A. H. Nayfeh, and C. M. Chin. Multiple resonances in suspended cables: direct versus reduced-order models. *International Journal of Non-linear Mechanics*, 34(5):901–924, SEP 1999. ISSN 0020-7462.

- [148] L. Renson, D. A. W. Barton, and S. A. Neild. Experimental tracking of limit-point bifurcations and backbone curves using control-based continuation. *International Journal of Bifurcation and Chaos*, 27(01):1730002, 2017.
- [149] L. Renson, T. L. Hill, D. A. Ehrhardt, D. A. W. Barton, and S. A. Neild. Force appropriation of nonlinear structures. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 474 (2214):20170880, 2018.
- [150] L. Renson, A. D. Shaw, D. A. W. Barton, and S. A. Neild. Application of control-based continuation to a nonlinear structure with harmonically coupled modes. *Mechanical Systems and Signal Processing*, 120: 449–464, 2019.
- [151] K. Röbenack. Computation of multiple Lie derivatives by algorithmic differentiation. *Journal of Computational and Applied Mathematics*, 213(2):454–464, 2008.
- [152] R. M. Rosenberg. The normal modes of nonlinear n-degree-of-freedom systems. *Journal of applied Mechanics*, 29(1):7–14, 1962.
- [153] F. Schilder and H. Dankowicz. Continuation core and toolboxes (COCO). Available at https://sourceforge.net/projects/cocotools/, 2017.
- [154] A. D. Shaw, A. R. Champneys, and M. I. Friswell. Normal form analysis of bouncing cycles in isotropic rotor stator contact problems. *International Journal of Mechanical Sciences*, 155:83–97, 2019.
- [155] S. W. Shaw and C. Pierre. Normal modes of vibration for non-linear continuous systems. *Journal of sound and vibration*, 169(3):319–347, 1994.
- [156] Y. Shen, N. Béreux, A. Frangi, and C. Touzé. Reduced order models for geometrically nonlinear structures: Assessment of implicit condensation in comparison with invariant manifold approach. *European Journal* of Mechanics-A/Solids, 86:104165, 2021.
- [157] O. Shoshani, S. W. Shaw, and M. I. Dykman. Anomalous decay of nanomechanical modes going through nonlinear resonance. *Scientific reports*, 7(1):1–8, 2017.
- [158] S. C. Sinha and R. Pandiyan. Analysis of quasilinear dynamical systems with periodic coefficients via Liapunov-Floquet transformation. *International Journal of Non-Linear Mechanics*, 29(5):687–702, 1994.
- [159] C. S. M. Sombroek, P. Tiso, L. Renson, and G. Kerschen. Numerical computation of nonlinear normal modes in a modal derivative subspace. *Computers & Structures*, 195:34–46, 2018.
- [160] N. Srinil and G. Rega. Two-to-one resonant multi-modal dynamics of horizontal/inclined cables. Part II: Internal resonance activation, reduced-order models and nonlinear normal modes. *Nonlinear Dynamics*, 48(3):253–274, MAY 2007. ISSN 0924-090X. doi: {10.1007/s11071-006-9087-z}.
- [161] G. Stephenson and P. M. Radmore. *Advanced mathematical methods for engineering and science students*. Cambridge, 1990.
- [162] S. H. Strogatz. Nonlinear Dynamics and Chaos. Perseus Books Group, 2019.
- [163] R. Szalai, D. Ehrhardt, and G. Haller. Nonlinear model identification and spectral submanifolds for multidegree-of-freedom mechanical vibrations. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 473(2202):20160759, 2017.
- [164] W. Szemplinska. The Behaviour of Nonlinear Vibrating Systems: Volume II: Advanced Concepts and Applications to Multi-Degree-of-Freedom Systems, volume 12. Springer Science & Business Media, 1990.
- [165] J. M. T. Thompson and H. B. Stewart. Nonlinear dynamics and chaos. John Wiley: Chichester, 2002.
- [166] W. T. Thompson. Theory of vibration with applications. Chapman Hall: London, 1993. 4th Ed.
- [167] S. P. Timoshenko. Vibration problems in engineering. Van Nostrand, 1937.
- [168] C. Touzé. A normal form approach for nonlinear normal modes, 2003. Publications du LMA, number 156, (ISSN: 1159-0947, ISBN:2-909669-20-3), Marseille.

- [169] C. Touzé. Normal form theory and nonlinear normal modes: Theoretical settings and applications. In *Modal analysis of nonlinear mechanical systems*, pages 75–160. Springer, 2014.
- [170] C. Touzé and M. Amabili. Nonlinear normal modes for damped geometrically nonlinear systems: Application to reduced-order modelling of harmonically forced structures. *Journal of Sound and Vibration*, 298 (4-5):958–981, DEC 22 2006. ISSN 0022-460X. doi: {10.1016/j.jsv.2006.06.032}.
- [171] C. Touzé and O. Thomas. Non-linear behaviour of free-edge shallow spherical shells: Effect of the geometry. *International Journal of non-linear Mechanics*, 41(5):678–692, 2006.
- [172] C. Touzé, O. Thomas, and A. Chaigne. Hardening/softening behaviour in non-linear oscillations of structural systems using non-linear normal modes. *Journal of Sound and Vibration*, 273(1-2):77–101, MAY 21 2004. ISSN 0022-460X. doi: {10.1016/j.jsv.2003.04.005}.
- [173] C. Touzé, A. Vizzaccaro, and O. Thomas. Model order reduction methods for geometrically nonlinear structures: a review of nonlinear techniques. *Nonlinear Dynamics*, 105:1141–1190, 2021.
- [174] J. M. Tuwankotta and F. Verhulst. Symmetry and resonance in Hamiltonian systems. *SIAM Journal on Applied Mathematics*, 61(4):1369–1385, 2001.
- [175] A. F. Vakakis, L. I. Manevitch, Y. V. Mikhlin, V. N. Pilipchuk, and A. A. Zevin. *Normal modes and localization in nonlinear systems*. Wiley: New York, 1996.
- [176] C. I. Van Damme, M. S. Allen, and J. J. Hollkamp. Updating geometrically nonlinear reduced-order models using nonlinear modes and harmonic balance. *AIAA Journal*, 58(8):3553–3568, 2020.
- [177] Z. Veraszto, S. Ponsioen, and G. Haller. Explicit third-order model reduction formulas for general nonlinear mechanical systems. *Journal of Sound and Vibration*, 468:115039, 2020.
- [178] F. Verhulst. Integrability and non-integrability of Hamiltonian normal forms. *Acta Applicandae Mathematicae*, 137(1):253–272, 2015.
- [179] A. Vizzaccaro, A. Givois, P. Longobardi, Y. Shen, J.-F. Deü, L. Salles, C. Touzé, and O. Thomas. Nonintrusive reduced order modelling for the dynamics of geometrically nonlinear flat structures using three-dimensional finite elements. *Computational Mechanics*, 66(6):1293–1319, 2020.
- [180] A. Vizzaccaro, L. Salles, and C. Touzé. Comparison of nonlinear mappings for reduced-order modelling of vibrating structures: Normal form theory and quadratic manifold method with modal derivatives. *Nonlinear Dynamics*, pages 1–36, 2020.
- [181] A. Vizzaccaro, Y. Shen, L. Salles, J. Blahoš, and C. Touzé. Direct computation of nonlinear mapping via normal form for reduced-order models of finite element nonlinear structures. *Computer Methods in Applied Mechanics and Engineering*, 384:113957, 2021.
- [182] M. Volvert and G. Kerschen. Phase resonance nonlinear modes of mechanical systems. *Nonlinear Dynamics*, 511:116355, 2021.
- [183] D. J. Wagg and S. A. Neild. *Nonlinear vibration with control For flexible and adaptive structures.* Springer, 2nd edition, 2015.
- [184] M. Wall, M. S. Allen, and R. J. Kuether. Observations of modal coupling due to bolted joints in an experimental benchmark structure. *Mechanical Systems and Signal Processing*, 162:107968, 2022.
- [185] G. B. Warburton. *The Dynamical Behaviour of Structures: Structures and Solid Body Mechanics Series*. Elsevier, 1964.
- [186] P. M. B. Waswa and S. Redkar. A direct approach for simplifying nonlinear systems with external periodic excitation using normal forms. *Nonlinear Dynamics*, 99(2):1065–1088, 2020.
- [187] W. Wei and H. Yabuno. Subcritical hopf and saddle-node bifurcations in hunting motion caused by cubic and quintic nonlinearities: Experimental identification of nonlinearities in a roller rig. *Nonlinear Dynamics*, 98(1):657–670, 2019.

- [188] S. M. Wooden and S. C. Sinha. Analysis of periodic-quasiperiodic nonlinear systems via Lyapunov-Floquet transformation and normal forms. *Nonlinear Dynamics*, 47(1):263–273, 2007.
- [189] K. Worden and G. R. Tomlinson. *Nonlinearity in structural dynamics*. IOP, 2000.
- [190] P. Yu, W. Zhang, and Q. Bi. Vibration analysis on a thin plate with the aid of computation of normal forms. *International Journal of Non-linear Mechanics*, 36(4):597–627, JUN 2001. ISSN 0020-7462. doi: {10.1016/S0020-7462(00)00023-8}.
- [191] Y. Zarmi. From single-to multiple-soliton solutions of the perturbed KdV equation. *Physica D: Nonlinear Phenomena*, 237(23):2987–3007, 2008.
- [192] W. Zhang, K. Huseyin, and M. Ye. On the computation of the coefficients associated with high order normal forms. *Journal of sound and vibration*, 232(3):525–540, 2000.
- [193] W. Zhang, F. X. Wang, and J. W. Zu. Computation of normal forms for high dimensional non-linear systems and application to non-planar non-linear oscillations of a cantilever beam. *Journal of Sound & Vibration*, 278(4-5):949–974, DEC 22 2004. ISSN 0022-460X. doi: {10.1016/j.jsv.2003.10.021}.
- [194] W. Zhang, F. Wang, and M. Yao. Global bifurcations and chaotic dynamics in nonlinear nonplanar oscillations of a parametrically excited cantilever beam. *Nonlinear Dynamics*, 40(3):251–279, 2005.
- [195] V. F. Zhuravlev. A new algorithm for Birkhoff normalization of Hamiltonian systems. *Journal of applied mathematics and mechanics*, 61(1):9–13, 1997.